# Low-Cost Small-Scale Wind Power Generation

#### David Michael Whaley

Thesis submitted for the degree of **Doctor of Philosophy** 



The School of Electrical & Electronic Engineering, Faculty of Engineering, Computer & Mathematical Sciences The University of Adelaide, Australia

July, 2009

© Copyright 2009 David M. Whaley All Rights Reserved.



Dedicated to my late grandmother, Χρισούλα Ηλιάδου (Chryssoula Iliadis)

## **Table of Contents**

Table of Contents	v
Abstract	xi
Statement of Originality	xiii
Acknowledgements	XV
List of Publications	xvii
Conventions	xix
Nomenclature	xxi
List of Figures	xxxi
List of Tables	xxxix
1 Introduction	1
1.1 Wind Energy	1
1.1.1 Electricity Usage and Conventional Generation	
1.1.2 Alternative Energy Sources	3
1.1.3 Large and Small-Scale Turbine Classification	4
1.1.4 Small-Scale Turbine Development	5
1.1.5 Technology Improvement	6
1.1.6 Market Growth $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$	6
1.2 Principles of Wind Power	7
1.2.1 Wind and Turbine Power	7
1.2.2 Coefficient of Performance	8

	1.2.3	Principles of Turbine Operation	9
1.3	Small-	Scale Wind Turbines	11
	1.3.1	Applications	11
	1.3.2	Turbine Properties	12
	1.3.3	Generator Varieties	13
	1.3.4	Current Trends	15
	1.3.5	Inductance Classification of PM Generators	16
1.4	Stand	alone Power Converters	18
	1.4.1	Common Power Converters	18
	1.4.2	Uncontrolled Rectifier Operation	21
	1.4.3	Switched-Mode Rectifier	23
	1.4.4	Inverter Operation	25
	1.4.5	Power Comparison of Standalone Converters	27
1.5	Grid-(	Connected Inverters	29
	1.5.1	Introduction	29
	1.5.2	Line-Commutated Inverters	30
	1.5.3	Self-Commutated Inverters	32
	1.5.4	Transformer Type	35
	1.5.5	Voltage Source Topologies	37
	1.5.6	Current Source Topologies	40
	1.5.7	Current Trends	42
	1.5.8	Proposed Grid-Connected Inverter	43
1.6	Thesis	o Overview	44
	1.6.1	Aim of Research	44
	1.6.2	Justification for Research	44
	1.6.3	Original Contributions	45
	1.6.4	Thesis Structure	45

#### I Investigation of Switched-Mode Rectifier for Standalone Power Converter 49

<b>2</b>	Hig	h Indu	ctance PM Generator Characteristics	51
	2.1	Introd	uction	51
		2.1.1	Ideal Machine Model and Current vs. Voltage Locus	54

	2.1.2	Inductance Classification	55
2.2	Machi	ine Characterisation	57
	2.2.1	Realistic Machine Model and Effect on Loci	57
	2.2.2	Test Arrangement	58
	2.2.3	Open-Circuit Test	60
	2.2.4	Short-Circuit Test	62
	2.2.5	Machine Losses	65
	2.2.6	Machine Properties	66
2.3	Machi	ine Modelling	67
	2.3.1	Analytical Model	67
	2.3.2	$\mathrm{PSIM}^{\textcircled{B}}$ Model	70
	2.3.3	Power Maximisation	72
	2.3.4	Model Comparison	73
2.4	Resist	vive Load Testing	74
	2.4.1	3ph Resistive Loading	74
	2.4.2	DC Resistive Loading	77
2.5	Chapt	ter Summary	83
S;			
SWI	tched-	Mode Rectifier Operation	85
3.1	tched- Introd	Mode Rectifier Operation	<b>85</b> 85
3.1	tched- Introd 3.1.1	Mode Rectifier Operation luction	<b>85</b> 85 86
3.1	tched- Introd 3.1.1 3.1.2	Mode Rectifier Operation luction	<b>85</b> 85 86 88
3.1	Introd 3.1.1 3.1.2 3.1.3	•Mode Rectifier Operation         luction	<ul> <li>85</li> <li>85</li> <li>86</li> <li>88</li> <li>90</li> </ul>
3.1 3.2	tched- Introd 3.1.1 3.1.2 3.1.3 Dynar	Mode Rectifier Operation         luction	<ul> <li>85</li> <li>85</li> <li>86</li> <li>88</li> <li>90</li> <li>92</li> </ul>
3.1 3.2	tched- Introd 3.1.1 3.1.2 3.1.3 Dynar 3.2.1	Mode Rectifier Operation         luction	<ul> <li>85</li> <li>85</li> <li>86</li> <li>88</li> <li>90</li> <li>92</li> <li>92</li> </ul>
3.1 3.2	tched- Introd 3.1.1 3.1.2 3.1.3 Dynar 3.2.1 3.2.2	Mode Rectifier Operation         luction	<ul> <li>85</li> <li>85</li> <li>86</li> <li>88</li> <li>90</li> <li>92</li> <li>92</li> <li>92</li> <li>92</li> </ul>
3.1 3.2	tched- Introd 3.1.1 3.1.2 3.1.3 Dynar 3.2.1 3.2.2 3.2.3	Mode Rectifier Operation         luction         Switched-Mode Rectifier Model         SMR Operation         SMR Properties         SMR Properties         Imometer Testing         Test Arrangement         SMR Test Results         Summary of Testing	<ul> <li>85</li> <li>85</li> <li>86</li> <li>88</li> <li>90</li> <li>92</li> <li>92</li> <li>92</li> <li>92</li> <li>92</li> <li>92</li> <li>92</li> </ul>
3.1 3.2 3.3	tched- Introd 3.1.1 3.1.2 3.1.3 Dynar 3.2.1 3.2.2 3.2.3 Wind	Mode Rectifier Operation         luction         Switched-Mode Rectifier Model         SMR Operation         SMR Properties         SMR Properties         mometer Testing         Test Arrangement         SMR Test Results         Summary of Testing         Tunnel Testing	85 85 86 88 90 92 92 92 92 92 92
3.1 3.2 3.3	tched- Introd 3.1.1 3.1.2 3.1.3 Dynar 3.2.1 3.2.2 3.2.3 Wind 3.3.1	Mode Rectifier Operation         luction         Switched-Mode Rectifier Model         SMR Operation         SMR Properties         SMR Properties         mometer Testing         Test Arrangement         SMR Test Results         Summary of Testing         Tunnel Testing         Test Arrangement         Summary of Testing         Test Arrangement         Summary of Testing         Test Arrangement         Summary of Testing         Test Arrangement         Test Arrangement	85 85 86 88 90 92 92 92 92 92 91 00
3.1 3.2 3.3	tched- Introd 3.1.1 3.1.2 3.1.3 Dynar 3.2.1 3.2.2 3.2.3 Wind 3.3.1 3.3.2	Mode Rectifier Operation         luction	85 85 86 88 90 92 92 92 92 92 9100 100
3.1 3.2 3.3	tched- Introd 3.1.1 3.1.2 3.1.3 Dynar 3.2.1 3.2.2 3.2.3 Wind 3.3.1 3.3.2 3.3.3	Mode Rectifier Operation         luction         Switched-Mode Rectifier Model         SMR Operation         SMR Properties         SMR Properties         mometer Testing         Test Arrangement         SMR Test Results         Summary of Testing         Tunnel Testing         Turbine Coefficient of Performance         Open-Loop Control Mode	85 85 86 88 90 92 92 92 92 92 91 00 100 100 102
3.1 3.2 3.3	tched- Introd 3.1.1 3.1.2 3.1.3 Dynar 3.2.1 3.2.2 3.2.3 Wind 3.3.1 3.3.2 3.3.3 3.3.4	Mode Rectifier Operation         luction         Switched-Mode Rectifier Model         SMR Operation         SMR Properties         SMR Properties         mometer Testing         Test Arrangement         Summary of Testing         Tunnel Testing         Test Arrangement         Turbine Coefficient of Performance         Open-Loop Control Mode         Comparison of Control Modes	<ul> <li>85</li> <li>85</li> <li>86</li> <li>88</li> <li>90</li> <li>92</li> &lt;</ul>
	<ul> <li>2.2</li> <li>2.3</li> <li>2.4</li> <li>2.5</li> <li>2.5</li> </ul>	<ul> <li>2.2 Mach</li> <li>2.2.1</li> <li>2.2.2</li> <li>2.2.3</li> <li>2.2.4</li> <li>2.2.5</li> <li>2.2.6</li> <li>2.3 Mach</li> <li>2.3.1</li> <li>2.3.2</li> <li>2.3.3</li> <li>2.3.4</li> <li>2.4 Resist</li> <li>2.4.1</li> <li>2.4.2</li> <li>2.5 Chap</li> </ul>	<ul> <li>2.2 Machine Characterisation</li> <li>2.2.1 Realistic Machine Model and Effect on Loci</li> <li>2.2.2 Test Arrangement</li> <li>2.2.3 Open-Circuit Test</li> <li>2.2.4 Short-Circuit Test</li> <li>2.2.5 Machine Losses</li> <li>2.2.6 Machine Properties</li> <li>2.3 Machine Modelling</li> <li>2.3.1 Analytical Model</li> <li>2.3.2 PSIM<sup>®</sup> Model</li> <li>2.3.3 Power Maximisation</li> <li>2.3.4 Model Comparison</li> <li>2.4 Resistive Load Testing</li> <li>2.4.1 3ph Resistive Loading</li> <li>2.4.2 DC Resistive Loading</li> <li>2.5 Chapter Summary</li> </ul>

#### II Investigation of Grid-Connected Inverter based on Switched-Mode Rectifier Topology 115

4	Sim	ulatio	n and Test of 150W GC Inverter	117
	4.1	Introd	luction	. 117
		4.1.1	Inverter Requirements	. 118
		4.1.2	Desirable Features	. 120
	4.2	Propo	sed Concept	. 122
		4.2.1	Inverter Overview	. 122
		4.2.2	Constant Current Source	. 123
		4.2.3	Current Wave-Shaper	. 124
		4.2.4	Unfolding Circuit	. 126
		4.2.5	Low-Pass Filter	. 127
	4.3	Test S	Set-up, Implementation and Simulation	. 128
		4.3.1	Dynamometer Test Arrangement	. 128
		4.3.2	Constant Current Source - PM Generator	. 131
		4.3.3	Power Electronics and Control Implementation	. 133
		4.3.4	Inverter Simulation	. 135
	4.4	Exper	imental Testing	. 138
		4.4.1	Proof of Concept - Resistive Loading	. 138
		4.4.2	Constant Current Assumption	. 141
		4.4.3	Grid-Connected Testing	. 145
	4.5	Chapt	er Summary	. 153
<b>5</b>	Inv	erter A	Analysis and Control	155
	5.1	Analy	sis of Non-Ideal Constant Current Source	. 155
		5.1.1	Fluctuating Input Current	. 156
		5.1.2	Rectifier Ripple	. 160
		5.1.3	The Resulting Input Current	. 162
	5.2	Total	Harmonic Distortion	. 164
		5.2.1	Fluctuating Input Power	. 164
		5.2.2	Rectifier Ripple	. 166
		5.2.3	The Resulting Input Current	. 167
		5.2.4	PWM Switching Schemes	. 171
		5.2.5	The Inverter Output Current	. 175

		5.2.6	Reducing Harmonic Distortion
	5.3	Low-P	Pass Filter Analysis
		5.3.1	Filter Response
		5.3.2	Filter Damping
		5.3.3	Damped Filter Response and Configuration Comparison 184
		5.3.4	Design Considerations
		5.3.5	Harmonic Attenuation and Distortion
		5.3.6	Power Factor Requirements
		5.3.7	Power Loss
		5.3.8	Design Trade-Offs - Unipolar PWM Waveform
		5.3.9	Effect of Non-Ideal Current Source
		5.3.10	Alternative (Third-Order) Filter Configurations
	5.4	Feed-F	Forward Control
		5.4.1	Introduction
		5.4.2	Aim of Proposed Feed-Forward Control
		5.4.3	Controller 1 (FFC 1): Sample Machine Frequency
		5.4.4	Controller 2 (FFC 2): Sample Inverter Input Current
		5.4.5	Comparison of Feed-Forward and Open-Loop Control
	5.5	Chapt	er Summary
6	$\mathbf{Des}$	ign an	d Simulation of 1kW GC Inverter System 225
	6.1	Turbir	ne Sizing and Machine Parameter Selection
		6.1.1	System Assumptions
		6.1.2	Turbine Power and Size Calculations
		6.1.3	Generator Equivalent Circuit Parameter Selection
	6.2	Low-P	Pass Filter Design
		6.2.1	Design Criteria
		6.2.2	Component Selection
		6.2.3	Filter Simulation - Ideal Current Source
		6.2.4	Filter Simulation - PM Generator Current Source
	6.3	Demo	nstration of Feed-Forward Control
		6.3.1	Control Implementation
		6.3.2	Proof of Concept at Rated Wind Speed
		6.3.3	Current Command Variation at Rated Wind Speed
	6.4	Inverte	er Simulation for Wide Wind Speed Range

		6.4.1 Turbine Characteristics	. 252
		6.4.2 Power Control Modes	. 254
		6.4.3 Optimised Component Selection	. 255
		6.4.4 Inverter Simulations	. 256
		6.4.5 Efficiency Analysis	. 259
	6.5	Chapter Summary	. 262
7	Con	clusions and Future Work	263
	7.1	Summary and Conclusions	. 263
	7.2	Original Contributions	. 265
	7.3	Recommendations for Future Work	. 267
A	ppen	dices	269
$\mathbf{A}$	$\mathbf{PW}$	M Control Strategies and Low-Pass Filter Design Trade-Offs	269
	A.1	PWM Switching Schemes	. 269
		A.1.1 Bipolar and Unipolar Pulse-Width Modulation	. 269
		A.1.2 Selective Harmonic Elimination	. 269
		A.1.3 Current Hysteresis	. 270
		A.1.4 Space Vector Modulation	. 271
	A.2	Low-Pass Filter Design	. 272
		A.2.1 Power Loss vs. THD Trade-Off - Unipolar PWM Case	. 272
в	Rel	evant Publications	273
	B.1	Wind Turbine Control using SMR Paper	. 274
	B.2	Novel Low-Cost Grid-Connected Inverter Paper	. 284
$\mathbf{C}$	Mic	rocontroller Code	291
	C.1	Switched-Mode Rectifier	. 291
	C.2	Grid-Connected Inverter	. 294
Re	efere	nces	299

### Abstract

This research investigates a low-cost generator and power electronics unit for smallscale (<10kW) wind turbines, for both standalone and grid-connected applications. The proposed system uses a high-inductance permanent magnet generator together with a switched-mode rectifier (SMR) to produce a variable magnitude output current. The high inductance characteristic allows the generator to operate as a current source, which has the following advantages over conventional low-inductance generator (voltage source) systems: it offers simple control, and avoids the need for bulky / costly energy storage elements, such as capacitors and inductors.

The SMR duty-cycle is controlled in an open-loop manner such that 1) maximum power is obtained for wind speeds below rated, and 2) the output power and turbine speed is limited to safe values above rated wind speed. This topology also has the ability to extract power at low wind speeds, which is well suited to small-scale wind turbines, as there is often limited flexibility in their location and these commonly see low average wind speeds.

The thesis is divided into two parts; the first part examines the use of the SMR as a DC-DC converter, for use in standalone applications. The duty-cycle is essentially kept constant, and is only varied for maximum power tracking and turbine speed / power limiting purposes. The SMR operates in to a fixed voltage source load, and has the ability to allow current and hence power to be drawn from the generator even at low wind and hence turbine speeds, making it ideal for battery charging applications. Initial dynamometer testing and limited wind-tunnel testing of a commercially available wind turbine show that turbine power can be maximised and its speed can be limited by adjusting the SMR duty-cycle in an open-loop manner.

The second part of the thesis examines the use of the SMR as a DC-AC converter for grid-connected applications. The duty-cycle is now modulated sinusoidally at the mains frequency such that the SMR produces an output current that resembles a fullwave rectified sinewave that is synchronised to the mains voltage. An additional H- bridge inverter circuit and low-pass filter is used to unfold, filter and feed the sinusoidal output current in to the utility grid. Simulation and initial resistive load and preliminary grid-connected tests were used to prove the inverter concept, however, the permanent magnet generator current source is identified as non-ideal and causes unwanted harmonic distortion.

The generator harmonics are analysed, and the system performance is compared with the Australian Standard THD requirement. It is concluded that the harmonics are caused by 1) the low-cost single-phase output design, 2) the use of an uncontrolled rectifier, and 3) the finite back-EMF voltage. The extent of these harmonics can be predicted based on the inverter operating conditions. A feed-forward current compensation control algorithm is investigated, and shown to be effective at removing the harmonics caused by the nonideal current source. In addition, the unipolar PWM switching scheme, and its harmonic components are analysed. The low-pass filter design is discussed, with an emphasis on power factor and THD grid requirements. A normalised filter design approach is used that shows how design aspects, such as cutoff frequency and quality factor, affect the filter performance. The filter design is shown to be a trade-off between the output current THD, power loss, and quality factor.

The final chapter summarises the thesis with the design and simulation of a 1kW single-phase grid-connected inverter. The inverter is designed based on the low-pass filter and feed-forward compensation analysis, and is shown to deliver an output current to the utility grid that adheres to the Australian Standards.

## **Statement of Originality**

This work contains no material which has been accepted for the award of any other degree or diploma in any university or other tertiary institution to David M. Whaley and, to the best of my knowledge and belief, contains no material previously published or written by another person, except where due reference has been made in the text.

I give consent to this copy of my thesis, when deposited in the University Library, being made available for loan and photocopying, subject to the provisions of the Copyright Act 1968.

I also give permission for the digital version of my thesis to be made available on the web, via the Universitys digital research repository, the Library catalogue, the Australasian Digital Theses Program (ADTP) and also through web search engines, unless permission has been granted by the University to restrict access for a period of time.

Signed

Date

### Acknowledgements

First and foremost I wish to thank my supervisor and mentor, Dr Wen L. Soong, for his constant guidance, support, encouragement, help and advice during the course of my postgraduate studies. I also thank my co-supervisor, Associate Professor Nesimi Ertugrul, for his assistance and advice.

I am sincerely grateful to the School of Electrical and Electronic Engineering and the Adelaide Graduate Centre, at The University of Adelaide, for providing financial assistance as part of an Australian Research Council Discovery Grant, DP0342874. I would also like to thank the School of Electrical and Electronic Engineering for providing facilities to allow me to carry out this research. I am extremely grateful for the support from the School's workshop staff in the construction of the test rigs and machine set-up, in particular the assistance provided by Stuart Brand and Ian Linke. I am also thankful to Stan Woithe and Bruce Lucas from the School of Civil and Environmental Engineering in allowing the use of their wind tunnel. I would also like to thank Dr Damien Leclercq, from the School of Mechanical Engineering, for technical discussions regarding wind turbines.

I thank all the members of the School of Electrical and Electronic Engineering and especially the members of the 'Power and Control Systems' group, the administration team, and the members of the Computer Support Request team, for their friendliness and help over the years.

Finally, I wish to thank members of my family for their constant support throughout this journey, especially from my fiancè, Allison J.K. Gill. I would also like to thank my brother, Paul J. Whaley for providing financial assistance at various times throughout the course of the project. Lastly, a special thanks to my mother, Anastasia Whaley, for providing much needed financial assistance and patience during the latter stages of this research project.

### List of Publications

- G. Ertasgin, D.M. Whaley, N. Ertugrul and W.L. Soong, "Implementation and Performance Evaluation of a Low-Cost Current-Source Grid-Connected Inverter for PV Application", in Proceedings of the IEEE International Conference on Sustainable Energy Technologies, Nov. 2008, Singapore.
- [2] G. Ertasgin, D.M. Whaley, N. Ertugrul and W.L. Soong, "Analysis and Design of Energy Storage for Current-Source 1-ph Grid-Connected PV Inverters", in Proceedings of the IEEE Applied Power Electronics Conference and Exposition, Feb., 2008, pp. 1229 – 1234.
- [3] G. Ertasgin, D.M. Whaley, N. Ertugrul, and W.L. Soong, "A Current-Source Grid-Connected Converter Topology for Photovoltaic Systems", in Proceedings of Australasian Universities Power Engineering Conference, 2006.
- [4] D.M. Whaley, G. Ertasgin, W.L. Soong, N. Ertugrul, J. Darbyshire, H. Dehbonei, and C.V. Nayar, "Investigation of a Low-Cost Grid-Connected Inverter for Small-Scale Wind Turbines Based on a Constant-Current Source PM Generator", in Proceedings of the IEEE Industry Electronics, Nov., 2006, pp. 4297 – 4302.
- [5] C.Z. Liaw, D.M. Whaley, W.L. Soong, and N. Ertugrul, "Implementation of Inverterless Control of Interior Permanent Magnet Alternators", in *IEEE Transactions* on *Industry Applications*, vol. 42, no. 2, Mar. - Apr., 2006, pp. 536 – 544.
- [6] D.M. Whaley, W.L. Soong, and N. Ertugrul, "Investigation of Switched-Mode Rectifier for Control of Small-Scale Wind Turbines", in Proceedings of the IEEE Industry Applications Conference, vol. 4, Oct., 2005, pp. 2849 – 2856.
- [7] C.Z. Liaw, D.M. Whaley, W.L. Soong, and N. Ertugrul, "Implementation of Inverterless Control of Interior Permanent Magnet Alternators", in Proceedings of the IEEE Industry Applications Conference, Oct., 2004.

[8] D.M. Whaley, W.L. Soong, and N. Ertugrul, "Extracting More Power from the Lundell Car Alternator", in Proceedings of Australasian Universities Power Engineering Conference, 2004.

Portions of the work presented in this thesis have been previously published. The material in Chapters 2 and 3 correspond to work in publications [6], whilst Chapter 4 corresponds to the work presented in publication [4]. Reprints of these publications are found in appendix B, for convenience.

## Conventions

This thesis employs the IEEE reference style for citations, and is written using Australian English, as defined by the Macquarie English Dictionary 2005.

All voltages and currents shown in figures and equations are expressed as RMS (root-mean squared) quantities, unless otherwise stated.

The *hat* symbol is used in Chapters 5 and 6 to indicate peak value, i.e.  $\hat{\alpha}$  and  $\hat{\alpha}_0$  indicate the peak values of  $\alpha$  and  $\alpha_0$ , respectively. Similarly, the *check* symbol is used in Chapter 6 to represent the nadir (minimum) value, e.g.  $\check{\beta}$  represents the minimum value of  $\beta$ .

Measured data is represented by hollow points, e.g. circles, squares, diamonds etc. and is often accompanied by solid lines that correspond to the equivalent analytical or computer based simulations. Multiple cases of measured (and simulated) data commonly appear on a single figure, and are differentiated by colour and shape. In contrast, coloured / shaded points represent calculated data. These are also shown with solid lines, however, these are for aesthetic purposes, i.e. they simply join the calculated data.

The above convention is used for the majority of this thesis, i.e. Chapters 2 to 5, however, the convention is modified for Chapter 6, as the data presented in this chapter is either simulated or analytically calculated. The simulated data, of Chapter 6 is hence shown as shaded points, whilst the analytical calculations are shown by the solid lines.

## Nomenclature

$\alpha$	normalised rectifier voltage	pu
$\alpha_0$	ratio of grid to open-circuit rectifier voltage	
$\alpha_{cu}$	temperature coefficient of copper	/°C
$\beta$	normalised rectifier current	pu
$\check{eta}$	normalised minimum inverter input current	pu
$\beta_{app}$	normalised approximated rectifier current	pu
$\beta_{exp}$	normalised experimental rectifier current	pu
$\beta_{id}$	normalised ideal rectifier current	pu
$\Delta$	difference	
δ	skin depth	m
$\eta_{gen}$	generator efficiency	%
$\eta_{inv}$	inverter efficiency	%
λ	tip-speed ratio	
$\mu$	permability	H/m
ω	machine angular speed	rad/s
ω	turbine angular speed	rad/s
$\omega_{cn}$	normalised cutoff frequency (relative to $f_1$ )	pu
$\omega_e$	electrical angular frequency	rad/s

$\omega_g$	grid angular frequency	rad/s
$\omega_m$	mechanical angular frequency	rad/s
$\phi$	filter delay	$\deg$
$\phi$	power factor angle	deg
$\Psi_m$	RMS flux linkage	Wb or Vs
ρ	air density	$\rm kg/m^3$
σ	conductivity	$(\Omega m)^{-1}$
$\widehat{\alpha}_0$	peak value of $\alpha_0$	
ξ	saliency ratio	
C	capacitance	F
$c_p$	turbine coefficient of performance	
d	duty-cycle	%
$d_a$	adjusted duty-cycle	%
$d_i$	stored duty-cycle	%
dB	decibels	
E	induced back-EMF voltage	V
f	frequency	Hz
$f_1$	fundamental frequency	Hz
$f_{cn}$	normalised cutoff frequency (relative to $f_{sw}$ )	pu
$f_c$	cutoff frequency	Hz
$f_m$	machine frequency	Hz
$f_{res}$	resonant frequency	Hz
$f_{sw}$	switching frequency	Hz

H(s)	filter transfer function	
$h_1$	fundamental harmonic magnitude	%
$h_f$	harmonic frequency	Hz
$h_m$	harmonic magnitude at $m$ multiples of $f_1$	%
$h_{tot}$	total harmonic components	%
Ι	current	А
$I^*$	compensation current command	А
$i_{c exp}(t)$	compensated current using the experimental I-V locus	pu
$i_{c\ id}(t)$	compensated current using the ideal I-V locus	pu
$I_{ch}$	characteristic current	А
$i_c(t)$	time-varying compensated current	А
$I_{DC}$	DC current	А
$I_d$	damping resistor current	А
$I_f$	damping resistor current (from inverter)	А
$I_g$	grid drawn current (from grid)	А
$I_{inv}$	inverter output current	А
$I_{in}$	input current	А
$I_L$	line current	А
$i_{out \ (id)}(t)$	normalised time-varying ideal output current	pu
I <sub>out</sub>	output current	А
$I_{ph}$	phase current	А
$i_{R\ (id)}(t)$	normalised time-varying ideal rectifier voltage	pu
$I_{R min}$	minimum rectifier output current	А

$I_R$	rectifier output current	А
$i_R(t)$	normalised time-varying rectifier current	pu
$i_{ws\ (id)}(t)$	normalised time-varying ideal wave-shaper current	pu
$I_{ws}$	wave-shaper current	А
j	$\sqrt{-1}$	
k	back-EMF constant	V/rpm
$k_{ph}$	phase back-EMF constant	V/rpm
L	inductance	Н
$L_1$	transformer primary inductance	Н
$L_2$	transformer secondary inductance	Н
$L_{eq}$	equivalent inductance	Н
$L_{ph}$	phase inductance	Н
$L_s$	stator inductance	Н
m	number of machine phases	
m	positive integer	
$m_a$	modulation index	%
n	machine / generator speed	rpm
n	positive odd integer	
n	transformer turns ratio	
$n_k$	machine speed	k rpm
Р	power	W
Р	real power	W
p	number of machine pole-pairs	

$P_{CU}$	copper loss	W
$P_d$	damping resistor power loss	W
$P_{IFW}$	machine iron, friction and windage loss	W
P <sub>inv in</sub>	total inverter input power	W
$P_{inv}$	inverter output power	W
$P_{in}$	input power	W
$P_{loss}$	$\mathrm{SMR} /\mathrm{generator}$ power loss	W
$P_L$	machine power loss	W
$P_{SMR}$	SMR output power	W
$P_{sw}$	switching power loss	W
$P_T$	wind turbine power	W
$P_W$	wind power	W
pk - pk	peak to peak	
Q	quality factor	
Q	reactive power	VAr
$Q_C$	capacitive reactive power	VAr
$Q_L$	inductive reactive power	VAr
R	resistance	Ω
r	blade radius	m
$R_1$	transformer primary resistance	Ω
$R_2$	transformer secondary resistance	$\Omega$
$R_{cold}$	cold resistance	$\Omega$
$R_d$	damping resistance	$\Omega$

$R_{eq}$	equivalent resistance	Ω
$R_{hot}$	hot resistance	Ω
$R_L$	load resistance	Ω
$R_{ph}$	phase resistance	Ω
$R_s$	stator resistance	Ω
rect(t)	normalised time-varying rectifier ripple	pu
S	apparent power	VA
S	number of stator slots	
S	$j\omega$	
Т	torque	Nm
t	time	s
$t_{off}$	device <i>turn-off</i> time	s
$t_{on}$	device <i>turn-on</i> time	S
$t_q$	thyristor turn-off time	s
V	voltage	V
v	wind speed	m/s
$V_C$	capacitor voltage	V
$V_{DC}$	DC link voltage	V
$V_{DC}$	DC voltage	V
$v_{eq}$	turbine equivalent wind speed	m/s
$V_{g\ pk}$	peak grid voltage	V
$V_g$	grid voltage	V
$v_g(t)$	normalised time-varying grid voltage	pu

$v_i$	internal wind tunnel wind speed	m/s
$V_L$	line voltage	V
$V_{ph\ pk}$	generator phase peak voltage	V
$V_{R \ pk \ OC}$	peak rectifier voltage	V
$v_r$	rated wind speed	m/s
$v_R(t)$	normalised time-varying rectifier voltage	pu
$v_{ws}(t)$	normalised time-varying current wave-shaper voltage	pu
X	reactance	Ω
$X_{ph}$	phase reactance	Ω
$X_s$	stator reactance	$\Omega$
$Z_{0n}$	normalised characteristic impedance	pu
$Z_0$	characteristic impedance	Ω
$Z_s$	stator impedance	Ω

#### Acronyms

AC	alternating current
AS	Australian Standard
CCS	constant current source
$\mathcal{CM}$	control modes
CSI	current-source inverter
CWS	current wave-shaper
DC	direct current
DCC	duty-cycle command
CCS CM CSI CWS DC DCC	constant current source control modes current-source inverter current wave-shaper direct current duty-cycle command

DFT	discrete Fourier transform
ESR	equivalent series resistance
F&P	Fisher & Paykel <sup>®</sup>
FC	filter configuration
FFT	fast Fourier transform
GC	grid connected
GCI	grid-connected inverter
HF	high-frequency
IFW	iron, friction and windage
IPM	interior permanent magnet
IR	International Rectifier <sup>®</sup>
ISA	integrated starter alternator
LA	Lundell alternator
LF	line-frequency
MPPT	maximum power point tracker
NEG	net energy gain
NICS	non-ideal current source
OC	open circuit
РМ	permanent magnet
pu	per-unit
PV	photovoltaic
PWM	pulse-width modulation
RMS	root-mean-squared

revolutions per minute rpm  $\mathbf{RR}$ rectifier ripple SCshort circuit  $\mathbf{SC}$ squirrel cage  $\mathbf{SG}$ synchronous generator SMR switched-mode rectifier  $\operatorname{SPM}$ surface permanent magnet THD total harmonic distortion TLtransformerless TSR tip-speed ratio UCG uncontrolled generation VSI voltage-source inverter WF wound field WR wound rotor

#### Abbreviations

CLcapacitive-inductiveI-Vcurrent vs. voltageLCinductive-capacitiveLCLinductive-capacitive-inductiveP-Vpower vs. voltageRLCresistive-inductive-capacitiveSPPslots per phase per pole

SW switch

THY thyristor

# List of Figures

1.1	Breakdown of worldwide electricity production for 2005	3
1.2	Typical 3-bladed wind turbine coefficient of performance	7
1.3	Coefficient of performance of various wind turbine rotors	8
1.4	Turbine power simulation	9
1.5	Turbine power, $c_p$ , speed and torque simulation $\ldots \ldots \ldots \ldots \ldots \ldots$	10
1.6	Furling concept demonstration	13
1.7	Comparison of early and modern small-scale turbine topologies used for	
	battery charging	14
1.8	Current vs. voltage locus of low and high-inductance PM generators	17
1.9	Comparison of DC power converter equivalent circuits	19
1.10	Rectifier circuit simplified equivalent modelling	21
1.11	Rectifier phasor diagram for various generator speeds	22
1.12	Generator voltage, current and power plot	23
1.13	SMR phasor diagrams for various generator speeds	24
1.14	SMR current and power plot	25
1.15	Simplified machine phase model under inverter operation	26
1.16	Inverter phasor diagram for various generator speeds	26
1.17	SMR current and power plot	27
1.18	Comparison of power converter output powers	28
1.19	Cost comparison of standalone and grid-connected inverters $\ldots \ldots \ldots$	29
1.20	Line-commutated current-source inverter topology	30
1.21	Current-source inverter with active compensation and passive filters $\ldots$ .	31
1.22	Comparison of voltage-source and current-source inverters	32
1.23	Comparison of bipolar and unipolar PWM output voltages / currents $\ \ . \ .$	33
1.24	Three-phase voltage-source inverter	37
1.25	Voltage-source inverter topology with reverse-blocking diode	38
1.26	Voltage-source inverter with boost converter	38

1.27	Back-to-back PWM voltage-source inverter	39
1.28	Hybrid voltage and current source inverter	39
1.29	Three-phase current-source inverter circuit	40
1.30	Line-commutated H-bridge inverter	41
1.31	Actively-commutated PWM current-source inverter	41
1.32	Soft-switching single-phase current-source inverter	42
1.33	Proposed low-cost grid-connected inverter	43
1.34	Thesis structure	46
2.1	Fisher & Paykel surface PM machine	52
2.2	Comparison of interior and surface PM machines	53
2.3	Simplified PM machine model	54
2.4	I-V loci of surface PM, DC and interior PM machines	55
2.5	Operating regions of high and low-inductance PM machines	56
2.6	Realistic phase and equivalent delta connected model, and simplified ma-	
	chine representation	57
2.7	Effect of varying stator resistance on PM machine voltage loci	58
2.8	PM and DC machine test arrangement	59
2.9	PM machine load arrangement	60
2.10	F&P surface PM machine open-circuit characteristic	61
2.11	Generator open-circuit voltage waveform at 1000 rpm	61
2.12	F&P surface PM machine short-circuit characteristic	62
2.13	F&P surface PM machine inductance vs. phase current and generator speed	63
2.14	F&P machine stator temperature increase	64
2.15	Machine open and short-circuit losses	65
2.16	PM generator phase model	67
2.17	Normalised PM machine line (AC) voltage loci	69
2.18	$\operatorname{PSIM}^{\textcircled{\text{\tiny (B)}}}$ delta-connected PM machine model with AC and DC resistive loads	70
2.19	Normalised PM machine DC voltage loci	71
2.20	Peak normalised DC power vs. generator speed prediction, comparing the	
	analytical and $\mathrm{PSIM}^{\mathbb{R}}$ models $\ldots \ldots \ldots$	72
2.21	AC and DC I-V and P-V loci, comparing ideal analytical and $\operatorname{PSIM}^\circledast$ model	73
2.22	PM machine load arrangement	74
2.23	PM machine AC voltage loci	75
2.24	Measured generator efficiency vs. output power	77

2.25	PM machine DC voltage loci
2.26	Maximum DC output power vs. generator speed
2.27	DC to RMS line voltage ratio
2.28	Measured generator and rectifier efficiency vs. DC output power 81
3.1	SMR equivalent circuit
3.2	Ideal rectifier and SMR voltage and current vs. duty-cycle
3.3	PSIM <sup>®</sup> model for the switched-mode rectifier
3.4	Voltage, current and power of generator for various load voltages 89
3.5	SMR operating regions
3.6	SMR internal components and control circuitry
3.7	SMR test arrangement
3.8	Rectifier output voltage vs. duty-cycle for various generator speeds 93
3.9	Rectifier output current vs. duty-cycle for various machine speeds 94
3.10	SMR output current vs. duty-cycle for various generator speeds $\ldots \ldots \ldots 95$
3.11	SMR output power vs. duty-cycle for various machine test speeds $\ldots$ $\ldots$ 96
3.12	Measured torque vs. duty-cycle for various generator speeds
3.13	Duty-cycle corresponding to maximum SMR power and generator torque . 97
3.14	Measured SMR efficiency vs. SMR output power for various machine speeds 98
3.15	Total efficiency vs. SMR output power characteristic for various generator
	speeds
3.16	Wind tunnel test arrangement
3.17	Equivalent wind speed calculation at blade sweep area
3.18	Estimated coefficient of performance vs. tip-speed ratio curve
3.19	Estimated equivalent vs. wind tunnel wind speed curve
3.20	Estimated turbine power vs. speed
3.21	Estimated turbine torque vs. speed
3.22	SMR output power vs. duty-cycle
3.23	Turbine speed vs. duty-cycle
3.24	Measured duty-cycle vs. wind speed comparison for both control modes 108
3.25	Turbine and SMR operating characteristics for both control modes 109
3.26	Estimated SMR and generator efficiency vs. SMR output power 110
4.1	Simple block diagram of a grid-connected wind turbine, and inverter output
	current
4.2	Lagging and leading power factors

4.3	Overview of proposed grid-connected inverter	122
4.4	Circuit diagram and input and output current for the constant current source	123
4.5	Current wave-shaper circuit	124
4.6	Comparison of SMR circuit used for the DC-DC converter, and the modified	
	SMR circuit used for the inverter	125
4.7	Comparison of constant and time-varying SMR duty-cycle	125
4.8	Unfolding circuit, and input and output currents	126
4.9	Inverter low-pass filter circuit, and input and output currents	127
4.10	Preliminary resistive / capacitive load test arrangements	128
4.11	Grid-connected inverter test arrangement	129
4.12	Grid-connected inverter test arrangement equivalent circuit	130
4.13	Measured transformer equivalent inductance and resistance	130
4.14	Photograph of an outer-rotor PM generator	131
4.15	I-V locus of PM generator	132
4.16	Photograph of power electronic components of the grid-connected inverter .	133
4.17	Microcontroller hardware, and software flow chart	134
4.18	PSIM <sup>®</sup> grid-connected inverter model	135
4.19	Preliminary PSIM <sup>®</sup> simulation proving the grid-connected inverter concept	137
4.20	Simulated effect of modulation index variation on inverter operation	137
4.21	Proof of inverter concept using an ideal current source	139
4.22	Proof of inverter concept using PM generator as current source	140
4.23	Effect of load resistance and generator speed on I-V locus	142
4.24	Inverter input and output currents for resistive load case 1	142
4.25	Inverter input and output currents for resistive load case 2	143
4.26	Inverter input and output currents for resistive load case 3	144
4.27	Inverter output current and voltage of intermediate grid-connected case	145
4.28	Inverter output current and voltage for the pure grid-connected case	146
4.29	Grid-connected inverter output current for various grid voltages	147
4.30	Filter capacitance vs. grid voltage	148
4.31	Resonant frequency and quality factor vs. grid voltage	149
4.32	Inverter output current THD vs. grid voltage	149
4.33	Grid-connected inverter output current for various modulation indices	150
4.34	Grid-connected inverter output current and voltage for various modulation	
	indices	151
4.35	Inverter output current THD vs. modulation index	152

5.1	Normalised ideal and non-ideal inverter input and output currents	156
5.2	Output voltage of various inverter stages	158
5.3	Normalised ideal and experimental generator I-V loci	158
5.4	Derivation of normalised ideal inverter input current	159
5.5	Normalised inverter input and output currents for various values of $\hat{\alpha}$ , using	
	the ideal and experimental I-V loci	161
5.6	Effect of rectifier on normalised inverter input and output current	161
5.7	Effect of rectifier and fluctuating output power on the normalised ideal	
	inverter input and output currents	162
5.8	Normalised inverter input and output currents for two cases of $\widehat{\alpha}$ equal to	
	0.2pu	163
5.9	Effect of the fluctuating input power on the inverter output current and	
	the harmonics for the ideal rectifier I-V locus	165
5.10	Effect of the fluctuating input power on the inverter output current and	
	the harmonics for the experimental rectifier I-V locus	165
5.11	THD vs. $\hat{\alpha}$ comparison using the ideal and experiential locus	166
5.12	Output current distortion and FFT analysis, caused by the rectifier ripple .	167
5.13	Output current distortion and FFT analysis, caused by the rectifier ripple	
	and fluctuating input power	167
5.14	Output current THD vs. modulation index for various values of $\hat{\alpha}$ , using	
	the ideal and experimental I-V loci	168
5.15	THD vs. $\hat{\alpha}$ comparison using the ideal and experiential locus	169
5.16	Turbine speed and the resulting $\hat{\alpha}_0$ vs. wind speed	170
5.17	Modulation index and the resulting $\hat{\alpha}$ vs. wind speed $\ldots \ldots \ldots \ldots \ldots$	170
5.18	Open-loop output current THD vs. wind pseed	171
5.19	Bipolar and Unipolar PWM waveforms	172
5.20	Harmonic spectra of the bipolar and unipolar PWM waveforms $\ \ldots \ \ldots$	172
5.21	Distortion, fundamental magnitude and THD vs. modulation index for	
	unipolar PWM waveform	173
5.22	Effect of the rectifier and fluctuating input power on the inverter input and	
	output currents	175
5.23	Harmonic spectrum of the inverter output current waveform	175
5.24	THD vs. modulation index for various values of $\widehat{\alpha}_0$	176
5.25	CL type low-pass filter	177
5.26	Normalised low-pass filter frequency response	178

5.27	Notch filter circuit	179
5.28	Notch filter frequency response	179
5.29	Typical voltage and current source inverter low-pass filter configurations .	182
5.30	CL filter damping, using resistance	184
5.31	Comparison of damped CL low-pass filter gain and phase margins, for two	
	damping resistances	185
5.32	LC filter gain and phase margin at the resonant frequency, for various	
	resistances	186
5.33	Filter delay at the fundamental frequency vs. quality factor for various	
	cutoff frequencies	187
5.34	Cutoff frequency and characteristic impedance plots vs. normalised induc-	
	tance and capacitance	190
5.35	Harmonic spectrum of unipolar PWM waveform and filter gains of each	
	damped low-pass filter	191
5.36	Unipolar PWM harmonic spectrum, showing the effect of varying the cutoff	
	frequency on the filter gain	192
5.37	Unipolar PWM harmonic spectrum, showing the effect of varying the filter	
	quality factor on the filter gain	193
5.38	THD vs. filter cutoff frequency and quality factor	194
5.39	Real, reactive and apparent power triangle corresponding to the grid power	
	factor limits	195
5.40	Real and reactive power flow diagram of the inverter filter	195
5.41	Low-pass filter capacitance and inductance design region $\ldots \ldots \ldots \ldots$	198
5.42	Damping resistor current determined by theory of superposition	199
5.43	Filter power loss vs. output current THD, for various cutoff frequencies	202
5.44	Design Regions of each filter configuration, considering THD and power los	s203
5.45	THD and power loss contours within the design region of each filter type $% \left( {{{\rm{D}}_{{\rm{D}}}}_{{\rm{D}}}} \right)$ .	204
5.46	THD vs. cutoff frequency and quality factor, showing the effects of fluctu-	
	ating output power and rectifier ripple	209
5.47	Third-order damped low-pass filters	210
5.48	Inverter input and compensated output current, showing concept of feed-	
	forward current compensation	212
5.49	Comparison of modulation index, and input and output currents for the	
	uncompensated and feed-forward compensation schemes	214
5.50	Approximated experimental I-V locus	215
5.51	Comparison of modulation index, and input and output currents for the	
------	---	
	uncompensated and feed-forward compensation schemes	
5.52	Compensated output current THD vs. rectifier ripple phase angle error $~$ . $~$ 218	
5.53	Summary of proposed feed-forward controller	
5.54	Comparison of modulation index, and input and output currents for the	
	uncompensated and feed-forward compensation schemes	
5.55	Inverter output current THD and fundamental magnitude vs. $\widehat{\alpha}_0$ using the	
	open-loop and both feed-forward control algorithms	
5.56	Comparison of inverter output current and power vs. wind speed, using	
	both open-loop and feed-forward control algorithms	
6.1	Turbine $c_p$ curve used to calculate turbine operating speed $\ldots \ldots \ldots \ldots 228$	
6.2	Inverter input and peak compensated output current	
6.3	Designed machine I-V locus	
6.4	Quality factor, THD, power loss and cutoff frequency tradeoff curves, used	
	to design the 1kW grid-connected inverter	
6.5	Simulated inverter output currents and voltage using an ideal current source	
	and open-loop control	
6.6	THD prediction using the PM generator current source	
6.7	Simulated inverter currents using the PM generator current source and	
	open-loop control	
6.8	Relevant part of PSIM® circuit showing feed-forward control implementation 242	
6.9	Feed-forward control proof of concept	
6.10	Comparison of feed-forward (compensated) and open-loop inverter output	
	currents, for 4 kHz switching case	
6.11	Comparison of compensated and open-loop inverter output currents, for	
	$10 \mathrm{kHz}$ switching case	
6.12	Apparent power vs. current command, for power less than rated 246	
6.13	Inverter power factor vs. apparent power, for power less than rated $\ldots$ 247	
6.14	Inverter output current THD vs. apparent power, for power less than rated 247	
6.15	Inverter input and output currents, for desired power greater than rated $248$	
6.16	Inverter control signals and output current, for power greater than rated $~$ . 249	
6.17	Apparent power vs. current command for wide power range	
6.18	Inverter power factor vs. entire apparent power range	
6.19	Output current THD vs. entire range of apparent power	

6.20	Wind, turbine, generator and inverter output power vs. wind speed char-
	acteristic
6.21	Turbine and generator operating characteristics vs. wind speed
6.22	Generator $\hat{\alpha}$ and $\check{\beta}$ vs. wind speed
6.23	Current command vs. wind speed for both control modes
6.24	Inverter simulation using control mode 1, for various wind speeds 257
6.25	Inverter simulation using control mode 2, for wind speeds beyond rated $\ . \ . \ 258$
6.26	Comparison of inverter apparent power and compensated current THD
	vs. wind speed characteristic, using both control modes $\ldots \ldots \ldots \ldots \ldots 259$
6.27	Inverter power loss break-down
6.28	Inverter efficiency breakdown vs. output power
A 1	
A.1	Principle of sinusoidal pulse-width modulation
A.2	Harmonic elimination PWM control signal
A.3	Principle of current hysteresis control scheme
A.4	Filter output current THD vs. power loss vs. cut-off frequency

# List of Tables

1.1	Net energy gain comparison of various renewable and non-renewable energy
	sources
1.2	Comparison of small and large scale wind turbine properties
1.3	Generator and power converter summary, used for automotive and wind
	power generation $\ldots \ldots 18$
1.4	Comparison of small-scale wind turbine generator DC power converters $\ . \ . \ 20$
1.5	Comparison of transformer and transformer less inverter properties $36$
1.6	Comparison of inverter properties for various transformer topologies 36
2.1	Physical, measured and calculated PM generator properties
3.1	SMR component properties
4.1	Current harmonic limits of Australian Standard 4777.2
4.2	Measured PM generator properties
4.3	Semiconductor properties
4.4	Simulated and measured resistive load inverter performance $\ldots \ldots \ldots 141$
4.5	Simulated and measured inverter performance for cases 1–3
5.1	Unipolar PWM waveform harmonic analysis, for various modulation indices 174
5.2	Low-pass filter property comparison, for four damping resistor locations $187$
5.3	Low-pass filter variable vs. parameter
5.4	Summary of apparent, real and reactive power, under grid power factor
	requirement extreme cases
5.5	Comparison of damping resistor current, for various filter configurations 200 $$
5.6	Effect of varying $C_n$ on filter design region $\ldots \ldots 206$
5.7	Effect of varying $f_{sw}$ on filter design region
6.1	Summary of calculated turbine properties for the proposed 1kW GCI $\ .$ 228

6.2	Designed generator properties
6.3	Filter base quantities
6.4	Filter component values for the proposed 1kW system
6.5	Simulated inverter performance using an ideal current source
6.6	Simulated inverter performance using an the PM generator current source . $240$
6.7	Current compensated inverter performance using the PM generator and
	rectifier current source
6.8	Comparison of open-loop and compensated inverter performance 245
6.9	Optimised inverter semiconductor properties
6.10	MOSFET vs. IGBT turn on and off times, and maximum * calculated losses $256$

## Chapter 1

## Introduction

This chapter introduces wind energy and justifies it as a competitive renewable and nonpolluting energy alternative to fossil fuels. Its utilisation, recent growth and technological advances are briefly discussed. The principles of wind energy conversion, along with smallscale turbines and their generators are examined. Common standalone and grid-connected power converters are discussed, and a novel low-cost grid-connected inverter topology is proposed. The chapter concludes with the aims of the research project, discusses its original contributions and describes the structure of the thesis.

## 1.1 Wind Energy

Wind is defined as the movement of air from regions of high pressure to regions of low pressure, and is primarily caused by uneven heating of the Earth's surface, where an estimated 1% of the Sun's energy is converted to wind [1]. The second cause of large-scale air movement, also referred to as *atmospheric circulation*, is the rotation of the planet or the *Coriolis* effect [1–3].

The definition of wind (moving mass), implies it has an associated kinetic energy. This is converted to mechanical energy using a wind mill or wind turbine, which slows the passing wind; the energy conversion takes place by virtue of the law of *conservation* of energy. The instantaneous use of the wind's energy is referred to as *wind power*.

#### Wind Power Utilisation

Wind power has been used for over three thousand years [4, 5], firstly by the Persians and more recently the Europeans and Americans in the mid  $13^{th}$  and  $19^{th}$  centuries, respectively. It has been used for transportation, e.g. ship propulsion, and more recently mechanical applications such as milling grains, sawing wood and pumping water etc. [2, 3, 6]. It was not until the advent of the electrical machine, in 1888, that wind power was used to generate electricity [6]. Note that the term *windmill* refers to a device that extracts wind energy for mechanical applications, whilst a *wind turbine* is used to generate electricity.

#### **Recent Growth**

Global wind power has been growing at an average rate of 28% per annum over the last decade [7] in response to the growing demands for electricity. Worldwide electricity production has been steadily increasing at a rate of 3.4% per year on average between 1973 and 2002 [8], as a result of population growth and the industrialisation of developing countries.

## 1.1.1 Electricity Usage and Conventional Generation

As it is well known, electricity is a form of energy that is widely used in domestic, commercial and industrial applications. It can be easily converted to other forms of energy including mechanical, thermal, light and chemical and is readily and effectively transported over long distances by transmission lines.

Figure 1.1 gives a breakdown of the sources of energy from which electricity was generated in 2005 [8]. It is seen that approximately two thirds of the world's electricity is generated by fossil fuels, such as coal, oil and gas. This causes two serious issues:

- 1. Environmental impacts fossil fuel combustion releases gases that contribute to pollution and impact the environment, wildlife and human health. These gases include sulphur dioxide, carbon dioxide and nitrous oxides, which are responsible for acid rain, global warming and photochemical smog, respectively.
- 2. Depleting reserves fossil fuels, such as coal, oil and gas, are formed in the ground over long periods of time by chemical and physical changes to plant matter under conditions of high temperature and pressure. Currently these fuels are being consumed at a much greater rate than they are being formed.

The combination of increasing electricity demands, depleting fossil fuel reserves and environmental issues, has created an urgent need to investigate non-polluting and renewable energy sources to generate electricity to ensure a sustainable future. NOTE: This figure is included on page 3 of the print copy of the thesis held in the University of Adelaide Library.

Figure 1.1: Breakdown of worldwide electricity production for 2005 [8].

#### 1.1.2 Alternative Energy Sources

Figure 1.1 showed that after fossil fuels, the next two largest sources of energy used for electricity generation are hydro schemes and nuclear fission, which each make up almost the remaining third of the total. Hydro schemes are non-polluting and renewable, however, requires large-scale water storage in dams and has limited potential for further utilisation. Nuclear power has been used for many decades but there is still significant public concern about its safety and possible long term pollution issues after incidents such as *Three-Mile Island* (1979) and *Chernobyl* (1985).

Of the remaining alternative energy sources, i.e. wind, photovoltaic, biomass etc., large-scale (> 250kW) wind power is one of the most economical. The majority of research effort has been placed on developing large-scale wind turbines, which has resulted in the cost of wind-generated electricity decreasing rapidly over the last 20 years. At present, it is comparable in price to that of coal-fired power [9]. This is mainly attributed to government incentives, improvements to aerodynamic efficiency, the evolution of power semiconductors and control methods, and improved grid integration [9,10].

#### Net Energy Gain

In addition to low cost, wind power offers the lowest *net energy gain* (NEG) period, compared to other renewable sources of energy. This period is defined as the operating time required to regain all the energy used to construct the electric power generation

source. The NEG of large wind turbines is in the order of months [11], compared to photovoltaic (PV) cells, which is in the order of years. This, is shown in Table 1.1 along with the NEG of various PV materials and other fossil-fuel based power generation. Note that the NEG period for both wind and PV cells depend on operating conditions, i.e. average wind speed and solar irradiance, respectively. Hence, as listed in the table, careful site location or selection of technology will reduce the NEG period of wind and photovoltaic systems, respectively.

Table 1.1: Comparison of net energy gain (NEG) periods of conventional and renewable energy sources [11].

NOTE: This table is included on page 4 of the print copy of the thesis held in the University of Adelaide Library.

Despite the low cost and short NEG period of large-scale wind turbines, small-scale wind energy systems are more expensive and have longer NEG periods, making them less competitive with conventionally derived electricity [4]. This is due to the majority of research focussing on large-scale wind turbine optimisation, as small blade efficiency improvements would significantly increase the turbine output power, whilst an increased blade efficiency will only slightly increase the output power of a small-scale wind turbine.

## 1.1.3 Large and Small-Scale Turbine Classification

The term *Wind Power* often manifests an image of large wind turbines arranged together to form a wind farm. These large-scale turbines are typically rated in the order of MW, and account for the majority of recent worldwide wind power growth.

Small-scale wind turbines are mainly used to generate electricity in remote areas. They differ from their large-scale counterparts in physical size, power rating, and by the types of generators, speed limiters and power converters used. The key differences between small and large-scale wind turbines are summarised in Table 1.2.

	Small Scale	Large Scale		
Blade Length	< 5m	20–50 m		
Rated Power	$< 100 \mathrm{kW}$	1–3 MW		
Turbine Speed	several hundred rpm	15–30 rpm		
Transmission Type	Directly Driven	Gearbox		
Concretor Types	DC,	Induction (SC and WR),		
Generator Types	PM synchronous	Synchronous (WF and PM)		
	Furling,	Blade pitch,		
Speed Control	Blade stall,	Rotor resistance (WR IM),		
	Power electronics control	Power electronics control		
Applications	Standalone,	Grid-connected		
Applications	Grid-connected			
Stater Crid Connected	via full-scale grid-	Directly,		
Stator Grid Connected	connected inverter	via full-scale converter		
Rotor Grid Connected	-	via partial-scale converter		

Table 1.2: Comparison of small and large scale wind turbine properties.

Note: SC, WR, PM and WF denote, squirrel cage, wound rotor, permanent magnet and wound field, respectively.

## 1.1.4 Small-Scale Turbine Development

In 1925, Marcelleus and Jacobs began working on the first high-speed, small-sized, affordable battery-charging wind turbines [12]. By the 1930's many Midwest manufacturers built wind-chargers, small battery-charging wind turbines that often provided the only source of electricity for many homesteads [13]; these were used to power lights and electric motors [14]. Their demise, however, was brought about in the late 1930's when the Rural Electric Association began spreading AC transmission lines across rural America [14]. Small DC wind turbines had no economical way of either interconnecting to the AC grid or supplying power for the many new appliances that began to fill farm households [12].

Interest in wind power was renewed in the early 1970's, as a direct result of the oil crisis [4, 13, 14]. Since then, a new generation of small-scale wind turbines appeared, i.e. DC generators were replaced with permanent magnet (PM) generators. Induction generators were also examined in the 1980's, and are now generally used for turbines rated higher than 10kW.

#### 1.1.5 Technology Improvement

Wind turbine technologies have significantly improved since the 1980's, allowing turbine size, efficiency, and ease of installation to increase. Currently the largest generator is about 100 times the output power of those used in 1980, and the rotor diameters have increased 8-fold [15]. In addition, turbine efficiency has increased due to improved blade design, and modern turbines are modular and hence quick and easy to install.

#### 1.1.6 Market Growth

The sales of both large and small-scale turbines has experienced substantial growth in recent years. The worldwide wind power market (which is predominantly large-scale) grew by 40% in 2005, bringing the cumulative installed capacity to about 59GW [7,16] which meets the electricity needs of more than 25 million households worldwide. Global wind power has been growing at an average rate of 28% per annum, over the last 10 years, and grew by 46% in 2007, with US \$9 billion invested in wind plants alone [17]. The installed capacity is expected to reach 160GW by 2012 [7]. The USA is claimed to be the fastest-growing wind power market for the last three years. Wind power accounted for 35% of all new US electricity generation capacity in 2007 [17].

The growth in small-scale wind turbines is more difficult to quantify, however, it is hence estimated that the small-scale turbine market is growing at roughly 40% per year [18]. It is also estimated that the US alone will sell a total of US \$25M-\$55M worth of such turbines by 2010 [19], depending on the market and other factors such as government rebates, tax incentives etc. Within this market there is an increasing demand for low-cost grid-connected wind turbines, and it is expected that by 2020 small-scale turbines will provide 3% of the US electrical energy consumption [18].

## **1.2** Principles of Wind Power

#### 1.2.1 Wind and Turbine Power

The power available in the wind,  $P_W$ , is directly proportional to the air density,  $\rho$ , the cube of the wind speed, v, and the blade sweep area  $\pi r^2$ , where r is the blade radius, as summarised by Equation (1.1). The turbine co-efficient of performance,  $c_p$ , represents the efficiency of the turbine in converting the power in the wind to mechanical power (turbine power),  $P_T$ , as shown in Equation (1.2). The  $c_p$  is a function of the tip-speed ratio,  $\lambda$ , where  $\lambda$  is the ratio of blade linear tip velocity and wind speed, summarised by Equations (1.3) and (1.4), where  $\omega$  is the turbine angular velocity.

$$P_W = \frac{1}{2} \rho \,\pi \, r^2 \, v^3 \tag{1.1}$$

$$P_T = c_P P_W \tag{1.2}$$

$$c_p = f\left(\lambda\right) \tag{1.3}$$

$$\lambda = \frac{\omega r}{v} \tag{1.4}$$

The  $c_p$  typically has a maximum value of 40–45%, as shown in Figure 1.2, which shows a typical three-bladed turbine  $c_p$  characteristic. The dashed line represents the theoretical peak  $c_p$  of 59%, known as *Betz' limit*, after its discovery by Albert Betz in 1919 [20].

> NOTE: This figure is included on page 7 of the print copy of the thesis held in the University of Adelaide Library.

Figure 1.2: Typical coefficient of performance,  $c_p$ , vs. tip-speed ratio,  $\lambda$ , curve of threebladed wind turbines [21]. The dashed line represents the theoretical peak  $c_p$  of 59%.

### **1.2.2** Coefficient of Performance

The turbine coefficient of performance vs. tip-speed ratio (TSR) characteristic allows a turbine's power characteristic to be predicted, for any wind speed. A comparison of various wind turbines and their respective  $c_p$  characteristics is shown in Figure 1.3 [22]. The figure shows that turbines with *high solidity* (total blade area) reach their peak  $c_p$  at low values of  $\lambda$ , as seen for the American, Dutch and Savonius turbines. These turbines hence have high starting torques [4]. In contrast, low solidity turbines, such as the two and three-bladed turbines, have a higher peak  $c_p$ , which occurs for higher TSR values, and hence produce low starting torque.



Figure 1.3: Comparison of various turbine coefficient of performance vs. tip-speed ratio characteristics [22].

High optimal TSR turbines are more difficult to start in low wind speeds, due to their low starting torque, and have the potential to produce unwanted acoustic noise. The starting problem can be solved by initially driving the turbine up to operational speed. The Darrieus rotor, for example, does not produce torque at standstill, and needs to be driven up to a rotational speed where aerodynamic forces take over [23].

## 1.2.3 Principles of Turbine Operation

#### **Turbine Operating Regions**

Modern small-scale and large-scale wind turbines use variable-speed operation, due to the benefits compared to constant-speed operation, such as higher energy yield (about 10%), reduced noise and reduced mechanical stress [10, 12, 24]. More energy is extracted from a variable-speed turbine as the turbine  $c_p$ , and hence power, is maximised by adjusting its speed in accordance with wind speed (see Equation (1.4)). Turbines operate using maximum power point tracking (MPPT) below rated wind speed, generally 12m/s. As the wind speed increases beyond rated, the turbine usually maintains constant output power until the shut-down wind speed. The shut-down speed is typically 25m/s (about twice that of rated speed) and the turbine is stopped for safety reasons at this wind speed.

#### **Turbine Power**

Figure 1.4 shows the simulated power vs. turbine and wind speed characteristic of a turbine with a blade radius of 1m and with the  $c_p$  curve shown in Figure 1.2. The dashed lines show that turbine power is maximised for a given wind speed by adjusting the turbine speed. The solid black lines represent the turbine operating range and show that power is maximised below rated wind speed (12m/s), whilst power is maintained at rated power (1.5kW) for high wind speeds, and that the turbine is stopped at 24m/s.



Figure 1.4: Simulated typical turbine power vs. (a) turbine, and (b) wind speed characteristic. The solid black lines show the power maximisation (MPPT) and constant power modes of operation, i.e. below and above rated wind speed (12m/s), respectively.

The above figures indicate that power is maximised below rated wind speed and maintained constant above rated wind speed by increasing and decreasing turbine speed, respectively. The latter reduces the turbine  $c_p$  and increases torque. This operating state is summarised by Figure 1.5, which shows the turbine's response to both MPPT and constant power operation. In practise, the turbine torque can be increased by controlling the generator output current via a power converter. This principle is experimentally verified later in Section 3.3.3, using a DC-DC converter, and shown by simulation in Section 6.4.1 using a grid-connected inverter.



Figure 1.5: Simulated turbine (a) power, (b)  $c_p$ , (c) speed, and (d) torque response to power maximisation and constant power modes below and above rated wind speed, respectively. The vertical dashed lines represent rated wind speed.

## 1.3 Small-Scale Wind Turbines

## 1.3.1 Applications

Small-scale wind turbines are used to generate electricity in one of three systems, these include standalone, grid-connected and micro-grid systems. Each system is discussed below:

## Standalone System

Early small-scale wind turbines were predominantly used to generate electricity in remote regions or properties such as farms or ranches, and are currently used in remote areas to provide power for communication systems etc. [4]. These are known as *standalone* systems, as the electrical power is generated without the presence of a utility grid. The advantage is that remote properties can operate from locally-generated electricity without incurring the high cost of extending the power grid to the property. The drawback is that the wind energy must be stored, for periods of low or null wind speeds. Although several energy storage options, such as thermal storage, inertia storage, storage by compressed air, hydrogen storage or electrical accumulators (batteries) exist [3], batteries are most commonly used.

## Grid-Connected System

Small-scale wind turbines are also used in regions where utility power lines provide electricity. The electricity generated by the turbine is directly fed to the utility via a *gridconnected* inverter (GCI). The main advantages of this arrangement are that energy storage, e.g. bulky deep-cycle battery, is avoided, and that utility operators purchase excess power fed to the grid. Despite the advantages, the cost of a GCI is about the same as a deep-cycle battery and its required charger; based on data obtained from a renewable energy technology supplier catalogue [25].

## Micro-Grid System

A micro grid is a small-scale power supply network that is designed to provide energy for a small community, such as a housing estate, isolated rural community or municipal region etc. [26]. These operate in a similar manner to large-scale power systems, i.e. micro grids aim to match the power with the real-time demand without energy storage. The main

challenges of micro grids are power system and frequency control issues. Micro grids are beyond the scope of this work.

## 1.3.2 Turbine Properties

Small-scale wind turbines use short fixed-pitch blades (up to a few metres), which operate at relatively high turbine speeds, due to their small radius (see Equation (1.4)). This high rotational speed avoids the need for a gearbox, which reduces maintenance and costs, and also improves reliability and efficiency at low wind speeds [27]. Small-scale wind turbines operate at variable speed to maximise the  $c_p$  and hence output power. The turbine speed needs to be limited under high wind speeds to avoid high stress on components [28], which may damage the turbine and generators. The turbine speed is limited by using *over-speed protection* mechanisms or *speed-limiting* devices.

#### **Turbine Speed Limiters**

Early small wind turbines that were used for simple battery charging applications used DC generators, and have some form of natural speed control. Although the induced voltage is proportional to the generator speed, the generator output voltage is set by the battery voltage, and hence the machine output current increases with wind speed. The torque is proportional to the machine output current and hence the generator speed is set by a balance of the wind turbine and electrical torque [6].

Alternative forms of turbine speed control include *furling* and *stall control*. Furling uses a governor-flyball or spring arrangement [3], which yaws or tilts the turbine out of the direction of the incoming wind [29]. Furling is a self governing action that occurs once the wind exceeds rated speed; the concept is illustrated in Figure 1.6. The turbine is shown to be perpendicular to the incoming wind, for wind speed  $v_1$ , which corresponds to a wind speed less than rated. The turbine begins to furl as the wind speed increases to  $v_2$  and then  $v_3$ , increasing the axial force on the turbine and hence the tension and length of the spring. The turbine is substantially furled for wind speed  $v_3$ . Note that the figure illustrates the concept for horizontal furling, however, vertical furling also exists, where the turbine appears similar to a helicopter [4,13].

Stall control is another method that limits turbine speed and power under high wind speed conditions. This method, unlike furling, does not require mechanical components, as the blades are specifically designed to be less efficient at high wind speeds [2,5]. The



Figure 1.6: Furling action taken by small-scale wind turbines under increasing wind speeds,  $v_{1-3}$ . The turbine begins to furl for  $v_2$ , and is largely furled for  $v_3$ . Note that the vector lengths indicate relative wind speeds.

resulting turbine rotates slower than fixed-pitch blades under increasing wind speeds, which reduces the tip-speed ratio and hence  $c_p$ .

The drawback of both furling and stall control is the unwelcome acoustic noise [30], caused by the high tip-speed ratio and the angle of the passing air with respect to the blades. Hence, lower noise speed limiting mechanisms are required. One solution is to control the machine current and hence torque using a power converter; an alternative, which is applicable only to DC generators, is to adjusting the field excitation.

## 1.3.3 Generator Varieties

#### Standalone Wind Turbines

Small-scale turbines of the 1930's used shunt-excited DC generators and were used to charge batteries. The generator was directly driven and its output was connected to a battery via an automatic contactor that closed, allowing current to flow from the generator to the battery, once the wind speed and induced voltage exceeded that of the battery. This arrangement is shown in Figure 1.7(a), which also shows that loads, such as electric motors or lights, were connected to the battery via switches.

The main drawback with this arrangement is that the full power is fed to the battery via a carbon brush and commutator arrangement, which requires regular brush replacement [23]. A diode replaced the contactor with the advent of semiconductors, and PM DC generators replaced the wound-field DC generators. The advantage of such a generator is a constant excitation, without the need for field current and hence field winding copper losses. The drawback however, is the lack of excitation control and that full power is still

transmitted through carbon brushes. Permanent magnet synchronous generators offer the best solution, as carbon brushes are not required. These however, require a full-bridge diode rectifier, due to their three-phase AC output voltage / current. The circuits of the DC, PM DC, and PM synchronous generators used for battery charging, are shown in Figure 1.7.



(c) PM synchronous generator

Figure 1.7: Comparison of early and modern small-scale turbine topologies used for battery charging, including (a) wound-field DC, (b) PM DC, and (c) PM synchronous generators.

#### Grid-Connected Wind Turbines

During the late 1970s, small-scale wind turbines were first used for grid-connected power generation. Research was carried out in the U.S. where generators such as induction, synchronous, and Lundell automotive alternators were used [23]. Although synchronous machines also use a carbon brush arrangement, they were favoured over DC machines, as the carbon brushes are not replaced as often. This is due to the small amount of current, that required to excite the field, that is transmitted through the slip ring / carbon brushes arrangement.

Induction machines were widely used both in Denmark and in the US, as they are cheaper than synchronous machines. Induction machine production peaked in 1981 in the US, however, these moved away from small-scale to medium-scale wind turbines, as those used for small-scale applications were deemed a failure [13]. The drawback of directly-connected synchronous machines was the requirement to operate at synchronous speed, such that power was fed to the grid at the grid frequency.

The advantages of variable-speed operation were known, and two methods to integrate variable-speed wind turbines were examined in the literature. The first used a *rotary inverter*, which consisted of a variable-speed AC generator that controls a DC machine that is coupled to a second AC generator. The concept was to operate the grid-side generator at synchronous speed, by controlling the speed of the DC machine. The main drawback of this arrangement is the poor peak efficiency, of about 60% [23], which is due to the use of three electrical machines.

The second method investigated to grid-connect variable-speed generators, was the use of power converters. This will be further discussed in Section 1.5.

## 1.3.4 Current Trends

A survey of currently available small scale wind turbines is presented in [31]. This catalogue suggests that for wind turbines rated below about 10kW, PM synchronous generators are used, whilst induction machines are used for rated powers above 10kW. The catalogue also suggests that small wind turbines rated at 1kW or less are generally not grid connected, whereas turbines upward of 1kW generally are, through the use of power converters. The survey shows that small scale turbine power is controlled by microprocessors or battery regulators, and that speed is controlled by furling or stall control.

PM excitation is generally favoured in modern small-scale turbine designs since it allows for higher efficiency, and does not require external excitation current [32]. In addition, these are capable of variable speed operation, which is required to maximise aerodynamic efficiency. The drawback of these generators, however, is the lack of back-EMF voltage control, i.e. the back-EMF voltage is controlled by the turbine speed. The use of a three-phase PM generator and three-phase rectifier are nowadays common. The turbine operating speed is controlled by the generator torque [33]. This is achieved using a power converter.

#### **Power Converters**

Power converters are required for directly-driven permanent magnet (PM) synchronous generators, as the variable-frequency and magnitude AC output current/voltage (pro-

duced by variable-speed operation), is unsuitable for transmission, storage or use. Batteries and the majority of power converters require a constant DC voltage, which is easiest achieved using an uncontrolled or controlled rectifier. Rectifiers can either be used alone or in conjunction with an additional standalone or grid-connected power converter. Standalone and grid-connected power converters will be discussed in detail in Sections 1.4 and 1.5, respectively.

## Permanent Magnet Material

Small-scale PM generators tend to use rare-earth magnets, such as *Neodymium Iron Boron* magnets. These are preferred as they have a magnetic energy product up to 10 times greater than those of *ferrite* and other permanent magnets, such as the *Aluminium Nickel Cobalt*, varieties [34]. Rare-earth magnets offer high remanent flux densities and can be used to produce machines which achieve high efficiencies and have high (generator) power densities. These properties make rare-earth magnets the magnet of choice for for many applications, including direct drive wind turbines [34].

## 1.3.5 Inductance Classification of PM Generators

PM generators used in wind turbine applications may have low or high inductance. The majority of PM generators have short-circuit currents that are greater than the rated current which are referred to as *low-inductance* machines. These machines are well-suited to drive the voltage-source inverters which are commonly used in grid-connected inverters. Low-inductance machines require an oversized power converter in order to achieve a wide field-weakening range, and can produce high currents under fault conditions. High currents can also be produced during over-speed conditions. If a power converter, such as the boost converter, is used, the induced back-EMF voltage can exceed that of the DC link, which can produce high currents.

In contrast, some generators have such high values of inductance that their shortcircuit current at high speed equals their rated current [35]. These machines are referred to as *high-inductance* generators and can readily achieve a wide constant power speed range. Note that care needs to be taken not to open-circuit these machines at high speeds, due to their large back-EMF voltage.

A comparison of the current vs. voltage (I-V) locus of a conventional low-inductance and a high-inductance generator is shown in Figure 1.8. The figure shows the I-V characteristics of an ideal PM machine for speeds of n and 2n, and load resistance lines for high and low resistances. Note that the generator produces an open-circuit voltage that is proportional to its speed, whilst the short-circuit current is fixed for all speeds, as the stator reactance also increases with generator speed. The figure shows that a traditional low-inductance machine operates as a variable-magnitude voltage source, whilst a highinductance machine operates as a fixed-magnitude current source when its output voltage is much less than its open-circuit voltage. This concept, along with an explanation of the ideal machine model used to determine the loci, will be further discussed in Section 2.1.2.



Figure 1.8: Current vs. voltage locus of low and high-inductance PM generators, for generator speeds of 1 and 2pu.

## 1.4 Standalone Power Converters

The most common standalone power converters are discussed below, these are used for battery charging, turbine power maximisation, or to operate household appliances.

## 1.4.1 Common Power Converters

Power converters, such as inverters [36] and rectifiers have been used to obtain constant DC voltages from AC generators. Rectifiers have either been controlled [37,38], or uncontrolled and used as a switched-mode rectifier [39], or with a DC-DC converter. Common DC-DC converters include the *boost* [40–44], *buck* [45], and *buck-boost* converter [46], of which the boost converter is the most common.

Note that the use of rectifiers and power converters for small-scale wind generation, is similar to that for automotive power generation using an alternator. Hence, a summary of automotive and small-scale wind power generation research, including the types of power converters and generators investigated, is shown in Table 1.3, where IPM, ISA, LA and SG, are acronyms for interior PM, integrated starter-alternator, *Lundell* alternator and synchronous generator, respectively.

Table 1.3:	Summary	of the	research	and	${\rm the}$	type	of	generators	and	power	converters
used for sn	nall-scale w	ind and	d automo	tive	powe	er gen	nera	ation.			

Application	Wind	Automotive		
Inverter	IPM - Morimoto [36]	ISA - Lovelace [47]		
Controlled Bectifier	Jones [37]	LA - Liang [48]		
	PM - Raju <sup>*</sup> [38]	PM - Naidu [49]		
	PM - Venkataramanan [39]	LA - Hassan $[50]$ , Rivas $[51]$		
SMR		and Perreault [52]		
		IPM - Soong et al. $[53]$		
	PM - Chen <sup>*</sup> [40], Hao [41]	PM - Tolbert [54]		
Boost	SG - Prats <sup>*</sup> [42], Song <sup>*</sup> [44]			
	$Smith^*$ [43]			
Buck	PM - Chau <sup>*</sup> [45]	PM - Gutt [55]		
Buck-Boost	$SG^{*}$ [46]	PM - Tolbert [54]		

Note: \* indicates that the power converters are used as part of a grid-connected system.

#### **Converter Circuits**

The circuits of the controlled rectifier, inverter, buck, boost, buck-boost and switchedmode rectifier (SMR) are shown in Figure 1.9, whilst the key properties and differences are summarised in Table 1.4. Note that the figure does not include the uncontrolled rectifier. However, it is included in the table as it can be used for battery charging and other DC applications without an additional power (DC-DC) converter. In addition, the switches of each power converter, with the exception of the controlled rectifier, are controlled using pulse-width modulation (PWM) techniques.



Figure 1.9: Comparison of DC power converter equivalent circuits, showing the (a) controlled rectifier, (b) inverter, (c) buck, (d) boost and (e) buck-boost converters, and (f) the switched-mode rectifier. The converter DC output voltage,  $V_{out}$ , is shown for each converter, along with its DC or AC (line) input voltage, i.e.  $V_{in}$  or  $V_L$ , respectively.

Converter Topology	Switch	Control	Relative
	Count		Cost
Uncontrolled Rectifier	0	-	low
Controlled Rectifier	6	medium	medium
Inverter	6	complex	high
Buck	1	simple	medium
Boost	1	simple	medium
Buck-boost	1	simple	medium
SMR	1	simple	low-medium

Table 1.4: Comparison of small-scale wind turbine generator DC power converters

Although the buck, boost and buck-boost converters of Figure 1.9 are shown without an uncontrolled rectifier, this is required for use with a wind turbine as these converters require a DC input voltage. As such, the rectifier and boost converter combination appear similar in appearance to the SMR. However, the SMR avoids the rectifier output capacitor and inductor, making it cheaper and physically smaller than an equivalent boost converter. Note that the boost inductor, which is required to allow the voltage boosting action to occur, is conveniently provided by the machine inductance in the SMR. In addition, due to its widespread use, the boost converter is discussed further in this study together with the uncontrolled rectifier, inverter, controlled rectifier and SMR.

The high number of controllable switches accounts for the high and medium costs of the inverter and the controlled rectifier, respectively. In addition, both the controlled rectifier and the inverter require information about the generator rotor position, which complicates the control of each converter. In contrast, the SMR and boost converter do not require rotor position information and each have only a single controllable switch, which is controlled by a fixed duty-cycle PWM signal. These aspects reduce the control complexity and cost of each converter. Note that the boost converter with rectifier attracts a higher cost than the SMR, due to the required DC link capacitance and inductance.

Although inverters are conventionally used to convert DC voltage to AC voltage, they are also used to convert AC voltage to DC voltage. Despite its similar appearance to the controlled rectifier, the inverter uses forced-commutated switches, such as the MOSFET or IGBT, compared to thyristors that have controllable firing angles (if used as a rectifier). The inverter switches are more expensive than thyristors and are operated using a PWM switching scheme, which accounts for the high cost and complex control of inverter.

#### **Converter Output Power Comparison**

Although the SMR offers a low-medium cost and simple control, it is also important to consider the power characteristic of each converter. These are analysed and compared to that of a standard uncontrolled rectifier for reference. Note that the term "high" in high-inductance generator implies a rated current of 1pu.

### 1.4.2 Uncontrolled Rectifier Operation

The uncontrolled rectifier is the cheapest and the simplest power converter used in practical systems. It consists of 6 diodes and is used to convert AC voltage to DC, and as such can be used to charge a battery (see Figure 1.10(a)). Charging occurs providing the turbine speed is sufficiently high, i.e. the generator back-EMF voltage is greater than the battery voltage. Note that although the induced voltage is directly proportional to speed, the rectifier output (battery voltage) will remain fixed as the generator speed and hence back-EMF voltage increase.

The power characteristic of an uncontrolled rectifier can be determined by simplifying the generator, rectifier and battery models. The battery can be replaced by a variable resistive load. Since the rectifier forces the generator voltage and current to be in-phase [56], it can be modelled as a variable three-phase resistive load. The rectifier equivalent circuit and its simplified models, including the DC and three-phase resistive load models, are summarised in Figure 1.10. Note that the single-phase model also includes the generator induced voltage, E, generator phase reactance, X, equivalent load resistance, R, and resulting current, I, and voltage, V.



Figure 1.10: Uncontrolled rectifier modelling, showing (a) circuit diagram, (b) equivalent DC resistive loading, (c) simplified AC resistive loading, and (d) single-phase equivalent circuit. Note that the rectifier is represent by the dashed and solid boxes in (a) and (b), respectively.

#### Phasor Analysis

In this study, the machine equivalent output voltage, current and power is simulated for a DC link voltage of 1pu. It is assumed that a generator speed of 1pu yields a reactance and a generator back-EMF voltage of 1pu. The output voltage is the vector summation of the induced voltage and the voltage drop across the stator reactance, as shown in Equation (1.5). The machine power, P, is given by Equation (1.6), where  $\phi$  is the angle between the output voltage and current phasors (see Figure 1.11). The figure shows that  $\phi$  is equal to 0° as the rectifier forces the output current and voltage to be in-phase with each other. The phasor diagrams also show the angle between the induced voltage and current phasors,  $\theta$ ; the torque per ampere is maximised when this is equal to 0°.

$$V = E + j X I \tag{1.5}$$

$$P = V I \cos(\phi) \tag{1.6}$$



Figure 1.11: Rectifier phasor diagram showing induced voltage, E, output voltage, V, output current, I, and voltage drop across the inductor, jXI, for generator speeds of (a) 1, (b) 1.5 and (c) 2pu. Note that current does not flow for speeds  $\leq$  1pu, and that the induced and output voltage phasors are equal for case (a); they are shown to be separated by a small distance for simplicity.

As shown in the above phasor diagram, the machine current does not flow to the load for generator speeds less than 1pu. This occurs as the induced voltage, which is directly proportional to machine speed, must exceed that of the DC link voltage for current to flow. The current (and hence power) sharply increases with generator speed above speeds of 1pu, and levels off approaching its peak value of 1pu as the generator speed increases. This is known as *uncontrolled generation* (UCG) [56], as the power increases with speed and is not controllable [33]. This is illustrated in Figure 1.12.



Figure 1.12: Generator output (a) voltage, and (b) current and hence power, vs. generator speed characteristic.

Small-scale turbines are often located in areas that experience low average wind speeds. The resulting low turbine and hence generator speeds indicates that the use of an uncontrolled rectifier will prevent power flow at low wind speeds. In addition the power generated at high wind speeds is uncontrollable (see the above figure). The solution is to use a power converter that allows the generator current and hence torque to be controlled. This allows power to be extracted at low wind speeds, and controls the turbine speed such that the peak aerodynamic efficiency  $(c_p)$  is obtained for all wind speeds. Therefore, a power converter can effectively operate as a maximum power point tracker, and is hence often used as an intermediate stage with another power converter, e.g. a grid-connected inverter (as seen in Table 1.3, references [38, 40, 42–46]).

#### 1.4.3 Switched-Mode Rectifier

The switched-mode rectifier operates in a similar manner to that of the boost converter, i.e. it increases (boosts) its input voltage to a fixed output voltage. The boosting action is

used to effectively reduce the rectifier output voltage to some value less than the induced machine voltage. This is achieved by varying the converter duty-cycle, and allows power to be produced by the machine even at low generator speeds. As such, the SMR is modelled in the same manner as the uncontrolled rectifier, except the effective rectifier output power can be controlled by the duty-cycle.

#### Phasor Analysis

The rectifier phasor diagrams of a SMR, corresponding to maximum output power below generator speeds of 1.41pu, are shown in Figure 1.13. Those corresponding to speeds greater than 1.41 pu are identical to those of the uncontrolled rectifier (Figures 1.11(b) and 1.11(c)), and are hence not repeated here. The output current and voltage phasors, of the previous section, was shown to be in-phase due to the resistive load model. In addition, the output current is equal to 0.71pu over the speed range, which corresponds to maximum power that was obtained when the phase resistance and reactance of Figure 1.10(d) are equal in magnitude. This is known as *load-matching*. Note that the phasor diagrams of Figure 1.13 are identical in shape, but scale in magnitude with generator speed.



Figure 1.13: SMR phasor diagrams showing induced voltage, E, output voltage, V, output current, I, and voltage drop across the inductor, jXI, for generator speeds of (a) 0.71, (b) 1 and (c) 1.41pu.

The generator current and power (corresponding to the peak power obtainable using the SMR) is shown in Figure 1.14 as a function of generator speed. The peak generator (and SMR) output power increases linearly with generator speed until a speed of 1.41pu, after which, the peak power increases in an identical manner as the uncontrolled rectifier. As such, the SMR is said to operate as boost rectifier at low generator speeds, whilst for speeds above 1.41pu, it acts as an uncontrolled rectifier under maximum output power operation. Note that any power below the solid line in Figure 1.14(b) can be obtained using the SMR. This is further discussed in Chapter 3, along with boost rectification.



Figure 1.14: Generator output (a) current, and (b) power vs. speed, corresponding to maximum power operation, using the switched-mode rectifier (SMR). The dashed line represents the current and power obtained using an uncontrolled rectifier.

## 1.4.4 Inverter Operation

The inverter is also capable of extracting power from the generator at low speeds, however, it requires an accurate rotor position sensor, which attracts a higher cost and complicates the control. The inverter is able to obtain rated current for all generator speeds and maintain it in-phase with the induced voltage at low generator speeds to maximise the output power. The inverter is hence able to extract more power than that obtainable using the SMR at low generator speeds. The corresponding machine model, under inverter operation, is shown in 1.15.

## Phasor Analysis

The machine phasor diagrams corresponding to inverter operation are shown in Figure 1.16, for various speeds. Consider Figure 1.16(a), which corresponds to the phasor



Figure 1.15: Simplified machine phase model operating with an inverter, showing the induced voltage, E, reactance, X, and output voltage, V, and current, I.

diagram at a generator speed of 0.71pu. Under this condition, the torque and hence machine power are maximised as  $\theta$  is equal to 0°, i.e. the current and induced voltage phasors are in-phase with each other. The output voltage is now twice that obtained using the SMR (1 vs. 0.5pu), whilst the current is now also 41% larger (1 vs. 0.71pu). In addition, the current and output voltage phasors now have a 45° phase shift, i.e. power factor of 0.71 lead. Despite the phase shift, the increased current and voltage allows the inverter to extract twice the output power of the SMR (see Equation (1.6)).



Figure 1.16: Inverter phasor diagram showing induced voltage, E, output voltage, V, output current, I, and voltage drop across the inductor, jXI, for generator speeds of (a) 0.71, (b) 1 and (c) 1.41pu. Note that this figure shows  $\phi$ , rather than  $\theta$  as shown in the previous phasor diagrams.

The above phasor diagram shows that the voltage and current phasors remain fixed at 1pu for all speeds greater than 0.71pu. The phase angle of the current relative to the induced voltage,  $\theta$ , is shown to increase with speed. This reduces the power factor angle,  $\phi$ , which allows the inverter to approach unity power factor and hence a peak power of 1pu, as the generator speed increases. The machine current and hence power vs. speed characteristic, using an inverter, is summarised in Figure 1.17.



Figure 1.17: Generator output (a) current, and (b) power vs. speed, corresponding to maximum power operation, using the inverter (INV). The dashed line represents the current and power obtained using an uncontrolled rectifier.

#### 1.4.5 Power Comparison of Standalone Converters

Figures 1.14(b) and 1.17(b) clearly show that both the SMR and inverter obtain power from the generator at low speeds, unlike the uncontrolled rectifier. Further comparison shows that the inverter is able to extract twice the peak power compared to the SMR at low generator speeds, whilst at high speeds, they both asymptote towards the same power, i.e. 1pu. Recall that both the SMR and inverter are able to reduce the power to any level below their maximum value. This is summarised in Figure 1.18, which also compares the power range of the uncontrolled and controlled rectifier.

The figure clearly shows the benefit of a controlled power converter, i.e. power can reduced to any desired value, unlike the rectifier whose output power is governed by the generator speed. The controlled rectifier is unable to extract power from the generator at low speeds. The SMR can extract power at low generator speeds and offers simple control. This is achieved by effectively load-matching the generator to the voltage source

#### CHAPTER 1. INTRODUCTION

load by choosing an appropriate duty-cycle. The inverter is able to offer twice as much power at low speeds than the SMR. This is achieved as the inverter offers greater control of the generator output current. Note that there is little difference between the high-speed output power of the inverter and SMR.



Figure 1.18: Comparison of power converter output powers, showing that obtainable using an (a) uncontrolled rectifier, (b) controlled rectifier, (c) switched-mode rectifier (SMR), and (d) inverter. The dashed lines of (b)-(d) represent the power obtained by the (uncontrolled) rectifier, and are shown for reference. Subfigure (d) also compares the power obtained by the SMR, for reference.

Hence, the choice of power converter is selected based on a low-speed output power vs. cost trade-off. Despite producing only half of the inverter's low speed output power, the SMR is selected due to the absence of sensors, which reduce the total cost, and its simple control.

## 1.5 Grid-Connected Inverters

Grid-connected (GC) inverters used for small-scale wind turbines use a rectifier to obtain a DC voltage from the generator. The majority of the inverter is hence similar to those used for photovoltaic (PV) installations, and therefore the literature about PV inverter topologies is also reviewed and discussed.

## 1.5.1 Introduction

Small-scale wind turbines feed power to the utility grid using a grid-connected inverter (GCI). Unlike standalone inverters that control the output voltage, GCIs control the output current [57], as they are effectively connected to an infinite bus (fixed voltage source). However, these inverters must meet strict grid requirements, such as total harmonic distortion, power factor and DC current injection specifications, which will be discussed later in Section 4.1.1. Note that these requirements complicate the inverter design and control, which increase the total cost. This is illustrated in Figure 1.19 which compares the cost of commercially available standalone and GC inverters. The costs given in the figure are obtained from a catalogue [25], and are shown as a function of inverter power rating.



Figure 1.19: Cost comparison of standalone (ST) and grid-connected (GC) inverters. The solid and dashed lines represent cost trendlines for ST and GC inverters, respectively.

The above figure clearly shows the increased cost associated with a GCI, i.e. about AUS\$1000 more than a similar-sized standalone inverter. The survey also shows that the input voltage increases with power, for standalone inverters. This is required to prevent high input currents (and hence losses), and to improve the overall efficiency. The figure also shows that the inverter cost can be optimised by using a 12V inverter for power levels below 650W, whilst a 24V inverter should be used below 2.4kW.

Inverters can be classified by many attributes, including input type, the switch / commutation type, controllable output variable, and transformer type, which are discussed in the following section.

## 1.5.2 Line-Commutated Inverters

Early GCIs used for PV applications in the late 1980's were line-commutated currentsource inverters and in the order of several kilowatts [58]. This type of inverter was also used with wind turbines as the technology had been used for high-voltage DC transmission since the 1970's [59] and it was a mature, efficient, robust, low-cost and reliable technology [33, 58, 60]. However, the main drawbacks of this inverter are the reactive power drawn from the grid, which results in a power factor of between 0.6 and 0.7 [58] lag, and the high harmonic content of the output current. The output current of a line-commutated inverter is typically a square-wave of 120° conduction angle, which contains large loworder harmonic components in the range of 250–350Hz [61]. An example of the output current and the line-commutated current-source topology is shown in Figure 1.20. Note that the inverter shown in the figure corresponds to a three-phase CSI, and that neither the rectifier or the generator are included.



Figure 1.20: Circuit diagram and square-wave output current of the line-commutated current-source grid-connected inverter topology.

#### Harmonic Reduction

Compensators (based on auxiliary PWM VSI with line-current feedback [33]) and passive tuned filters have been examined in [60] to reduce the output current distortion and improve the power factor. The use of *Notch* filters will be examined in detail in Section 5.2.6, despite being large and costly for low-order harmonic frequencies [61]. Notch filters, tuned to the  $5^{th}$  and  $7^{th}$  order harmonics, together with a second-order damped filter, tuned to the  $11^{th}$  order harmonic, were shown to significantly improve the power factor. However, they were ineffective at removing the low-frequency harmonics.

In contrast, the use of active power compensators (active parallel filters) was shown to significantly improve the output current distortion. This is achieved by controlling the compensators to generate AC currents which form an ideal output waveform, when combined with the distorted current [60]. Despite their success, the addition of active compensators increases system complexity and cost. An outline of a line-commutated inverter with passive filters and active compensation is shown in Figure 1.21.



Figure 1.21: Current-source inverter with active compensation and passive filters, examined by Chen [60]. Note that the rectifier and PM generator are also shown, along with a three-phase passive filter<sup>\*</sup>.

Note that active compensators are modern power electronic devices [60], which use self-commutating semiconductor switches, such as the IGBT or MOSFET. As it is well known, these switches have replaced thyristors in many power electronic converters, due to the recent rapid development of the semiconductor device industry [58,62]. As such, the majority of commercially available GCIs now use self-commutating switches, and highfrequency (above 3 kHz [63]) pulse-width modulation (PWM) control strategies. The resulting high-frequency switching harmonics are attenuated using a smaller and lower loss filter than that required for a line-commutated inverter [61], and allow much greater control of the power factor. A survey of in [62] shows that modern inverters obtain power factors of 0.9 or greater, over a wide range of operating conditions.

## 1.5.3 Self-Commutated Inverters

Two types of self-commutated inverter exist, these are the voltage-source inverter (VSI) and the current-source inverter (CSI). The VSI requires constant DC input voltage, which is provided by a DC link capacitor, whilst a CSI requires a constant DC current, which is provided by a DC link inductor. These passive elements decouple the grid from the input power source and provide energy storage. This is required as it allows the inverter input and output powers to differ and prevents input current or voltage ripple, which affects output current distortion; this is further examined in Section 5.1.1. However, a trade-off exists as increased storage capacity increases the overall inverter cost and power losses.

In addition, the arrangement of semiconductor switches, feedback controllers and lowpass filter arrangements of the VSI and CSI topologies differ; low-pass filters are discussed later in Section 5.3. The CSI requires a series-connected diode with each IGBT or MOS-FET to provide unidirectional current flow and bidirectional voltage blocking [64, 65]. These diodes, however, increase the overall cost and conduction (power) losses [65]. Similarly, the switches of the VSI require anti-parallel (freewheeling) diodes to provide bidirectional current flow and unidirectional voltage blocking capability [64]. The key differences between the self-commutating VSI and CSI topologies are highlighted in Figure 1.22.



Figure 1.22: Comparison of self-commutating (a) voltage-source inverter (VSI), and (b) current-source inverter (CSI) topologies. The key difference are the energy storage element and switch / diode configuration. Note that the switches are either MOSFETs or IGBTs.
Open-loop PWM switching schemes, such as the common bipolar and unipolar sinusoidal schemes, could yield an alternating current from an ideal current source, as shown in Figure 1.23. In practise, however, feedback is required as input currents / voltages vary. Both types of inverters use feedback control systems, that sample and adjust control signals, to deliver a sinusoidal current to the grid. This differs from standalone inverters that traditionally sample and regulate the output voltage [57]. The fixed grid voltage regulates the inverter output, allowing the output current of a CSI to be controlled in proportion to the modulation factor, which makes the CSI appropriate for a power conversion system [66].



Figure 1.23: Ideal (a) input, and (b) bipolar PWM and (c) unipolar PWM output currents.

Note that the bipolar output current is obtained by switching the state of both phase legs, whilst the unipolar scheme requires only one phase leg state to change [67]. The unipolar sinusoidal PWM switching scheme, including a harmonic analysis, is further discussed in Section 5.2.4.

### **PWM Control Strategies**

Grid-connected inverters use a variety of feedback based PWM control strategies to control the output current, which include current hysteresis and space-vector modulation schemes [68]. A comparison of the operation of a 100kVA IGBT VSI using each control strategy, is reported in [68]. The space-vector control scheme results in the least current THD, i.e. 5%, however, it is the most complex scheme, has a slow current response, and attracts the highest cost. The current hysteresis scheme yields a current with 7.7% THD, and is well suited to fast-varying input powers, as it provides over-current protection and offers high power factor.

More information regarding both the open-loop and feedback based PWM control signals, including the derivation of control signals, is given in appendix A.1.

#### Voltage vs. Current-Source Inverters

Voltage-source inverters are generally regarded to be more efficient than an equivalent CSI, as the DC link capacitor has lower power losses compared to the resistive and core losses of a DC link inductor [69]. Despite this, the output voltage of a VSI must exceed that of the peak grid voltage, implying that either a high-voltage input source or the use of a step-up transformer or boost converter is required. The use of either increases the total cost and reduces the overall efficiency [64].

In contrast, the average output voltage of a CSI is equal to that of the grid. The CSI is also advantageous over the VSI as its output fault current is limited. However, care must be taken using the CSI to prevent open circuiting the DC link inductor, which may destroy inverter components. Similarly, VSIs must avoid simultaneous gating of two switches in the same leg to prevent high currents destroying components. Hence *dead time* must be introduced to the VSI controller. The drawback, however, is that dead time causes output current distortion [64]. The amount of distortion is proportional to the length of the dead time [70].

#### Switching Loss

The main drawback of self-commutating inverters is the higher switching power loss. This is caused by commutating the switches whilst the current flowing through (or the voltage drop across) the device is non zero, and is also referred to as *hard-switching*. The switching loss,  $P_{sw}$ , is proportional to the device turn-on and turn-off times,  $t_{on}$  and  $t_{off}$ , the output current,  $I_{out}$ , DC link voltage,  $V_{DC}$ , and the switching frequency,  $f_{sw}$ , as summarised in Equation (1.7) [67], assuming an inductive load.

$$P_{sw} = \frac{1}{2} (t_{on} + t_{off}) V_{DC} I_o f_{sw}$$
(1.7)

Some inverter topologies use zero-voltage or zero-current switching techniques to reduce switching losses. These techniques involve commutating a switch when the current through or the voltage across the device is zero. These are known as *soft-switching* techniques. Examples of GCI topologies that employ soft-switching techniques are discussed in Section 1.5.6.

# 1.5.4 Transformer Type

Three transformer topologies exist for grid-connected inverters, these include the *line-frequency* (LF), *high-frequency* (HF) and *transformerless* (TL) topologies. Each topology is discussed below.

### Line-Frequency Transformer

Line-frequency transformers are designed to operate at the grid frequency (50 or 60Hz) and are located between the grid and inverter output. The role of the transformer is to decouple (isolate) the inverter from the grid, provide voltage boosting (step-up transformer) and eliminate DC currents. The transformer also provides line reactance and a small amount of resistance that may assist with filtering or filter damping.

The majority of PV grid-connected inverters used in Korea employ LF transformers [71]. However, despite the above mentioned benefits, there is a trend to move away from this topology as these transformers are heavy, bulky and expensive [71].

### **High-Frequency Transformers**

The physical size and cost of a transformer can be reduced by increasing its operating frequency [71]. Hence, transformers of frequencies 400Hz and above are used in some grid-connected inverters. Despite the size and cost benefits of a HF transformer, the GCI requires two inverter stages and additional components, which reduce the overall inverter efficiency and reliability [72], and increases control complexity [58]. In addition, DC currents may be injected in to the grid, which cause saturation and overheating of utility distribution transformers.

High-frequency transformers are preferred and used in inverters that have low rated powers, whilst LF transformers are generally used in high-power inverters [73, 74].

### Transformerless Topologies

The transformerless inverter topology has been examined in [75], as it is physically smaller, lighter, cheaper and more efficient than topologies that contain transformers. This is verified in Table 1.5, which compares the peak efficiency, weight and cost of the transformer and TL GCI topologies [76].

Despite the benefits of the TL topology, it is unable to isolate the energy source from the grid, and can not remove DC components at the output. In addition, low voltage Table 1.5: Comparison of single-phase grid-connected inverter properties [76].

NOTE: This table is included on page 36 of the print copy of the thesis held in the University of Adelaide Library.

sources such as parallel-connected PV cells require voltage boosting to transmit power to the grid. This is achieved using an additional DC-DC converter, which increases cost and reduces efficiency. The common-mode voltage and lack of isolation between the grid, PV cells and associated wiring are probable reasons that the TL topology is prohibited in Italy and the United Kingdom [58].

# Cost / Performance Trade-Off

As stated above, each transformer topology has certain advantages and disadvantages associated with it. A summary of key transformer and transformerless GCI topology properties are given in Table 1.6.

Table	1.6:	Comparison	of	line-frequency,	high-frequency	and	${\it transformerless}$	grid-
connec	ted in	verter topolog	gy I	properties.				

Topology           Property	Line-Frequency	High-Frequency	Transformerless
Operating Frequency	50  or  60 Hz	>400Hz	-
Size (relative)	Large	Medium	Small
Weight (relative)	$\operatorname{High}$	Medium	Low
Cost (relative)	High	Medium	Low
Provide Isolation ?	Yes	Yes	No
Remove DC Components ?	Yes	No	No
Provide Line Reactance ?	Yes	No	No
Boost Voltage ?	Yes	Yes	No

It should be noted that voltage boosting and isolation are important issues for PV applications, due to their low voltage and roof-top installations, respectively. These issues, however, are less likely to be of concern for a grid-connected wind turbine as an oversized generator can be used to produce higher voltages, and turbines are often located upon high towers away from inadvertent human interaction.

# 1.5.5 Voltage Source Topologies

### Uncontrolled Rectifier

The use of an uncontrolled rectifier together with a current-controlled voltage-source inverter is the simplest type of GCI. This topology is reported in [68] and is shown in Figure 1.24. The inverter uses a fast analogue-to-digital converter (ADC) and microcontroller to adjust the inverter control signals based on the sampled output current, to produce the desired sinusoidal output currents.



Figure 1.24: Common 3-phase voltage-source grid-connected inverter.

The main disadvantages of the above inverter topology are the limited life span, due to the DC link (electrolytic) capacitor, and the poor performance at low wind speeds. Low wind speeds cause the generator to rotate at low speeds, where the resulting induced voltage is less than the required DC link voltage. Under this condition the DC link capacitor is being charged through the inverter anti-parallel diodes, which causes excessive reactive power flow [77]. In addition, the controller loses its ability to control the output current and hence injects a highly distorted current, that may contain a significant DC component, to the grid.

To prevent high output current distortion and/or DC current injection, a normal IGBT inverter should disconnect itself from the grid at low wind speeds (3–5m/s [77]). This implies that the wind turbine will fail to extract energy at these low wind speeds, and may account for as much as 20% of the turbine's annual energy yield. A solution investigated in [77], was the inclusion of a reverse-blocking diode between the DC link capacitor and the inverter output stage, as shown in Figure 1.25. The diode is introduced to ensure unidirectional power flow at low wind speeds, i.e. prevent the grid charging the DC link capacitor. This solution has only limited success as the inverter is still unable to inject a high quality sinusoidal current to the grid at low wind speeds, due to the presence of low-order harmonics. This causes the overall power factor to be lagging and increases generator losses [77].

NOTE: This figure is included on page 38 of the print copy of the thesis held in the University of Adelaide Library.

Figure 1.25: Single-phase voltage-source inverter with reverse-blocking diode, as investigated in [77], to prevent reactive power flow and high current distortion at low wind speeds.

### Low Wind Speed Power Improvement

DC-DC power converters, such as the boost [40, 42, 44], buck [45], and buck-boost [46] converters have been used as an intermediate stage of a grid-connected inverter (see Table 1.3). These converters are used to maximise the turbine power by controlling the turbine speed such that the peak  $c_p$  is obtained. As such, these converters operate as maximum power point trackers (MPPT); a term which is more commonly used for PV GCI topologies.

The use of a boost converter is most common [78], and has the advantage of effectively adjusting the rectifier output and hence generator output voltage. This allows powel at low generator (wind) speeds to be extracted, as previously seen in Section 1.4.3. The equivalent circuit of the combined boost converter and VSI is shown in Figure 1.26. The main drawback of this arrangement is the use of two electrolytic capacitors, which reduce the overall reliability, and the necessity of a boost inductor.

NOTE: This figure is included on page 38 of the print copy of the thesis held in the University of Adelaide Library.

Figure 1.26: Voltage-source grid-connected inverter with boost converter.

The use of back-to-back PWM inverters was investigated in [79] to optimally control a PM generator in a wind turbine, and is now widely used [9]. The circuit is shown in Figure 1.27, which has a high switch count. This, together with the additional requirement of a rotor position sensor and a complex controller, significantly increases the overall cost. Despite this drawback, the use of the inverter input stage allows better utilisation of the generator, compared to that of a boost converter and will allow the use of a smaller generator (see Figure 1.18).

NOTE: This figure is included on page 39 of the print copy of the thesis held in the University of Adelaide Library.

Figure 1.27: Back-to-back PWM voltage-source grid-connected inverter configuration investigated in [79].

# **High-Frequency Transformer**

A high-frequency transformer PV GCI was constructed and tested in [71]. The input stage consists of a DC-DC converter containing a transformer operating at 20kHz and is used to boost the PV array output voltage and provide an output current which resembles a full-wave rectified sine-wave, as shown in Figure 1.28. The transformer output current is filtered by a small inductance before being fed to an *unfolding* circuit to produce the desired AC output current.

NOTE: This figure is included on page 39 of the print copy of the thesis held in the University of Adelaide Library.

Figure 1.28: High-frequency (HF) transformer inverter topology presented in [71]. The inverter consists of HF PWM DC-DC converter and a line-frequency H-bridge unfolding circuit. Note that various current and voltage waveforms illustrate the inverter operation.

Despite the advantage of the reduced size and cost of the transformer, the HF transformer inverter requires several additional components, which increase its complexity and can reduce its reliability and efficiency [72], compared to that of a LF transformer inverter.

# 1.5.6 Current Source Topologies

### **PWM** Controlled Inverter

Various control strategies were examined with a sensorless self-commutated PWM currentcontrolled GCI in [21]. The wind energy system, which is shown in Figure 1.29, consists of an uncontrolled rectifier, DC link inductor and PWM current-source inverter. The turbine output power is maximised by adjusting the DC link voltage according to wind speed. This is achieved by varying the inverter modulation index, which ultimately adjusts the pule width (duty-cycle) of the PWM control signal. Five alternative power maximisation control strategies, including one based on an open-loop controller, were shown to improve the power maximisation capability of the turbine without significantly increasing the complexity of the control.



Figure 1.29: Three-phase current-source PWM grid-connected inverter circuit.

The main drawback of the above inverter is the high switching losses, due to the high-frequency PWM control signals, and the resistive and core losses associated with the DC link inductor.

# **H-Bridge Inverter**

The line-frequency commutated H-bridge inverter is used as part of a PV GCI, as shown in Figure 1.30. The inverter input current is modulated by another converter stage and resembles that of a rectified sinusoid. The H-bridge inverter hence acts as an *unfolding circuit*, and operates at twice the grid frequency producing a sinusoidal output current.

The advantage of this circuit is the zero-current and zero-voltage switching scheme that effectively eliminates switching losses, i.e. only semiconductor conduction losses remain



Figure 1.30: Line-commutated H-bridge inverter used as part of a photovoltaic gridconnected inverter, reported in [80]. The inverter input current, which resembles a rectified sinusoid, and the resulting sinusoidal output current is also shown.

[80]. The circuit cost could be reduced by using thyristor switches, which would naturally commutate based on the input current zero-crossings.

### Alternative Soft-Switched Inverters

Two soft-switching current-source inverters have been investigated in the literature. The first study, investigated in [81] is shown in Figure 1.31, where a DC voltage source and DC link inductor provide the current-source input current. The inverter itself is based on a line-commutated topology, i.e. it uses thyristors. The key difference is that an additional parallel-connected IGBT is used to short the current-source output to ground using a high-frequency PWM control scheme. This allows the thyristors to commutate naturally, which is referred to as *active commutation*. However, the IGBT in this topology is hard-switched, which will cause significant switching loss.



Figure 1.31: Actively-commutated thyristor based current-source inverter investigated in [81], showing the circuit and ideal input and output current waveforms.

The advantage of the above circuit is that rugged, high-power, low-cost thyristors are used for the H-bridge components [81]. These commutate naturally as a result of the high-frequency IGBT switching. The high-frequency PWM switching implies the size of the DC link inductor and filter capacitors can be reduced in size, and that the output current harmonic contents can be well controlled [81]. In addition, the use of thyristors, which have a blocking capability, avoids the need for series-connected reverse-blocking diodes, unlike a self-commutating CSI.

An alternative soft-switching CSI topology was also investigated for a PV GCI in [69]. The inverter topology is summarised in Figure 1.32, which consists of a DC link commutation circuit, and an IGBT H-bridge inverter with series-connected diodes. The commutation circuit is comprised of two semiconductor switches, two diodes and an LC resonant circuit, which provides a current path such that zero-current switching of the H-bridge can be achieved. The drawback, however, is the added cost and the increased control complexity of the LC resonant circuit, compared to that of a single IGBT used in [81].



Figure 1.32: Soft-switching single-phase current-source grid-connected inverter topology examined by Han et al. [69], showing the current source, commutation circuit, current-source H-bridge inverter, and line output filter.

Note that the PV cell is treated as the current source, as PV cells have characteristics of both voltage and current sources [69].

### 1.5.7 Current Trends

A survey of PV GCIs [62], revealed that all modern inverters are of the voltage-source type. The majority (81%) of these control the inverter output current, whilst the remaining 19% control the inverter output voltage. A second survey done in [58], indicates that 55% of GCI incorporate a line-frequency transformer. High-frequency transformer topologies make up 0.3% of the market share, despite the size and cost benefits of such a transformer.

About 45% of the market share is made up of transformerless topologies, with about 96% of these employing a boost converter, whilst 4% incorporate a buck converter.

# 1.5.8 Proposed Grid-Connected Inverter

Despite the current trend of using voltage-source inverters, the proposed GCI is based on a CSI topology. This is shown in Figure 1.33, and will be further discussed in Chapter 4. The topology is similar to that investigated in [81], however, the proposed topology uses a high-inductance PM generator and uncontrolled rectifier as the inverter current source, rather than a DC voltage source and series-connected DC link inductor. A MOSFET is used to allow zero-current switching of the thyristor H-bridge inverter, which actively commutates the thyristors. Note that this will be referred to as the *current wave-shaper* in Part II (Chapters 4–6) of this study. A capacitive-inductive, CL, low-pass filter is used in the output of the inverter to remove high-frequency PWM harmonic components. In addition, the proposed inverter concept will be simulated and experimentally verified with operation into a voltage source (rather than a three-phase resistive load as tested in [81]).



Figure 1.33: Proposed grid-connected inverter topology, including the PM generator current source, active commutator, thyristor H-bridge and low-pass filter.

It should be noted that the above topology has also been investigated for PV-based GCIs, as part of another PhD research project. Some of the publications were listed on page xvii of this thesis.

# 1.6 Thesis Overview

## 1.6.1 Aim of Research

The aim of this research is to design, test and validate a novel low-cost power converter topology to control a commercially-available small-scale wind turbine. The DC-DC and proposed grid-connected inverter, both of which are based on the switched-mode rectifier topology, must be able to i) maximise turbine power below rated wind speed, ii) maintain rated power above rated wind speed, and iii) reduce the turbine speed at high wind speeds, to prevent turbine damage.

The operation of each power converter is determined from open-loop dynamometer testing. In addition, the turbine and DC-DC power converter is tested using a small wind tunnel, whilst a high-powered (1kW) grid-connected inverter is simulated over wide range of wind speeds.

# 1.6.2 Justification for Research

There is growing interest in renewable energy sources due to sustainability and fossil fuels concerns. It has been estimated that the small-scale wind turbine market is growing at roughly 40% per year [18], and that the US alone will sell US a total of \$25M-\$55M by 2010 [19], depending on the market and other factors such as government rebates, tax incentives etc. Within this market there is an increasing demand for low-cost grid-connected wind turbines, and it is expected that by 2020 small-scale turbines will provide 3% of the US electrical energy consumption [18].

Reducing the cost of the power electronic converter, which make up a significant portion of the cost of a renewable energy conversion system, including solar [82] and wind, will encourage more people to utilise renewable sources of energy. Increased wind or solar power utilisation will reduce the demand for conventionally derived electricity, whilst meeting the increased energy demand. In addition to the environmental benefits, e.g. reduced greenhouse gas emissions, customers who utilise grid-connected inverters are able to reduce their electricity bills, as grid operators purchase their customer's excess electricity.

# 1.6.3 Original Contributions

The first original aspect of this research is to investigate the suitability of a SMR to control a high-inductance PM generator as part of small-scale wind turbine. This is different to the work presented in [39] where a conventional low-inductance generator is studied. The turbine was simulated and shown to operate at the peak  $c_p$  over a range of wind speeds, using a sliding mode current control strategy that compared the switch current to a value corresponding to peak power from a wind-speed look-up table. The work presented in this thesis, however, shows that the SMR can maximise and also limit turbine power, and limit turbine speed. This is simulated and experimentally verified using both a dynamometer and small wind tunnel.

The second original contribution is the investigation of a novel grid-connected inverter topology. The novel current-source inverter is based on the SMR topology and uses a highinductance PM generator and rectifier as a constant-current source. In this research, the inverter operation is simulated and experimentally verified using a dynamometer test rig. The inverter topology, including a detailed generator model, low-pass output filter, and feed-forward controller, is analysed. An optimised high-powered (1kW) grid-connected inverter is designed and simulated, and shown to effectively control a wind turbine over a wide range of wind speeds.

# 1.6.4 Thesis Structure

This thesis contains five main chapters that are separated into two parts, as shown in Figure 1.34. Part I (Chapters 2 and 3), investigates the use of the SMR as a standalone (DC-DC) power converter. The selected high-inductance PM generator is described, modelled, characterised, simulated and experimentally tested in Chapter 2. The analytical and computer-based (PSIM<sup>®</sup>) models are compared with experimentally measured results

Chapter 3 discusses the equivalent circuit model and operating regions of the SMR. The SMR is simulated using PSIM<sup>®</sup> and is shown to match the experimentally measured data. The SMR is tested using both a dynamometer and wind tunnel test facility, as these arrangements allow the generator and wind speeds to be controlled, respectively. The SMR duty-cycle is varied in an open-loop manner, and its role as a power maximiser, and power and turbine speed limiter are successfully demonstrated.

Part II (Chapters 4, 5 and 6), investigates a novel grid-connected inverter topology based on the high-inductance PM generator and SMR topology. Chapter 4 discusses the



Figure 1.34: Thesis structure, showing part and chapter breakdown.

grid requirements, and the preliminary design, implementation and experimental testing of the proposed grid-connected inverter. The inverter concept is proved for a 150W case, by PSIM<sup>®</sup> simulations and by open-loop experimental testing.

Chapter 5 analyses in detail the proposed grid-connected inverter, low-pass filter design and a feed-forward based controller. The chapter begins by identifying the PM generator and uncontrolled rectifier (current source), and the PWM-switching scheme as harmonic sources. The harmonics caused by the unipolar PWM switching are also examined, and techniques to reduce all sources of harmonic distortion are discussed.

The low-pass filter and open-loop control algorithm are identified as the main cause of output current harmonic distortion. As such, the design of the low-pass filter for this type of current-source inverter, is thoroughly analysed. Several damped filter configurations are compared, and the parallel damped inductor is identified as the optimal configuration, offering the best power loss vs. harmonic attenuation trade-off. In addition, Chapter 5 analyses the power factor, harmonic attenuation and damping resistance power loss, and summarises a normalised low-pass filter design approach. The chapter also analyses two feed-forward control algorithms that adjust the inverter modulation index and hence dutycycle in real-time, to eliminate the input current harmonic components.

Chapter 6 combines the principles of wind energy conversion, described in Section 1.2, with the inverter analysis to design a higher-power (1kW) grid-connected inverter. The inverter is simulated using PSIM<sup>®</sup>, to investigate the control of the turbine power, torque and speed over a wide range of wind speeds, whilst delivering a high quality current waveform and hence power to the grid.

Finally, Chapter 7 summarises the aims and original contributions of the thesis. The chapter also identifies areas of work that are opportunities for further investigation.

# Part I

# Investigation of Switched-Mode Rectifier for Standalone Power Converter

# Chapter 2

# High Inductance PM Generator Characteristics

This chapter examines the machine characteristics of the selected permanent magnet generator. The machine construction and its high-inductance property are discussed. Experimental tests are performed to determine its parameters , which are then used to model the machine. The machine is simulated using analytical models and simulation packages, and experimentally tested over a wide range of operating speeds and loads to verify the models. Particular attention is paid to the DC load modelling and testing, as the PM generator will be used with a three-phase bridge rectifier for the majority of this research project.

# 2.1 Introduction

The generator selected for this project is a three-phase synchronous surface permanent magnet (PM) machine, that was originally designed and manufactured by Fisher & Paykel<sup>®</sup> (F&P) for motor applications. The machine was first reported in 1992 [83], and is directly coupled to the agitator and spin-bowl of F&P's range of Smart Drive<sup>TM</sup> washing machines. The direct-drive system operates effectively as the motor is able to provide high torque at low speeds while still capable of operating at high speeds; these features are required for the washing (agitation) and spin-drying stages, respectively.

Direct-drive systems replace the more conventional belt driven mechanical agitation and braking systems, which significantly improves the efficiency of the appliance [83]. Despite this, the motor requires additional components, such as a variable speed controller and Hall-effect sensors, to be operated effectively. These components, however, are not required for generator applications, and together with the machine's ready availability, make the machine a convenient low-cost generator. The high torque and wide operating speed characteristic of the F&P surface PM machine make it ideal for a low-cost small-scale wind generator.

An example of an F&P PM machine is shown in Figure 2.1, which shows a 42 tooth concentrated-winding star-connected stator arrangement. These windings are non-overlapped coils (coils wound around a single tooth), which produce very short end-winding length and hence reduce the copper weight and copper loss of the machine [35,84]. In addition, concentrated windings reduce the machine manufacturing cost, due to the winding simplicity, and increase the inductance compared to conventional distributed windings, for the same magnet flux linkage [35]. This high-inductance property is useful when designing power converters, and is further discussed in Section 2.1.2.



Figure 2.1: Fisher & Paykel<sup>®</sup> surface PM machine, showing the three-phase terminals, star point, concentrated stator windings, and permanent magnets.

The above surface PM machine uses an outer rotor configuration, as shown in Figure 2.2. The outer rotor configuration has several advantages over the more conventional interior rotor configuration, these include [85]:

- higher output torque due to larger rotor volume,
- magnets are more easily retained against centrifugal forces [86],



Figure 2.2: Cross-sectional views of (a) inner, and (b) outer rotor PM machines. Note that the stator windings are not included, for simplicity.

Commercially, the F&P surface PM machines are available in two varieties, these are the 56 magnetic pole / 42 stator teeth, and the 48 pole / 36 teeth configurations. The latter was introduced to reduce cogging torque [83]. These machines are classified as fractional slot winding types, as they have less than 1 slot per pole per phase (SPP). The machine SPP value is calculated according to Equation (2.1) [87], where S is the number of slots, p represents the number of magnetic poles pairs, and m is the number of phases; hence both F&P machines have a SPP value of 0.25. Fractional-slot PM machines with a SPP value of 0.25 or less allow the machine structure to have a lower output torque ripple, though the average output torque may be lower compared with PM machines with higher values of SPP [87].

$$SPP = \frac{S}{2p \cdot m} \tag{2.1}$$

Fractional-slot concentrated winding PM machines with a wide range of SPP values exist [87]. Each SPP value offers different machine characteristics [35] and a SPP design procedure, based on specific criteria, is summarised in [35, 87]. PM machines with a SPP value of 2/5 or 2/7 are identified as the best candidates. Despite this, the F&P surface PM machine is used for the small-scale test wind turbine in this study, as these are readily available. In addition, since this research project also investigates power electronic converter design, implementation, analysis, simulation and testing, an optimised machine design is not considered here.

# 2.1.1 Ideal Machine Model and Current vs. Voltage Locus

A simplified model of a surface PM machine with a three-phase resistive load, consists of a back-EMF voltage source, E, series stator reactance, X, and load resistance, R, as shown in Figure 2.3. The back-EMF voltage source has a voltage and frequency which are both proportional to the machine speed. Note that both the generator open-circuit voltage and stator reactance increase linearly with generator speed. Therefore, the subsequent short-circuit current remains fixed. The resulting normalised current vs. voltage (I-V) locus is shown in Figure 2.4(a). The locus is shown for generator speeds,  $\omega$ , equal to 1 and 2pu, and load resistances of 0.3 and 10 pu.

The surface PM machine I-V locus is compared to those of the DC and interior PM (IPM) machines, in Figure 2.4(b). The loci comparison is obtained from [88] and is shown for reference only. Note that the interior PM machine shows a voltage over-shoot, i.e. its peak voltage under load is larger than its open-circuit voltage. This phenomenon occurs as the result of its saliency. The IPM locus shown in the figure represents a saliency ratio,  $\xi$ , of 4, which corresponds to a voltage over-shoot of about 16%. The saliency ratio represents the ratio of q-axis to d-axis inductance, and significantly increases the complexity of the machine model, compared with a non-salient machine. More information regarding IPM machine modelling can be found in [88].



Figure 2.3: Simplified PM machine model, showing the back-EMF voltage source, E, stator reactance, X, variable load resistance, R, and the resulting output current, I, and voltage, V.



(a) I-V locus for surface PM machine (b) Comparison of various machine I-V loci

Figure 2.4: Ideal I-V loci of (a) surface PM machine (SPM) and (b) surface PM, DC and interior PM, IPM, machines. Note that the SPM machine loci is shown for speeds,  $\omega$ , of (a) 1 and 2pu, and (b) 1 pu; the IPM locus is shown for a saliency ratio,  $\xi$ , of 4, and has a 16% voltage over-shoot (peak voltage of 1.16pu).

### 2.1.2 Inductance Classification

Conventional surface PM synchronous machines generally have *low*-inductance stator windings as the permanent magnets mounted on the rotor surface behave as large air gaps in the machine's magnetic circuit [35]. The resulting characteristic current,  $I_{ch}$ , tends to be significantly higher than the machine's rated current, causing limitations on the machine's operation, due to the possibility of high currents. The characteristic current is expressed by Equation (2.2), where  $\Psi_m$  and  $L_s$  represent the RMS magnet flux linkage and stator inductance, respectively.

$$I_{ch} = \frac{\Psi_m}{L_s} \tag{2.2}$$

High stator currents increase copper losses and increase the temperature of the stator windings. Exposure of the rotor magnets to high temperatures for long periods may produce metallurgical changes that may impair the ability of the material to be magnetised, and may even render the magnet non-magnetic [89,90]. The magnet's magnetisation will be reduced to zero if it reaches the *Curie* temperature. Despite this, the magnet may be re-magnetised providing no metallurgical changes have been made.

There is limited opportunity to reduce the magnet flux linkage in a conventional SPM machine, without degrading the machine's torque production capability [35]. Therefore, the  $I_{ch}$  can only be reduced by increasing the stator inductance, which is (conveniently) achieved by using fractional slot concentrated stator windings. The F&P test machine considered here is a *high*-inductance machine, which implies the  $I_{ch}$  is comparable to the machine rated current. One distinct advantage of any high-inductance machine is that the short-circuit or fault current magnitude is limited to roughly the machine's rated current.

A comparison of the operating regions of both high and the more conventional low inductance PM machines is shown in Figure 2.5. Conventional low-inductance PM generators operate at output currents which are small compared to the short-circuit current,  $I_{sc}$ , and output voltages that are close to the open-circuit voltage,  $V_{oc}$ . The output voltage is proportional to the machine speed and hence the generator acts like a variable voltage source. This output voltage variation complicates the design of a power converters, e.g. a voltage-source grid-connected inverter, as these require a constant input voltage. In contrast, a high-inductance PM generator has output voltages which are small compared to its  $V_{oc}$ , whilst the output current is close to the short-circuit current. The machine acts as a constant current source, whose magnitude is unaffected by operating, providing its output voltage is significantly less than its open-circuit voltage.



Figure 2.5: Operating regions of high and low-inductance PM machine, s operating at speeds  $\omega$  and  $2\omega$ . The resulting open-circuit voltages occur at  $V_{oc}$  and  $2V_{oc}$ , whilst the short-circuit current for each case is equal to  $I_{sc}$ . The high-inductance machine operates near  $I_{sc}$  in the constant current region. Low-inductance machines operate close to  $V_{oc}$ , i.e. the variable voltage source region.

# 2.2 Machine Characterisation

The surface PM machine is modelled and simulated, using mathematical analysis and simulation software, prior to laboratory testing. However, the ideal PM machine model is replaced with one that is more realistic, which is later used to predict the machine performance. The detailed machine model, characterisation tests and parameters are discussed below.

# 2.2.1 Realistic Machine Model and Effect on Loci

The ideal surface PM machine model was discussed in Section 2.1.1. Stator resistance is now added to the ideal phase model, which allows copper losses to be modelled. The realistic phase model of the surface PM machine is shown in Figure 2.6(a). The delta connected machine equivalent circuit is shown in Figure 2.6(b), and a simplified model is shown in Figure 2.6(c).



Figure 2.6: Realistic surface PM machine model, showing (a) phase model, (b) delta connected equivalent circuit, and (c) symbolic representation . Note that the machine star point does not exist for a delta connected machine.

The effect of the stator resistance on the I-V locus and the resulting power vs. voltage (P-V) locus is illustrated in Figure 2.7, for stator resistances of 0.1 and 0.2pu. Note that the ideal curves ( $R_s = 0$ pu) are shown in the figure for reference, and that these are indicated by the dashed lines. The figure indicates that the introduction of  $R_s$  does not affect the open-circuit voltage, however, it does affect the machine output current. The short-circuit current is slightly reduced, whilst the machine current and hence power, between the short-circuit and open-circuit operating points, is significantly reduced. This is highlighted in Figure 2.7(b), which shows that a stator resistance of 0.2pu reduces the peak machine output power by 18%, compared to the ideal model ( $R_s = 0$ ). Note that this difference is largely due to the copper loss,  $P_{CU}$ , which is proportional to the square of the RMS current, I, and stator resistance, as summarised in Equation (2.3).



Figure 2.7: Effect of introducing the stator resistance to the ideal model, showing the (a) current vs. voltage (I-V) and (b) the resulting power vs. voltage (P-V) loci, for stator resistances,  $R_s$ , of 0.1 and 0.2 pu. The dashed line represents the loci of the ideal model  $(R_s = 0)$ , and is shown for reference.

$$P_{CU} = I^2 R_s \tag{2.3}$$

The required machine parameters, such as k,  $L_s$  and  $R_s$ , are determined by a series of characterisation tests. The stator resistance is determined by applying a DC current (whilst the machine is stationary) to the stator windings and measuring the resulting voltage drop. In contrast, the back-EMF constant and stator inductance are found by rotating the machine at various speeds; k is found from the *open-circuit test*, whilst  $L_s$  is determined from the *short-circuit test*.

### 2.2.2 Test Arrangement

The surface PM machine is characterised over a wide range of speeds in the laboratory using a directly-coupled, separately-excited 5kW DC machine. The DC machine speed is controlled by adjusting the armature voltage, which is supplied by the rectified output of a 3-phase Variac; it is fine-tuned using a rheostat (variable resistor) that is series connected to the separately-excited field winding. The machine speed is measured using a simple Hall-effect sensor that fits in the PM machine air gap. The sensor delivers a 50% dutycycle square wave voltage to a frequency meter. The machine speed, n, is calculated as shown in Equation (2.4), where P is the number of machine pole-pairs, and f represents the electrical frequency (Hz). Note that the machine speed is calibrated using an optical tachometer.

$$n = \frac{60\,f}{p} \tag{2.4}$$

In addition to the Hall-effect device, the stator of the PM machine is fitted with a reaction torque transducer. This force measurement sensor allows the shaft torque and hence PM machine mechanical input power,  $P_{in}$ , to be accurately measured. The input power is the product of the shaft torque, T, and the machine speed,  $\omega$ , in rad/s, as shown in Equation (2.5). The machine test arrangement and the general block diagram is shown in Figure 2.8.



$$P_{in} = T\omega \tag{2.5}$$

(a) Block diagram

(b) Photograph

Figure 2.8: Laboratory machine test arrangement, showing (a) block diagram, and (b) photograph. The directly coupled DC and PM machines, torque transducer and frequency (speed) sensor are shown in each figure, whilst the photograph also shows the DC power supply (rectified 3-phase variac), rheostat and a load-bank.

### Load Arrangement

The machine is characterised by three tests, these are the DC resistance, open-circuit (OC) and short-circuit (SC) tests. The machine is loaded in various ways during these tests, as summarised by Figure 2.9. Note that the machine is stationary during the resistance test to prevent induced voltages affecting the readings, whilst it is rotated at various speeds during the OC and SC tests. In addition, each line voltage was measured for the resistance and OC tests, whilst each line current was measured for the SC test.



Figure 2.9: PM machine characterisation arrangements, showing the (a) resistance measurement, (b) open-circuit (OC), and (c) short-circuit (SC) arrangements. The measured voltage and currents are represented by V and I, respectively. Note that the machine rotates for the OC and SC tests, and that all line voltages are measured during the resistance and OC tests, whilst all line currents are measured during the SC test.

### 2.2.3 Open-Circuit Test

The generator back-EMF constant is determined from the open-circuit test. The PM generator is driven over a wide range of speeds (50–1000 rpm), whilst the line voltages and input loss torque are measured. This test arrangement is seen in Figure 2.9(b), whilst the measured results are displayed in Figure 2.10. As can be seen from the figure, both the induced voltage (also see Equation (2.6)) and the input torque vary linearly with speed. The difference, however, is that the induced voltage intercepts the origin, unlike the torque plot. The input torque is vertically offset by 0.3Nm at zero speed, which represents the generator starting torque which includes the hysteresis and peak cogging torques, and the torque required to overcome the bearing and seal friction [86]. The measured OC test torque is further discussed in Section 2.2.5.

$$E = k\,\omega\tag{2.6}$$



Figure 2.10: Open-circuit characteristic, showing measured (a) line voltage, and (b) generator torque vs. generator speed. Note that the lines represent curve-fitted lines; the back-EMF constant, k, and torque curve equation is included.

An example of the measured generator open-circuit line voltages is shown in Figure 2.11, for a generator speed of 1000 rpm. The high-quality sinusoidal voltage waveforms shown each contain a small amount of total harmonic distortion (less than 0.5%). The distortion is attributable to the machine design and is hence independent of the generator speed. The term *total harmonic distortion* is defined and further discussed in Section 5.2.



Figure 2.11: Measured generator open-circuit line voltage at 1000 rpm, showing (top)  $V_{AB}$ , and (bottom)  $V_{BC}$ . The zero position for each waveform is shown by the arrow on the left; the vertical and horizontal scales are 50V and 1ms per division, respectively.

## 2.2.4 Short-Circuit Test

The short-circuit (SC) test is performed to calculate the stator inductance. The stator terminals are shorted together and the current is measured whilst the machine is operated up to 1000 rpm (see Figure 2.9(c)). The phase current and generator torque is plotted in Figure 2.12, as a function of generator speed. The current is shown to rapidly increase at low generator speeds, and approach a limit of about 9.26A; this corresponds to the machine's characteristic current (see Equation (2.2)). In contrast, the measured SC machine torque is shown to rapidly increase at low speeds, reaching its peak of approximately 9.1 Nm at 92 rpm. The torque then decreases with increasing machine speed.



Figure 2.12: Short-circuit characteristic, showing predicted and measured (a) phase current, and (b) generator torque vs. generator speed. The measured and predicted data are represented by points and solid lines, respectively.

Note that the fundamental difference between the characteristic and short-circuit current is the inclusion of stator resistance in the calculation of the latter. The ratio of the SC to characteristic current is 0.997:1, at 1000 rpm.

The machine phase reactance,  $X_{ph}$ , and inductance,  $L_{ph}$ , are calculated from the measured SC current, the calculated induced phase voltage,  $E_{ph}$ , and the measured stator (phase) resistance,  $R_{ph}$ . These parameters are given by Equations (2.7) and (2.8), respectively, where  $\omega_e$  is the electrical frequency and is equal to  $2\pi f$ .

$$X_{ph} = \sqrt{\left(\frac{E_{ph}}{I_{ph}}\right)^2 - R_{ph}^2}$$
(2.7)

$$L_{ph} = \frac{X_{ph}}{\omega_e} \tag{2.8}$$

The calculated phase inductance is shown in Figure 2.13 as a function of phase current and generator speed. Figure 2.13(a) shows that the phase inductance decreases at low currents and approaches a value of 2.87mH at high currents, whilst Figure 2.13(b) shows a relatively constant phase inductance over the range of 200–1000 rpm. Given that this is a surface PM machine the inductance is expected to be relatively constant. In addition, the bulk of experimental work presented in this thesis concentrates on a generator speed range of 200–1000 rpm, thus a constant inductance corresponding to the value at high currents was used in the model. This was also used to calculate the simulated curves of short-circuit current and input torque shown in Figure 2.12, and is shown as the dashed line in Figure 2.13.



Figure 2.13: Measured phase inductance vs. (a) phase current, and (b) generator speed characteristic, from the short-circuit test. The dashed line represents the constant inductance used in the modelling.

Note that it is likely that errors in the measurements and models used cause the large inductance variation at low currents. Although this work focusses on the 200–1000 rpm speed range, it is recommended that a phase inductance test is performed at stand still, in the future, to further verify the relatively constant phase inductance.

### Stator Temperature

The high currents under SC or loaded operation produce high copper losses in the stator, which are dissipated as heat throughout the copper windings, causing the stator temperature to increase. In addition to possibly damaging or demagnetising the permanent magnets, high temperatures can damage the copper insulation causing short circuits in the stator windings. Safe operation of the test machine at high currents is tested by measuring the stator temperature during the SC test, using thermocouple that is directly connected to a stator winding. The measured temperature rise of the stator is shown in Figure 2.14 for a generator speed of 1000 rpm, along with a fitted curve, whose time constant is 10 minutes.



Figure 2.14: Stator temperature increase,  $\Delta$ , vs. time under short-circuit conditions for a generator speed of 1000 rpm. The points represents experimentally measured data, whilst the solid line and equation corresponds to a fitted curve.

The maximum temperature rise is shown to be about 25°C, which implies that the machine is able to withstand high currents without compromising the stator insulation or permanent magnet integrity. It also indicates that the stator resistance will increase by about 10.7%, as summarised by Equation (2.9), where  $R_{cold}$ ,  $R_{hot}$ ,  $\Delta T$  and  $\alpha_{cu}$ , represent the initial (*cold*) resistance, the *hot* resistance, the difference between hot and cold stator temperatures, and the temperature coefficient of copper, respectively. Note that  $\alpha_{cu}$  is equal to  $4.27 * 10^{-3}/°C$  at 0°C [91].

$$R_{hot} = R_{cold} \left( 1 + \alpha_{cu} \,\Delta T \right) \tag{2.9}$$

### 2.2.5 Machine Losses

The torque measurements from the open and short-circuit tests allow the machine losses to be calculated. The input power from the OC test gives the machine iron, friction and windage (IFW) losses. The OC input torque curve of Figure 2.10(b) is linearly proportional to the generator speed, and hence the expected IFW (power) losses will be proportional to the square of the machine speed (recall Equation (2.5)). The calculated OC IFW loss is shown in Figure 2.15, as the dashed line.

The figure also shows the total, copper and IFW losses, under SC conditions. The total loss is equal to the machine input power which is calculated from the measured input torque (see Equation (2.5)). The total loss is shown to increase rapidly at low generator speeds and continues to increase monotonically at high speeds, due to the shape of the SC torque characteristic. The copper loss is calculated knowing the stator resistance and the measured current. This loss is expected to increase sharply at low generator speeds, due to the shape current increase (see Figure 2.12(a)). The copper loss is also expected to approach a fixed value at high generator speeds, as the SC current approaches the characteristic current. Note that the stator resistance is assumed constant, although in



Figure 2.15: Calculated generator open and short-circuit losses vs. generator speed. The losses shown include: open-circuit (OC) iron, friction and windage (IFW), the short-circuit (SC) IFW, copper, Cu, and total (Cu+IFW) losses.

practise it will increase due to the stator temperature rise and perhaps skin effect at higher frequencies (see Section 2.4.1).

The SC IFW loss is calculated as the difference between the total loss and the copper loss (total - Cu). This loss is also plotted in Figure 2.15, and is shown to increase sharply at low generator speeds, up to about 150rpm. The SC IFW loss then continues to slowly increase linearly with speed. This is unlike the OC IFW power loss curve which increases with the square of generator speed. Hence, the SC IFW loss will be greater than the corresponding OC loss, at low generator speeds, whilst it will be less than the OC IFW loss at high generator speeds. This is seen in the figure, which shows that the SC and OC IFW losses are equal at about 600 rpm.

# 2.2.6 Machine Properties

A summary of the test PM machine properties is shown in Table 2.1. These include the physical properties, the calculated parameters and a summary of the extreme (maximum) measured conditions.

	Parameter	Value
	Stator Connection	Delta
	Phases	3
Physical	Magnetic Poles	48
	Stator Teeth	36
	Slots per phase per pole	0.25
	Test Speed	1000 rpm
Mossurod	SC Phase Current	9.24 A
(maxima)	Ratio of $I_{SC}$ : $I_{ch}$	0.997:1
(IIIaxIIIIa)	Torque	9.08 Nm
	$\Delta$ Stator Temperature	$23^{\circ}\mathrm{C}$
Maagurad	Back-EMF Constant, $k_{ph}$	0.0668 V/rpm
Dependence	Phase Inductance, $L_{ph}$	$2.87 \mathrm{mH}$
Parameters	Phase Resistance, $R_{ph}$	$0.519 \ \Omega$

Table 2.1: Physical, measured and calculated PM generator properties.

Note: the ratio of short-circuit to characteristic current is shown for a generator speed of 1000 rpm.

# 2.3 Machine Modelling

The test PM generator is simulated over a wide range of speeds and AC and DC load conditions prior to experimental testing to gain an understanding of its power generation capabilities. Two models are used to predict the machine's generating performance, these are the analytical and PSIM<sup>®</sup> based models. The accuracy of the models are examined in Section 2.4, which compares the models to the experimentally measured data.

# 2.3.1 Analytical Model

The test machine is modelled as a synchronous generator with a load resistance,  $R_L$ . The single-phase machine model is shown in Figure 2.16. The machine <u>phase</u> parameters,  $k_{ph}$ ,  $X_{ph}$  and  $R_{ph}$ , are obtained from Table 2.1.



Figure 2.16: PM generator phase model.

The resulting phase current,  $I_{ph}$ , and voltage,  $V_{ph}$ , is calculated using Equations (2.10) and (2.11), respectively. These are converted to line quantities which are used to plot the machine I-V locus. Note that for the delta connected machine the phase and line voltages are equal in magnitude ( $|V_L| = |V_{ph}|$ ), whilst the line current,  $I_L$ , is given by Equation (2.12). The machine power, P, is calculated from Equation (2.13), where  $\cos(\phi)$ represents the power factor. Note that unity power factor is assumed here, as the load is assumed purely resistive. In practise, however, a small amount of inductance exists, which slightly reduces the power factor.

$$I_{ph} = \frac{E_{ph}}{|(R_{ph} + R_L) + j X_{ph}|}$$
(2.10)

$$V_{ph} = I_{ph} R_L \tag{2.11}$$

$$I_L = \sqrt{3} I_{ph} \tag{2.12}$$

$$P = \sqrt{3} V_L I_L \cos(\phi) \tag{2.13}$$

### Normalised AC Voltage Loci

The normalised output I-V and P-V loci are shown in Figure 2.17, for generator speeds of 200–1000 rpm. At each speed the voltage, current and power is normalised to the opencircuit voltage, the characteristic current and the maximum power possible at that speed (with zero stator resistance), respectively. The solid lines represent the predicted machine performance, based on the measured parameters including stator resistance, whilst the dashed line represents the predicted ideal case, i.e. zero stator resistance. It is shown that despite the stator resistance, the machine current and hence power approaches the ideal case as the generator speed increases. This occurs as the stator reactance increases with frequency (machine speed), and becomes the dominating impedance, i.e. the ratio of stator resistance to reactance approaches zero. The resulting current approaches its characteristic value (1pu), as previously shown in Figure 2.7. Note that the following assumptions are made regarding the phase model and the subsequent normalised voltage loci:

- the induced voltages are sinusoidal,
- the stator resistance is fixed, and equal to its DC measured value, and
- the load is purely resistive for all cases.

### **DC** Quantities

The above equations allow the AC I-V and P-V loci to be plotted. However, the equivalent DC voltage loci are also required, as the proposed power converters will use an uncontrolled full-wave three-phase rectifier. These DC loci can be obtained by converting the AC line voltages and currents to equivalent DC quantities. The equivalent DC bus voltage,  $V_{DC}$ , corresponding to an AC line voltage,  $V_L$ , is expressed in Equation (2.14), which assumes the peak phase voltage is equal to  $2/\pi V_{DC}$  [56]. The equivalent DC bus current,  $I_{DC}$ , corresponding to an AC line current,  $I_L$ , is shown in Equation (2.15). The equation is deduced from the three-phase power and DC voltage equations, and assumes a loss-less rectifier (i.e. the AC and DC powers are equal).


Figure 2.17: Normalised PM machine line voltage AC loci, showing (a) current, and (b) power loci, for generator speeds of 200–1000 rpm. The solid lines represent the various machine speed loci, whilst the dashed line corresponds to the ideal I-V locus (zero stator resistance).

$$V_{DC} = \frac{\pi}{\sqrt{6}} V_L = 1.283 V_L \tag{2.14}$$

$$I_{DC} = \frac{3\sqrt{2}}{\pi} I_L = 1.35 I_L \tag{2.15}$$

The resulting DC voltage loci differ from the AC voltage loci, due to the above equations linking the AC and DC quantities. Despite this, the normalised AC and DC voltage loci are identical. Note that the conversion of AC to DC quantities makes the following assumptions, and as such may not be accurate.

- the ratio of DC to AC quantities remain fixed for all conditions, and
- the rectifier voltage drop and resulting power loss is ignored.

# 2.3.2 PSIM<sup>®</sup> Model

A computer based model is used to provide insight in to the rectifier modelling and to verify the analytical AC predictions. PSIM<sup>®</sup> is used as it is designed to simulate power electronics and motor control circuits. Compared to other simulations tools such as SPICE, PSIM<sup>®</sup> simulates faster and avoids convergence problems, as the nodes are analysed using the trapezoidal rule integration algorithm [92]. Figure 2.18 shows the delta connected PM generator with both the three-phase AC and DC (rectifier) resistive loads. The rectifier diodes were modelled with a voltage drop of 1V. The resulting calculated normalised DC I-V and P-V loci are shown in Figure 2.19.



Figure 2.18: PSIM<sup>®</sup> delta-connected PM machine model showing (a) AC, and (b) DC resistive loads. Note that the three-phase load is star connected.



Figure 2.19: Normalised PM machine DC output voltage loci, showing (a) current, and (b) power loci, for generator speeds of 200–1000 rpm. The dashed line represents the ideal analytical (zero stator resistance) model, whilst the solid lines correspond to the various machine speed loci.

The normalised DC loci are shown for generator speeds of 200–1000 rpm, as solid lines, whilst the dashed line corresponds to the ideal analytical case. At each speed the DC output voltage is normalised to the equivalent DC voltage obtained (using Equation (2.14)) with the open-circuit AC voltage. The DC output current is normalised (using Equation (2.15)) with the characteristic current, whilst the power is normalised to the maximum AC power possible at that speed with zero stator resistance.

The current and resulting power are again shown to increase with generator speed, as the ratio of stator resistance to reactance decreases. Despite this, the loci do not appear to approach the ideal analytical curves as much as the AC case, which indicates that the analytical DC model has limited accuracy. This is most likely due to the incorrect assumptions made regarding the rectifier voltage drop, power losses, and AC to DC conversion ratios. The latter is briefly investigated in Section 2.4.2.

In contrast, the AC voltage loci with a three-phase resistive load predicted by PSIM<sup>®</sup> are identical to those predicted using the analytical phase model (see Figure 2.17).

## 2.3.3 Power Maximisation

The maximum normalised power is shown in Figure 2.20, as a function of generator speed. This data is obtained from the AC and DC normalised P-V loci. The figure compares the AC power predicted by both models, to the DC power predicted by PSIM<sup>®</sup>. Note that the power is normalised relative to that obtained from an ideal (zero stator resistance) PM generator with a three-phase resistive load. This is represented by the dashed line in



Figure 2.20: Model comparison showing the peak normalised DC output power vs. generator speed prediction, based on the analytical and PSIM<sup>®</sup> models. The dashed line represents the peak power obtained using the ideal analytical model.

the figure. Although both models predict the same AC loci, PSIM<sup>®</sup> consistently predicts about 0.1pu less power than the analytical model for the DC loci. This verifies that power is lost in the rectifier. This conduction loss is caused by the diode voltage drop, and is proportional to the rectifier output current.

#### 2.3.4 Model Comparison

The normalised AC and DC voltage loci predicted by PSIM<sup>®</sup>, for a generator speed of 600rpm, are compared in Figure 2.21. The ideal analytical case, corresponding to zero stator resistance and a loss-less rectifier (for the DC case), is also shown for reference. The effect of stator resistance is clearly seen, as is the reduced current and hence power for the DC case. This is not caused by stator resistance, and is likely to be caused by the assumptions used to predict the DC quantities (i.e. fixed AC to DC voltage and current ratios and the use of a loss-less rectifier), which have only limited validity. Both the analytical and PSIM<sup>®</sup> predictions should be compared with experimentally measured data, to validate their respective models.



Figure 2.21: Comparison of ideal analytical and PSIM<sup>®</sup> models, showing (a) I-V, and (b) P-V loci, for both AC and DC cases for a generator speed of 600rpm. The ideal analytical predictions assume zero stator resistance and loss-less rectifier (for the DC case), and are represented by the dashed line.

# 2.4 Resistive Load Testing

The PM generator is loaded using a variable resistance, to verify the analytical and PSIM<sup>®</sup> models. It is first loaded by a three-phase (AC) variable resistive load, and secondly by a rectifier (DC) variable resistive load. The DC loading is achieved using an uncontrolled full-bridge three-phase rectifier, and along with the AC loading is shown in Figure 2.22. Note that the AC resistive load was connected in both the star and delta configurations to achieve a wider range of load resistances, as the delta connection effectively reduces the resistance to one third that of the equivalent star case.



Figure 2.22: PM machine load arrangements, showing the (a) three-phase AC, and (b) DC resistive load arrangements. Note that the AC resistive load is shown for both star and delta connections, and that the dashed box represents the uncontrolled rectifier, as used for the DC resistive load case.

# 2.4.1 3ph Resistive Loading

The test machine is rotated at speeds of 200, 400, 600, 800 and 1000 rpm, whilst the load resistance is varied from very low to high resistances, i.e. 0.075 to 350Ω, respectively. The DC machine speed is kept constant via the field rheostat, as the varying load changes the PM generator output power and hence torque. The resulting measured current vs. voltage (I-V) and power vs. voltage (P-V) loci are shown in Figure 2.23 for the generator speeds mentioned above. The figure includes both the analytical and PSIM<sup>®</sup> model simulations, and the measured experimental data. Note that the currents and voltages are expressed as line quantities, whilst the power shown corresponds to total (three-phase) power.



Figure 2.23: PM machine line voltage AC loci, showing (a) current, and (b) power loci, for generator speeds of 200–1000 rpm. The points represent measured data, whilst the lines represent the analytical model and (matching) PSIM<sup>®</sup> simulations.

#### **Increasing Stator Resistance**

The generator I-V locus indicates that the simulations and experimentally measured data closely match, however, small discrepancies exist. These and are more clearly seen on the P-V locus, and are shown to increase with generator speed (frequency). This may be associated with an increase in stator resistance caused by the increase of stator temperature, and the *skin* and *proximity* effects. The rise in stator temperature accounts for a maximum stator resistance increase of 10% (see Section 2.2.4).

The skin effect causes current to flow on the outside of the conductors, which effectively reduces the conducting cross-sectional area (skin depth) and hence increases the resistance. The skin depth,  $\delta$ , is defined as the depth below the surface of the conductor at which the current density decays to 1/e (about 0.368) of the current density at the surface [67,93]; it is inversely proportional to square-root of the electrical frequency, f, the permeability,  $\mu$ , and conductivity,  $\sigma$ , of the conductor, as summarised by Equation (2.16).

$$\delta = \frac{1}{\sqrt{\pi f \,\mu \,\sigma}} \tag{2.16}$$

The proximity effect is related to the current distribution in the conductor to changing due to magnetic fields, and can be reduced by positioning the coil away from the slot opening [94]. The use of *Litz* wire, thin insulated strands bundled together, reduces both the skin and proximity effects. Despite this, Litz wire may not be a desirable solution due to its high cost, relatively poor thermal conductivity and low packing factor [94].

#### **Generator Efficiency**

The generator efficiency,  $\eta$ , is calculated as output power divided by input power, as summarised and simplified in Equation (2.17). The mechanical input power is obtained using the torque transducer and generator speed, whilst the electrical output power is measured using a power analyser. Note that the three-phase resistive loads contain only a negligible amount of inductance. This is verified by the power analyser which repeatedly measured a load power factor of 0.999 lag.

$$\eta = \frac{P_{out}}{P_{in}} = \frac{\sqrt{3} V_L I_L \cos(\phi)}{T \,\omega} \tag{2.17}$$

The measured generator efficiency is plotted as a function of total (3-phase) output power, in Figure 2.24, for generator speeds of 200–1,000rpm. The peak generator efficiency increases from about 83%, at 200rpm, to approximately 87% at 1,000rpm. The P-V locus (Figure 2.23) shows that the same output powers can be obtained, for a fixed generator speed, with two values of load resistance. The same output power can be obtained with a low resistance producing a high current and low voltage, as with a high resistance producing a high voltage and low current. This implies that for each speed, the generator will have two efficiency vs. power curves which meet at the peak power. This is shown in the figure below, and is more clearly seen for the high speed cases. The lower efficiency curve corresponds to operation with higher currents and hence copper losses, which is associated with operation in the constant current region of the generator I-V locus.



Figure 2.24: Measured generator efficiency vs. output power, for speeds of 200–1,000rpm.

#### 2.4.2 DC Resistive Loading

The test generator is again loaded by a variable resistance, however, an uncontrolled rectifier is used to obtain a DC output voltage, current, and power. The resulting DC I-V and P-V loci are shown in Figure 2.25, which includes the analytical and PSIM<sup>®</sup> model based simulations, and the measured experimental data. There is a close match between the measured results and the PSIM<sup>®</sup> simulations, giving confidence in the model.



Figure 2.25: PM machine DC output voltage loci, showing (a) current, and (b) power loci, for generator speeds of 200–1000 rpm. The points represent measured data, whilst the dashed and solid lines represent the ideal analytical model and PSIM<sup>®</sup> simulations, respectively.

In particular, the PSIM<sup>®</sup> model accurately predicts the non-linearity in the rectified I-V locus near the open-circuit point. This non-liberality is likely to be associated with discontinuous input currents under this condition.

#### Power Maximisation

The normalised generator I-V and P-V loci of Figure 2.7 indicate that the maximum generator output power ideally occurs for an output current and voltage of 0.71pu  $(1/\sqrt{2})$ . The roughly constant short-circuit current and the open-circuit voltage, which is proportional to speed, implies that the peak output power will increase linearly with generator speed. The peak DC output power can also be assumed to be linearly proportional to the generator speed, as the rectifier has a fixed voltage drop and hence power loss. This can be seen from the P-V locus above, but is more clearly seen in Figure 2.26, which shows the both the analytical and PSIM<sup>®</sup> DC predictions, as well as the peak measured power. Note that the peak measured power is slightly less than that predicted by PSIM<sup>®</sup>. This may be due to the limited range of resistances, which do not necessarily correspond to the peak DC output power. This is evident for the 600rpm P-V locus.



Figure 2.26: Maximum DC output power vs. generator speed. The points represent the experimentally measured data, whilst the solid and dashed lines correspond to the PSIM<sup>®</sup> and analytical model based simulations, respectively.

#### Modelling Discrepancies

The discrepancies between the analytical and PSIM<sup>®</sup> DC output predictions are clearly seen in Figures 2.25 and 2.26. This, along with the close match between the experimentally measured data and the PSIM<sup>®</sup> simulations indicate the simplifying assumptions, used with the analytical model, have only limited validity. This is briefly investigated here using a power analyser to compare the rectifier AC input and DC output current and voltage magnitudes. The rectifier output (DC) current is found to closely match that predicted by the analytical model seen in Equation (2.15), however, the DC voltage does not match the analytical prediction seen in Equation (2.14). This is likely due to the assumption that the generator currents and voltages are sinusoidal using the rectifier. In addition, it was shown in [63] that diode commutation, which can take up to 1ms at rated current, accounts for a lower mean voltage (on the DC link) than that expected.

The measured ratio of DC to RMS AC line voltage is shown in Figure 2.27, for generator speeds of 200–1,000rpm. The solid line in the figure represents the fixed AC to DC voltage ratio of the analytical model, whilst the dashed line corresponds to the calculated DC to RMS line voltage ratio. The calculated ratio assumes a sinusoidal voltage source



Figure 2.27: Measured DC to RMS line voltage ratio, for generator speeds of 200–1,000rpm. The solid line represents the fixed ratio, as used by the analytical model, whilst the dashed line represents the calculated ratio.

input, a fixed rectifier voltage drop (1V/diode) and a purely resistive load. This calculated ratio is shown to increase rapidly at small line voltages values, and becomes steady at high voltages, where it approaches 1.347. This occurs as the rectifier voltage drops are significant at low voltages, whilst these are negligible at high line voltages.

Although the measured ratio is shown to increase with line voltage in a similar manner to that of the calculated ratio, it approaches a value of about 1.2, which is somewhat less than the calculated limit of 1.347. In addition, the measured ratio is shown to rapidly increase once the line voltage exceeds about 0.91pu of its open-circuit voltage (see Figure 2.25(a)). The ratio then reaches a peak of 1.38 at open-circuit voltage, for each generator speed. This phenomenon is likely to be caused by the non-sinusoidal machine currents near the open-circuit point. Further investigation is required, however, it is beyond the scope of this work.

#### Generator + Rectifier Efficiency

The generator and rectifier efficiency is measured using the power analyser, and is shown in Figure 2.28, as a function of generator speed and DC output power. The combined efficiency appears similar to the generator (AC) efficiency (see Figure 2.24). However,



Figure 2.28: Measured generator and rectifier efficiency vs. DC output power, for generator speeds of 200–1,000rpm.

the efficiency is reduced at low powers (and hence generator speeds), due to the fixed maximum rectifier loss, which is more significant at low rectifier output powers. The rectifier loss is proportional to the fixed diode voltage drops and rectifier output current.

Note that the rectifier efficiency is combined with the generator efficiency here as both the DC-DC and the DC-AC power converters proposed in this study contain an uncontrolled full-bridge three-phase rectifier. The combined peak efficiency increases from 72%, at 200rpm, to 84% at 1,000rpm; compared to the previously measured peak generator efficiencies of 83 to 87%, over the same speed range (see Figure 2.24).

# 2.5 Chapter Summary

The use of a fractional-slot concentrated winding gives PM machines a high-inductance characteristic, which allows the machine to operate at currents close to the short-circuit current for prolonged periods of time. This was experimentally verified, which found that the temperature of the tested PM generator stator winding increased by about 25°C, after operating under three-phase short-circuit conditions at high speed for 40 minutes. This temperature rise corresponds to a stator resistance increase of 10.7%.

The high-inductance PM generator current vs. voltage (I-V) loci and hence power vs. voltage (P-V) loci are simulated based on simplified analytical modelling using the single-phase synchronous generator equivalent circuit. A second model based on a numerical circuit simulation software package, PSIM<sup>®</sup>, is also used. Both models predict similar generator AC output current and power vs. voltage loci, which closely match the experimentally measured data, especially at low generator speeds. Despite this, discrepancies exist, which increase with generator speed. These may be are caused by increasing stator temperature (and hence resistance), and by skin and proximity effects.

In contrast, when predicting the uncontrolled rectifier DC output characteristic, the analytical model, which is based on a converting AC to DC quantities by using a fixed ratio and assuming a loss-less rectifier, does not match the PSIM<sup>®</sup> based simulations. The PSIM<sup>®</sup> predictions match the experimentally measured data. This indicates the simplifying assumptions used in the DC analytical model have only limited accuracy. It is interesting to note that comparisons of the measured rectifier AC input and DC output quantities show that the AC to DC current ratio is fixed, whilst the AC to DC voltage ratio varies with the AC voltage.

The P-V loci indicate that output powers below that of the peak, can be obtained using two different voltage/current combinations. Therefore, two efficiency vs. power curves exist for each speed, which meet at the peak power. Regrettably, the least efficient curve corresponds to the constant-current region of the I-V locus, which is the proposed operating region, as the higher currents cause higher copper losses.

# CHAPTER 2. HIGH INDUCTANCE PM GENERATOR CHARACTERISTICS

# Chapter 3

# Switched-Mode Rectifier Operation

This chapter discusses the equivalent circuit, modelling and operating regions of the switched-mode rectifier (SMR). The SMR is experimentally tested with the high-inductance PM generator from Chapter 2, using a dynamometer and small wind tunnel. The SMR duty-cycle is varied in an open-loop manner using the dynamometer and wind tunnel, which allows the generator and wind speeds to be controlled, respectively. The SMR is shown to effectively control the PM generator such that maximum power is obtained at low wind speeds, and that the turbine speed is limited under high wind speed conditions.

# **3.1** Introduction

The switched-mode rectifier is a simple single-switch power converter that converts the variable frequency and magnitude generator output voltage to a fixed DC voltage. It consists of a three-phase uncontrolled full-wave rectifier, a high-speed switch that shorts the rectifier output at a duty-cycle, d, a reverse-blocking diode and a storage capacitor, as summarised in Figure 3.1. The SMR is similar in appearance and performance to a boost converter with an uncontrolled rectifier, however, it requires fewer passive components, as it uses the phase inductance of the high-inductance PM generator, rather than an external inductance (recall Figure 1.9).

The SMR can deliver a controlled output current to a voltage-source load, and is hence suitable for use as a battery charger. The fixed output voltage,  $V_{SMR}$ , (often referred to as the DC link voltage) together with the parallel SMR switch, and the machine phase inductance allows the boosting action to occur. This effectively allows the rectifier output voltage (and hence generator output voltage) to be reduced below the DC link voltage.



Figure 3.1: SMR equivalent circuit showing three-phase uncontrolled rectifier, switch, blocking diode and storage capacitor. The rectifier input and output, and SMR output currents and voltages are shown using subscripts  $_L$ ,  $_R$ , and  $_{SMR}$ , respectively. The SMR switch duty-cycle, d, is also shown.

The rectifier voltage,  $V_R$ , is linearly proportional to the duty-cycle, and the constant DC link voltage, as summarised in Equation (3.1). Similarly, the SMR output current,  $I_{SMR}$ , is proportional to the rectifier output current,  $I_R$ , and the duty-cycle, as shown in Equation (3.2). This is derived assuming the SMR is loss-less, i.e.  $V_{SMR}.I_{SMR} = V_R.I_R$ . These ideal rectifier and SMR output voltage and current relationships are summarised in Figure 3.2. Note that the rectifier current and DC link voltage are both assumed to be constant and equal to 1pu, and that the rectifier current and voltages are proportional to the generator line currents and voltages, as discussed in Chapter 2.

$$V_R = (1 - d) V_{SMR}$$
 (3.1)

$$I_{SMR} = (1 - d) I_R$$
 (3.2)

#### 3.1.1 Switched-Mode Rectifier Model

A switched-mode rectifier can be modelled as an uncontrolled rectifier with a voltagesource load, as the output voltage of the SMR is essentially constant. This model can be approximated by an uncontrolled rectifier with a resistive load, and thus further modelled as a three-phase resistive load. This approach is based on the assumptions that the rectifier forces the generator output current and voltage to be in-phase and that the current remains sinusoidal despite the non-sinusoidal voltages [56]. Using these assumptions,



Figure 3.2: Ideal SMR and rectifier output (a) voltages, and (b) currents vs. duty-cycle. The SMR output voltage,  $V_{SMR}$ , is fixed due to the voltage-source load, whilst the rectifier output current,  $I_R$ , is assumed constant.

the DC output voltage / current characteristic of the SMR can be modelled using the DC analytical model described in Section 2.3.1. Note that the DC load resistance is effectively varied by adjusting the duty-cycle.

Despite this, the DC analytical model was previously shown in Chapter 2 to significantly differ from the PSIM<sup>®</sup> simulations. In addition, the PSIM<sup>®</sup> predictions were shown to closely match the experimentally measured DC resistive load test data, giving confidence in this model. Therefore the remainder of simulations presented in this thesis are predicted using PSIM<sup>®</sup>. The PSIM<sup>®</sup> model of the SMR is shown in Figure 3.3, which includes the delta-connected PM (test) generator, and a voltage-source load. Note that the voltage source load fixes the voltage drop across the load resistance, and consists of a fixed load resistance, voltage source and reverse-blocking diode.



Figure 3.3: PSIM<sup>®</sup> model of the delta-connected PM generator and switched-mode rectifier connected to a voltage-source load. An IGBT is used as the SMR switch.

# 3.1.2 SMR Operation

The SMR operates in a similar manner to a PM generator and rectifier operating in to a voltage-source load, where the load voltage can be changed by varying the duty-cycle. Note that the high-frequency switching of the SMR, together with the machine inductance, causes the PM machine to see only the time averaged effect of the PWM switching. Hence, the SMR operation is described and compared to that of a PM machine generating into a DC voltage-source load (e.g. battery), using an uncontrolled three-phase rectifier. This is commonly referred to as uncontrolled generation (UCG) [56].

#### **Uncontrolled Generation**

Consider a PM machine that is connected to a rectifier and operates into a voltage-source load. As the speed increases from standstill, the DC output current remains at zero until the speed at which the induced back-EMF is equal to the DC link voltage. Beyond this speed the generator output current increases rapidly and asymptotes towards its characteristic current (Equation (2.2)) at high speeds. This occurs as the output voltage remains fixed, whilst both the induced voltage and stator inductance increase with speed, i.e. the machine approaches the short-circuit operating condition.

An example of a PM generator operating under the UCG condition is shown in Figure 3.4, which shows the rectifier output voltage, current and power vs. generator speed, for DC link voltages of 0.5, 1, 1.5 and 2pu. Consider the 0.5pu DC link voltage case. It is seen that the induced and DC link voltages are equal at a speed of 0.5pu, after which current begins to flow. This highlights that the machine is able to deliver reduced power at lower speeds and hence induced voltages, providing the DC link voltage is reduced. Similarly, more output power is obtained if the DC link voltage is increased at high generator speeds.

The figure also highlights that the maximum output power increases linearly with speed. This occurs as the induced back-EMF voltage and stator reactance are proportional to machine speed. In addition, the DC link voltage must also be increased with speed to maximise the rectifier output power. This is equivalent to matching the effective three-phase load resistance to the magnitude of the stator reactance. This concept is referred to as *load matching*, and allows maximum power to transfer from the source to the load. Maximum power corresponds to a current of 0.71pu, which occurs at a speed of 1.41 V pu.



Figure 3.4: Ideal (a) voltage, (b) current, and (c) power, of a generator operating in to a rectifier and a voltage-source load with DC link voltages of 0.5, 1, 1.5 and 2pu. The solid line in (c) represents the maximum power, which increases linearly with speed.

#### **SMR** Operating Regions

The above figure shows that output power can be obtained even at low generator speeds, by using a low DC link voltage. This, however, limits the power at high generator speeds. The SMR overcomes this by adjusting the rectifier voltage (via the duty-cycle) as a function of generator speed, which effectively allows the SMR to *load-match* the PM generator and load. This allows the generator to transfer maximum power to the voltage-source load (battery), whilst operating into a fixed output voltage. This is referred to as the *Boost Rectifier* operating mode, and occurs for generator speeds below 1.41pu, where a higher output power is obtained, compared to that using an uncontrolled rectifier.

In contrast, the induced back-EMF voltage significantly exceeds the DC link voltage at high speeds, and the PM generator can be considered as a constant current source under these conditions. The SMR output current and hence power can be reduced below that of an uncontrolled rectifier, by adjusting the duty-cycle. This is hence referred to as the *Current Chopper* mode. Both operating modes and the regions they occupy are summarised in Figure 3.5. In addition, the dashed line represents the power obtained using an uncontrolled rectifier operating into a voltage-source load of 1pu; it is shown for reference. Note that the SMR output power can be controlled such that any amount of power within the shaded regions is transferred from the generator to the load.



Figure 3.5: SMR operating areas, including the boost rectifier and current chopper regions. The dashed line represents the power obtained using an uncontrolled rectifier, and is shown for reference. The SMR output power can be controlled to any value within the shaded regions, by adjusting the duty-cycle.

#### 3.1.3 SMR Properties

The SMR used was designed and constructed for another application, and as such was designed with additional features such as over-voltage protection and dummy-load switching capabilities, which are beyond the scope of this work. The SMR components of interest include the IGBT, reverse-blocking diode, and the filter capacitors ( $2 \times 470 \mu$ F). A summary of the components a photograph showing these and other aspects of the SMR, are shown in Table 3.1 and Figure 3.6, respectively.

Component Property	Rectifier	IGBT	Diode	Capacitor
Manufacturer	SEMIKRON	IR	IR	BHC Components
Part Number	SKD230/16	IRG4PC40F	40EPF10	ALS31A471DE400
Rating	230A, 1600V	27A,600V	40A, 1000V	$400V (470 \mu F)$
Voltage Drop	1V per diode	$1.50\mathrm{V}$	$1.25\mathrm{V}$	-
Maximum ESR	_	-	-	$0.267\Omega$

Table 3.1: SMR component properties.

Note: IR and ESR denote International Rectifier, and equivalent series resistance, respectively.



(a) Components

(b) Control circuitry

Figure 3.6: Photograph of SMR (a) semiconductor components, and (b) control and over-voltage protection circuitry.

# 3.2 Dynamometer Testing

The SMR is experimentally tested using the same PM generator characterised in Chapter 2, as its input. The generator is directly coupled and driven by a 5kW DC machine using a dynamometer test rig (see Section 2.2.2). The generator speed is controlled by adjusting the DC machine armature voltage and field current. The SMR is tested by varying its duty-cycle over a wide range of fixed generator speeds.

## 3.2.1 Test Arrangement

The PM generator and SMR test arrangement is shown in Figure 3.7. The SMR is connected to a voltage-source load that simulates a 42V battery. This DC link voltage was chosen primarily as this is a multiple of 14V, which is required to charge a standard 12V battery. In addition, this voltage exceeds the rectified open-circuit voltage of the PM machine for speeds below about 450rpm (see Section 2.4.2). This is useful as it demonstrates how the SMR will operate in the boost rectifier operating mode under low wind speed (generator speed) conditions.



Figure 3.7: SMR dynamometer test arrangement, showing the PM generator, switchedmode rectifier, and the voltage-source load (VSL). The VSL consists of a DC voltage source, reverse-blocking diode and a load resistance; its output voltage is set to 42V.

## 3.2.2 SMR Test Results

The SMR is loaded by the voltage source whilst the PM machine is rotated at various speeds. The load voltage is fixed during the SMR tests, i.e. the SMR duty-cycle and machine speed are the only adjustable parameters. The SMR duty-cycle is swept from 0 to 100% for generator speeds of 200, 400, 600, 800 and 1,000rpm.

The simulated and measured rectifier voltages are shown in Figure 3.8 as a function of duty-cycle, d, and generator speed. The rectifier voltage is shown to be linearly proportional to (1-d), for speeds of 600rpm and above. This occurs as the rectified open-circuit voltage is greater than the DC link voltage for speeds above about 500rpm. In contrast, the rectified generator voltage is linearly proportional to (1-d) only for higher duty-cycles for the 200 and 400rpm cases. The voltage then sharply asymptotes towards its maximum value which corresponds to the rectified generator open-circuit back-EMF voltage. The rectifier can not transfer power to the SMR as its voltage is less than that set by the required voltage given by Equation (3.1). Even when the SMR duty-cycle is further reduced, the rectifier voltage remains unchanged for these generator speeds. This is highlighted in the figure for the 200 and 400rpm cases, which shows that the rectifier voltage must be less than 16.5 and 33V to transfer power from the generator to the SMR, respectively.



Figure 3.8: Rectifier output voltage vs. SMR duty-cycle, for generator speeds of 200–1,000 rpm. The solid lines represent  $PSIM^{\textcircled{B}}$  simulations, whilst the points correspond to measured data. Note that the simulated and measured data for the 600–1000 rpm cases are identical and vary linearly with (1-d), whilst the rectifier voltage reaches its maximum value at duty-cycles of 60 and 20%, for the 200 and 400 rpm cases, respectively.

Note that the DC link voltage seen at the rectifier is offset by the SMR reverse-blocking diode forward voltage drop (1.25V). In addition, the rectifier voltage is limited at high duty-cycles by the IGBT voltage drop (1.5V).

The above figure indicates that output current and hence power will only be obtained for a duty-cycle greater than 60 and 20%, for speeds of 200 and 400rpm, respectively. This is verified in Figure 3.9, which shows the rectifier output current vs. duty-cycle characteristic, for the generator speeds mentioned above. The figure clearly shows that current ceases to flow for the 200 and 400rpm cases, if the duty-cycle is set below 60 and 20%, respectively. This, together with Figure 3.8 indicates that the SMR operates in the boost rectifier region for generator speeds less than 500rpm, as the rectified induced back-EMF voltage is less than DC link voltage for these speeds. Power is transferred to the SMR as the DC link voltage is effectively reduced by increasing the duty-cycle. In contrast, the SMR operates in the current chopper region for generator speeds above 500rpm, as the rectified induced back-EMF voltage exceeds that of the DC link.



Figure 3.9: Rectifier output current vs. SMR duty-cycle, for generator speeds of 200–1000 rpm. The points represent measured data, whilst the solid lines correspond to PSIM<sup>®</sup> simulations.

The above figure also shows that the PM generator and rectifier start to operate as a constant current source at 1,000rpm. This occurs as the induced back-EMF voltage is significantly larger than the DC link voltage at this speed, and implies that the generator's ability to provide a relatively constant current decreases with speed, i.e. due to the decreasing induced voltage (see Section 2.1.2). The SMR output current is shown in Figure 3.10 as a function of duty-cycle for the various generator speeds tested. It is expected that the current varies linearly with (1-d) for a constant rectifier current (see Equation (3.2)). This is seen for the 1,000rpm case, as the induced back-EMF is significantly greater than the SMR output voltage, except at low duty-cycles. The generator behaves less ideally as the speed and hence back-EMF voltages decrease, resulting in a substantial drop in the rectifier current at low duty-cycles. Note that the SMR is able to maximise its output current and hence power (see Figure 3.11) for generator speeds of 800rpm or less. Note also that the SMR output power and current plots are identical in shape, as the SMR output voltage is fixed by the voltage-source load. In addition, the SMR output power curves appear similar to the DC resistive load test loci of Figure 2.25(b), for DC voltages between 0 and 42V. The difference, however, is that the x-axis is reversed due to the (1-d) relationship stated in Equation (3.1).



Figure 3.10: SMR output current vs. duty-cycle, for generator speeds of 200–1000 rpm. The points and solid lines represent measured data and PSIM<sup>®</sup> simulations, respectively.

The SMR output power is shown to vary with speed and duty-cycle, which implies that the rectifier and generator output power must also vary with these parameters. Therefore, the generator input power and hence generator torque must also vary with speed and duty-cycle. This is verified in Figure 3.12, which shows the measured machine torque as a function of duty-cycle and generator speed. The torque plot for the 200rpm



Figure 3.11: SMR output power vs. duty-cycle, for generator speeds of 200–1000 rpm. The solid lines and points represent PSIM<sup>®</sup> simulations and experimentally measured data, respectively.



Figure 3.12: Measured torque vs. duty-cycle, for generator speeds of 200–1000 rpm.

case appears similar to that of the short-circuit test (Figure 2.12(b)), however, the x-axis is again reversed. The torque curve peak broadens and shifts to the left as the generator speed increases. This implies that in addition to the SMR output power, the machine torque can be controlled by adjusting the duty-cycle, for a fixed generator speed.

#### Power and Torque Maximisation

Figures 3.11 and 3.12 indicate that the SMR duty-cycle corresponding to maximum SMR output power and generator torque, varies with generator speed. This is more easily seen in Figure 3.13, which also shows that output power will only be produced for generator speeds above 56rpm. This is because 56rpm corresponds to the generator speed at which the induced voltage exceeds the rectifier and SMR reverse-blocking diode forward voltage drops. In addition, the SMR can only maximise the generator power by varying the duty-cycle for generator speeds less than 824rpm, as this corresponds to the speed at which the peak rectified generator back-EMF voltage equals that of the DC link. As such, the SMR will extract maximum generator power for duty-cycles of 0% for all generator speeds above 824rpm (see Figure 3.5). This implies that the SMR operates in the boost rectifier mode for generator speeds above 824 rpm. Note that this range of speed is dependent on the DC link voltage, as shown in Figure 3.13(a) for a reduced  $V_{DC}$  of 28V.



Figure 3.13: Duty-cycle corresponding to maximum (a) SMR output power, and (b) generator input torque, vs. generator speed. The solid lines and points represent PSIM<sup>®</sup> simulations and measured data, respectively. The simulations in (a) correspond to DC link voltages of 42 and 28V.

The duty-cycle corresponding to maximum generator torque is also linearly proportional to the generator speed. Similarly, Figure 3.13(b) shows that the generator torque can be maximised (by adjusting the duty-cycle) for generator speeds between 130 and 870rpm, i.e. the generator operates below its peak torque outside of this speed range. Note that this range of speed also varies with the DC link voltage.

#### **SMR Efficiency**

The measured SMR efficiency is shown in Figure 3.14, as a function of SMR output power. The efficiency is plotted for each generator speed, and its peak is shown to increase with generator speed from 82.9% at 200rpm to 94.0% at 600rpm. The 200, 400 and 600rpm cases show two efficiency vs. power plots that meet at the peak power. This was previously seen in Figure 2.28, which occurs as the same power can be obtained for two different rectifier voltages (or duty-cycles in this case). Note that only one efficiency vs. power plot exists for the 800 and 1,000rpm cases as the voltage corresponding to peak DC power occurs at voltages greater than the DC link. This is seen in Figure 2.25(b), which shows that maximum power occurs at speeds of 800 and 1,000rpm when the DC voltages are 42.2 and 56.2V, respectively.



Figure 3.14: Measured SMR efficiency vs. power characteristic, for generator speeds of 200–1000 rpm.

#### System Efficiency

The measured torque allows the generator input power to be calculated, which is used to determine the total system efficiency. The system efficiency, i.e. that of the SMR and the PM generator, is shown in Figure 3.15. The figure shows that the peak efficiency increases from 65.9% at 200rpm to 80.9% at 600rpm, whilst it decreases slightly to 79.1% for the 1,000rpm case. This data indicates that the generator alone has a peak efficiency of approximately 79.5, 86.1 and 87.0%, for speeds of 200, 600 and 1,000rpm, respectively.



Figure 3.15: Measured total efficiency vs. SMR output power characteristic, for generator speeds of 200–1000 rpm. Note that the total efficiency includes the PM generator and SMR.

### 3.2.3 Summary of Testing

The SMR has demonstrated its ability to maximise the generator output power and control its input torque, over a wide range of speeds using the dynamometer test arrangement. This is important as the SMR is required to maximise the turbine power and limit its speed, at wind speeds below and above rated, respectively. Further testing, using a wind tunnel, is required to verify the SMR's ability to sufficiently control a small-scale wind turbine.

# 3.3 Wind Tunnel Testing

The SMR's ability to control a commercially available wind turbine, purchased from EcoInnovation<sup>®</sup> [95], was tested using a small wind tunnel. This allowed the wind speed to be controlled, whilst the SMR duty-cycle was varied. The generator used for this turbine is the F & P outer-rotor surface PM machine, which was tested in Chapter 2 and in Section 3.2. The generator (turbine) speed is again measured using the Hall-effect sensor, however, the torque transducer was unable to be fitted to the turbine. Therefore, the torque was calculated from other measured parameters and is used to estimate the turbine input power and hence blade coefficient of performance characteristics.

## 3.3.1 Test Arrangement

The wind tunnel test setup, made available by the School of Civil Engineering of the University of Adelaide, is predominantly used for fluid mechanics studies. The wind tunnel aperture diameter is 1m, which is less than the turbine blade diameter of 1.8m. Due to time and resource limitations, the turbine was mounted to a nearby tank, which is located approximately 4 metres from the wind tunnel, and shielded the lower half of the turbine. The turbine was fitted with a carbon brush / slip ring arrangement to allow yaw movement, however, the tail vane was fixed to the tank shortly after initial testing. This was required as the non-ideal test arrangement caused the turbine to oscillate significantly, which affected the turbine orientation (yaw) and wind turbine test results. A block diagram and photograph of the test setup is shown in Figure 3.16.



(a) Block diagram ( $\underline{not \ to \ scale}$ )

(b) Photograph

Figure 3.16: Wind tunnel test arrangement, showing (a) block diagram, and (b) photograph of the EcoInnovation<sup>®</sup> [95] wind turbine and the available wind tunnel. Note that the turbine and its tail vane are fastened to the water tank.

#### Wind Speed Estimation

Many factors, including nearby obstacles (not shown in Figure 3.16), reflections from the mounting tank, and the large distance between the wind tunnel and turbine, cause the wind speed to be non-uniform within the turbine blade-sweep area. An example of this is shown in Figure 3.17(a) for a wind tunnel (internal) wind speed of 17m/s. The wind speed is measured at 10 points around the blade circumference, using an anemometer. The turbine equivalent wind speeds,  $v_{eq}$ , is approximated as the cube-root of the mean of the cube of the measured wind speeds,  $v_i$ , as the wind power is proportional to the cube of the wind speed. This is summarised in Equation (3.3), where n = 10. The wind tunnel was operated at (internal) wind speeds of 14, 15, 16, 17 and 18m/s, which correspond to equivalent wind speeds of 4.00, 4.37, 4.68, 4.83 and 4.96 m/s, respectively, as shown in Figure 3.17(b). The results show another limitation of the test arrangement in that only relatively low equivalent wind speeds (at the turbine) could be achieved. Note that this range of wind speeds was used in the tests, as 14m/s corresponds to the turbine cut-in speed, whilst 18m/s is the wind tunnel's rated wind speed.

$$v_{eq} = \sqrt[3]{\frac{1}{n} \sum_{i=1}^{n} v_i^3}$$
(3.3)



Figure 3.17: Equivalent wind speed calculation, showing (a) measured wind speed around the blade-sweep area for wind tunnel (internal) speed of 17m/s, and (b) the estimated equivalent wind speed vs. internal wind speed characteristic.

#### 3.3.2 Turbine Coefficient of Performance

The turbine is operated at the five wind speeds mentioned above, whilst the SMR dutycycle is varied from 0 to 100%, at increments of 5%. The PM generator is again connected to the SMR, which has a DC link voltage of 42V. The turbine speed, SMR output current and duty-cycle are recorded for each test, as these parameters allow the turbine (generator) input power to be estimated (based on the results of Section 3.2.2). This allows the turbine coefficient of performance,  $c_p$ , to be determined, based on the equivalent wind speed. The  $c_p$  is calculated from Equation (3.4), where  $\rho$ , r,  $P_{SMR}$  and  $P_{loss}$  represent the air density, blade radius, SMR output power, and estimated SMR / generator power losses, respectively.

$$c_p = \frac{2\left(P_{SMR} + P_{loss}\right)}{\rho \pi r^2 v_{eq}^3}$$
(3.4)

The resulting estimated  $c_p$  curve is shown in Figure 3.18(a), for each  $v_{eq}$ . The peak  $c_p$  is shown to be 24.0%, for an  $v_{eq}$  of 4.96m/s. The results show a consistent shape but do not show a good correspondence in the vertical scaling. It was hence assumed that the lowest wind (4.00 m/s) speed measurement was accurate and the values of higher wind speeds were adjusted to produce  $c_p$  curves at these speeds which more closely matched the peak of the 4.00m/s  $c_p$  curve. The resulting modified  $c_p$  curves are shown in Figure 3.18(b),



Figure 3.18: Estimated coefficient of performance,  $c_p$ , vs. tip-speed ratio curve, showing (a) original, and (b) modified data. Note that the legend shows wind tunnel (internal) wind speeds, and that the solid line represents a fifth-order polynomial approximation.

along with a fifth-order polynomial approximation (see Equation (3.5)), shown by the solid line. The value of tip-speed ratio corresponding to the peak  $c_p$  of the curve, i.e. 5.6, is consistent with the trend shown in [22] for 1, 2 and 3 bladed turbines. Note that  $c_p$  curve is set to zero, for  $\lambda < 0.2$  and  $\lambda > 8$ , as the  $c_p$  must be greater than or equal to 0.

$$c_p(\lambda) = 8.1 * 10^{-5} * \lambda^5 - 0.0018 * \lambda^4 + 0.011 * \lambda^3 - 0.018 * \lambda^2 + 0.026 * \lambda - 0.0059$$
(3.5)

The resulting peak  $c_p$  is 18.5%, which is significantly less than that expected for turbine of radius 0.9m, i.e. a  $c_p$  of about 31%. This value is linearly interpolated from data given in [1], which states that wind turbines of diameter around 1m generally have a maximum  $c_p$  of about 0.3 (30%), whilst a turbine of diameter about 10m has a peak  $c_p$  of about 0.42 (42%). The difference between the expected and estimated  $c_p$  values may be accounted for by the use of low-cost (and possibly poorly designed / inefficient) blades, and experimental error, particularly in the wind speed estimation. Despite the lower than expected peak  $c_p$  value, the polynomial approximated  $c_p$  curve is used to predict the turbine power and torque in the following sections. The resulting modified equivalent wind speeds are 4.0, 4.4, 4.8, 5.1 and 5.4m/s, and are shown in Figure 3.19, as a function of wind tunnel wind speed. The modified wind speeds are compared to the originally calculated data (from Equation (3.3)) for reference. The modified data is now linearly proportional to the wind tunnel (internal) wind speed, as would be expected for a linear system.



Figure 3.19: Estimated equivalent wind speed vs. internal wind tunnel wind speed. The solid line represents the wind speeds calculated from the cube-root expression, i.e. Equation (3.3), whilst the points correspond to the modified  $c_p$  curve of Figure 3.18(b).

### 3.3.3 Open-Loop Control Mode

The wind turbine is subjected to the small range of available wind speeds, i.e. 4.0 to 5.4m/s, whilst the generator is loaded by the SMR. The duty-cycle is varied for each wind speed, whilst the SMR output current and turbine speed is recorded. The estimated and analytical turbine (generator input) power vs. speed is shown in Figure 3.20. The estimated turbine power, derived from the measured SMR output power, duty-cycle and turbine speed. The results shows a good correspondence to the analytical values, which are based on the analytical turbine equations (Section 1.2.1), the polynomial approximated  $c_p$  curve, and the estimated equivalent wind speeds, as discussed above. The dashed line in the figure represents the PM generator's open-circuit iron, friction and windage loss (see Section 2.2.5), which increases roughly linearly with the square of generator speed. This loss is significant compared to the turbine output power, which is due to the low values of the available wind speeds, and prevents measurements being taken at light loads (below the dashed line).



Figure 3.20: Estimated (from measurements) and analytical (lines) turbine power vs. measured turbine speed, for the small range of equivalent wind speeds. The dashed line represents the PM machine open-circuit iron, friction and windage (OCIFW) loss.

The resulting estimated machine input (turbine output) torque is shown in Figure 3.21 as a function of measured turbine speed. The dashed line represents the PM machine iron,
friction and windage loss, which only increases slowly with generator speed, as previously seen in Figure 2.10(b). Note that the vertical offset of 0.3Nm represents the torque required to overcome the machine's bearing, friction losses and cogging torque.

Note that the close match between the estimated and analytical turbine power and torque indicate that the modified  $c_p$  curve and the resulting equivalent wind speeds are accurate.



Figure 3.21: Estimated (symbols) and analytical (lines) torque vs. measured turbine speed, for a small range of wind speeds. The dashed line represents the PM machine's open circuit iron, friction and windage loss (OC IFW).

#### SMR Turbine Control

Although the above figures indicate that turbine power and torque vary with both turbine and wind speed, it is the manipulation of the SMR duty-cycle that has allowed the turbine speed and power to vary. The SMR output power is hence examined as a function of duty-cycle. This is shown in Figure 3.22 for the various estimated wind speeds mentioned above. The dashed line represents an artificial *rated* wind speed, which corresponds to the peak SMR output power at a wind speed of 4.8m/s. Note that this wind speed is required to demonstrate the SMR's ability to limit power beyond rated wind speed. The figure clearly shows that the SMR can maximise and limit its output power at different wind speeds, by adjusting the SMR duty-cycle.



Figure 3.22: Measured SMR output power vs. duty cycle, for the various low wind speeds available. The dashed line represents an artificial *rated* wind speed, which corresponds to the peak SMR output power obtained at an equivalent wind speed of 4.80m/s.

In addition, the turbine speed is shown to be controlled by adjusting the duty-cycle for various wind speeds. This is seen in Figure 3.23, which shows the turbine speed is nearly linearly proportional to the duty-cycle, at high duty-cycles. However, this speed tends to level off as the duty-cycle approaches 0%. This occurs as the DC link voltage is higher than the rectified generator back-EMF voltage, i.e. current and hence electrical power will not flow from the turbine to the voltage-source load. This was previously seen in the dynamometer testing section (see Section 3.2.2) for the 200 and 400rpm cases, where it was found that the generator must operate at a minimum speed of about 500rpm to transfer power to the SMR for a duty-cycle of 0%. The maximum turbine speed is 438rpm, which indicates that the SMR output power will be zero for a duty-cycle of 0% for all test wind speeds. This is verified in Figure 3.22, under which condition, the turbine will operate at its no-load speed.



Figure 3.23: Measured turbine speed vs. duty-cycle, for the small range of low equivalent wind speeds.

#### 3.3.4 Comparison of Control Modes

It has been demonstrated that the turbine power, torque and speed, and SMR output power can be controlled by adjusting the SMR duty-cycle. The results, however, are obtained from open-loop testing and do not demonstrate the SMR's ability to limit SMR and turbine power beyond rated wind speed. This is due to the available low equivalent wind speeds wind, and hence, an artificial rated wind speed of 4.80 m/s is selected to demonstrate the desired control principle. The corresponding rated SMR output power is 19.4W, which was shown by the dashed line in Figure 3.22.

Two control modes (CM) are examined here. The first (CM 1) maximises the SMR output power for all wind speeds, whilst the second (CM 2) limits the SMR output power to its rated value beyond rated wind speed. The corresponding SMR duty-cycle vs. wind speed plot is shown in Figure 3.24, for both control modes. The dashed line in the figure represents rated wind speed. The data is mainly measured, however, some points for CM 2, are interpolated from Figures 3.20–3.23. Consider CM 1, the duty-cycle corresponding to maximum SMR power is shown to decrease nearly linearly with wind speed. This is expected as the turbine speed that corresponds to maximum  $c_p$  and hence turbine power increases linearly with wind speed, whilst the rectifier voltage corresponding to maximum

DC power increases with turbine speed (recall Figure 2.25). Therefore, maximum power is extracted from the SMR by linearly increasing the rectifier voltage (duty-cycle) with wind (and hence turbine) speed.



Figure 3.24: Measured duty-cycle vs. wind speed comparison of both control modes (CM). CM 1 aims to maximise power for all wind speeds, whilst CM 2 limits power beyond rated.

In contrast, the duty-cycle of CM 2 sharply increases beyond rated wind speed. This is expected as the turbine must reduce its speed to maintain rated power. This is achieved by increasing the duty-cycle beyond the maximum power point, which increases the generator torque. This concept was previously shown in Figure 3.23, and is also clearly seen in Figure 3.25, which shows the turbine and SMR performance characteristics, i.e the turbine power, SMR output power, turbine torque and speed as a function of wind speed, for both control modes. The vertical dashed lines represent rated wind speed (4.8m/s), whilst the horizontal dashed lines of Figures 3.25(a) and 3.25(b) represent rated turbine and SMR output powers, respectively. Note that the turbine torque and power graphs of Figure 3.25 are calculated, whilst the SMR output power and turbine speed graphs are measured. The solid and dashed lines simply connect the data.

#### Control Mode 1 - Power Maximisation

Consider CM 1 in Figure 3.25. Recall that the turbine speed corresponding to maximum power ideally increases linearly with wind speed, whilst the turbine torque and maximum power increases with the square and cube of wind speed, respectively. This is verified by the CM 1 plots in Figure 3.25, with the exception of the highest wind speed. The graphs indicate the calculated equivalent wind speed of 5.4m/s is larger than expected, which is likely caused by the non-ideal test facility and experimental errors. Despite this, the

SMR output power is shown to continually increase with wind speed. Its shape resembles that of the turbine power vs. wind speed plot, indicating a relatively constant generator and SMR power loss.



Figure 3.25: Measured turbine and SMR operating characteristics for both control modes (CM), showing (a) turbine power, (b) SMR output power, (c) turbine torque, and (d) turbine speed vs. wind speed. CM 1 maximises the SMR power for all wind speeds, whilst CM 2 limits it (to rated power) above rated wind speed (4.80 m/s). The vertical dashed line represents rated wind speed, whilst the horizontal dashed lines represents rated power. Note that the solid and dashed lines simply connect the data.

#### Control Mode 2 - Constant Power

Consider the second control mode (CM 2) in Figure 3.25. The turbine speed, torque and power is identical to that of CM 1 for wind speeds below rated, however, these differ for wind speeds above rated. The turbine torque torque is increased to reduce the speed and

hence limit the turbine and SMR output powers. The turbine and SMR powers are shown to be maintained at rated values for wind speeds above 4.80 m/s (rated). Although this control mode successfully demonstrates both the power maximisation and power limiting modes, as required, the SMR was controlled in an open-loop fashion. A closed-loop version of this control algorithm must be implemented, such that the SMR maximises power, below rated wind speed, and limits power (and turbine speed) above rated wind speed, autonomously.

#### **SMR** and Generator Efficiency

The estimated SMR and generator (total) efficiency is plotted as a function of SMR output power in Figure 3.26, for the available low equivalent wind speeds. The peak efficiency increases from about 50% at a wind speed of 4.0 m/s to 70% at a wind speed of 5.4 m/s. This occurs as the generator and SMR power losses are most significant at low powers (wind speeds). These losses appear fixed (deduced from Figures 3.25(a) and 3.25(b)), and should become less significant as the SMR output power (wind speeds) increase. Each wind speed has two efficiency vs. SMR power plots, which meet at the peak SMR power, this occurs as the same DC power can be obtained for two different



Figure 3.26: Estimated SMR and generator efficiency vs. SMR output power, for equivalent wind speeds of 4.0-5.4 m/s.

rectifier currents/voltages. The higher of the two efficiency plots corresponds to the higher rectifier voltage (lower current) case, as previously seen with the DC generator and total (generator + SMR) efficiency vs. power curves, i.e. Figures 2.28 and 3.15, respectively.

### 3.4 Chapter Summary

The use of a SMR allows the rectifier output voltage to be controlled as it is proportional to the DC link output voltage and duty-cycle. This allows power to be transferred from the generator to the load at low speeds, which is not possible using an uncontrolled rectifier. Output power can be generated for a given speed, providing the rectifier voltage is set below the generator's rectified back-EMF voltage. This was experimentally verified using the dynamometer which saw that power was extracted for only a limited range of duty-cycles at low generator speeds. In contrast, power was obtained at any duty-cycle at high speed, as the rectified back-EMF voltage exceeds the DC link voltage.

The SMR output current is linearly proportional to the input current and SMR dutycycle, allowing simple output current control when using a constant-current input source. The generator, however, only behaved as a near ideal constant-current source at the higher tested generator speeds where the back-EMF voltage significantly exceeded the DC link (SMR output) voltage. The machine's ability to provide a constant current output decreased with generator speed and back-EMF voltage. At a given generator speed, the duty-cycle can be controlled to maximise the SMR output current and hence output power. The duty-cycle corresponding to the peak output power is linearly proportional to the generator speed. This relationship also applies to the generator torque, however, the range of duty-cycles for which this (and the power can be maximised) applies, varies with the DC link voltage.

For a constant generator speed, the SMR DC output power vs. duty-cycle curve is shown to be the mirror image of the uncontrolled rectifier DC output P-V locus, as the rectifier output voltage is proportional to the complement of the duty-cycle (1-d). Similarly, the torque vs. duty-cycle curve, for the lowest generator speed, appears as the mirror image of the short-circuit torque characteristic. The speed range for which the SMR output power can be maximised corresponds to the boost rectifier region, whilst the SMR operates as a current chopper for higher generator speeds. The range of speeds corresponding to each mode of operation varies with DC link voltage.

The SMR's ability to control the test wind turbine was demonstrated using a small wind tunnel. Despite the severe limitations of the available test facility used, it was demonstrated that the turbine power, torque and speed could be controlled by adjusting the duty-cycle. The turbine control requirements, such as power maximisation and speed reduction etc., were successfully demonstrated using an artificially reduced value of rated wind speed under open-loop conditions. The wind turbine blade characteristics were estimated based on the wind tunnel testing. The  $c_p$  curves, calculated for each wind speed, did not agree in magnitude, however, matched in shape. The estimated wind speeds were then adjusted such that each  $c_p$  curve agreed in both shape and magnitude. This resulted in a low range of equivalent wind speeds, ranging from 4.0 to 5.4m/s, and hence a *rated* wind speed of 4.8m/s was used.

The turbine speed was found to vary in proportional to the duty-cycle, and to be independent of wind speed at high duty-cycles. In contrast, the turbine speed reached a limiting value at low duty-cycles, as the resulting rectifier voltage exceeded the generator's rectified back-EMF voltage, i.e. power did not flow from the generator to the load. The generator was hence open-circuited and rotated at a speed governed by the wind speed and the generator no-load torque (open-circuit power loss).

## Part II

# Investigation of Grid-Connected Inverter based on Switched-Mode Rectifier Topology

## Chapter 4

## Simulation and Test of 150W Grid-Connected Inverter

This chapter introduces the principles and features of a grid-connected inverter. This is followed by the design and discussion of a novel 150W current source grid-connected inverter topology that is based on a high-inductance PM generator and rectifier acting as a constant current source. The inverter concept is proved by computer-based simulations and experimental testing that considers both resistive and grid-connected loads. The constant current assumption and effect of grid parameter variations, e.g. grid voltage and inverter modulation index are investigated.

### 4.1 Introduction

The primary function of a grid-connected inverter (GCI) is to convert power from a renewable energy source, to an AC output that is directly fed to the utility grid. A general block diagram of a small scale grid-connected wind turbine is shown in Figure 4.1. The figure also shows the inverter output current,  $I_{inv}$ , and the grid voltage,  $V_g$ , where the grid voltage is in-phase with the inverter current. This operating condition is known as *unity power factor*, which maximises the active power fed to, and minimises the reactive power drawn from the grid. The grid is treated as an *infinite bus*, i.e. a fixed frequency constant voltage source, which is unaffected by any load or fault condition.



Figure 4.1: Simple block diagram of a small scale grid-connected wind turbine system (left), and the inverter output current,  $I_{inv}$ , and grid voltage,  $V_g$ , for unity power factor operation (right).

#### 4.1.1 Inverter Requirements

Utility operators impose strict inverter power quality, power factor, direct current (DC) injection, and fault detection requirements. The grid-connected inverter is tested against the Australian Standards, using a power analyser [96]. The inverter must be shown to meet the standards before being made commercially available.

#### Power Quality

Nonlinear loads, such as adjustable-speed drives, arc furnaces, arc welders and fluorescent lights, draw non-linear currents which contain harmonics. These harmonic currents produce corresponding voltage harmonics and cause the supply voltage waveform to be distorted [97]. Excessive supply voltage distortion can affect the operation of sensitive loads, such as computers. Due to this, there are grid standards in place, e.g. the IEEE standard 519-1992 for utility operators and end users, and the Australian Standard (AS) 4777.1–3 for grid-connected inverters. Note that the THD requirements of AS 4777.2 are identical to those of IEEE standard 929-2000 [96].

The Australian Standard 4777.1–3 covers grid-connection of energy systems via inverters. According to AS 4777.2 the grid-connected inverter must meet mandatory total harmonic distortion (THD) requirements. The current must contain less than 5% THD, up to and including the  $50^{th}$  harmonic, as listed in Table 4.1. Although the grid voltage must also meet strict harmonic limits, the majority of grid-connected inverters control only the output current and hence are not required to monitor the output voltage harmonic distortion.

Table 4.1: Current harmonic limits of Australian Standard 4777.2 [96]

NOTE: This table is included on page 119 of the print copy of the thesis held in the University of Adelaide Library.

#### **Power Factor**

The power factor angle,  $\phi$ , refers to the phase angle of the current waveform, with respect to the voltage waveform. The power factor is defined as the cosine of the power factor angle, and is leading if  $\phi > 0^{\circ}$  (current waveform leads the voltage waveform), and lagging if  $\phi < 0^{\circ}$  (current waveform lags the voltage waveform). The AS 4777.2 specifies that the inverter must deliver a high quality current waveform at a power factor between 0.8 leading and 0.95 lagging, for inverter outputs between 20 to 100% of rated volt-amperes [96]. These power factors are illustrated in Figure 4.2, which shows a lagging power factor of 0.95 in Figure 4.2(a) and a leading power factor of 0.8 in Figure 4.2(b). Note that the power factor requirement considers the inverter as a load from the perspective of the grid [96].



Figure 4.2: Example of (a) 0.95 lagging and (b) 0.80 leading power factors, where  $\phi = -18.2^{\circ}$  and  $\phi = +36.9^{\circ}$ , respectively.

#### **Direct Current Injection**

Injection of DC currents into the AC utility grid is not desirable as this can cause saturation and hence overheating of utility distribution transformers. The Australian Standard 4777.2 requires that a grid-connected inverter does not inject more than 0.5% of rated current or 5mA of DC current (whichever is greater) in to the grid [96]. The use of a line frequency a transformer between the grid and inverter avoids this problem, however, is undesirable due to its cost, size and weight issues, as previously seen in Section 1.5.4.

#### **Fault Detection**

Aside from inverter controlled parameters such as output power quality, power factor and DC injection, the grid-connected inverter must also be able to detect a range of grid faults. These faults include under / over voltage, under / over frequency and islanding. The inverter is required to disconnect itself from the utility grid within two seconds of such a fault [98]. Fault detection is beyond the scope of this research.

#### 4.1.2 Desirable Features

Grid-connected inverters are designed to deliver a low distortion current to the grid at a high power factor, and in addition, be highly efficient and low cost [33, 99, 100]. The former were discussed in Section 4.1.1, whilst the latter are discussed below.

#### Cost

The cost of a GCI makes up a significant portion of the total cost of a small scale gridconnected wind turbine, and can discourage people from utilising wind power. Presently, a 1–7kW grid-connected inverter costs about AUS\$1000 more than a similar sized standalone inverter, as previously shown in Figure 1.19. The excessive costs are associated with additional components and the overall complexity of the system. This research aims to design and develop a low-cost grid-connected inverter to encourage wind power utilisation.

It has been estimated that the small-scale wind energy market is growing at roughly 40% per year [18], and that annual sales, in the US alone, will be between US \$27M and \$57M by 2010 [19]; this value depends on factors such as government rebates and tax incentives etc. Within this market there is an increasing demand for low-cost grid-connected wind turbines, and it is expected that by 2020 small-scale wind turbines will provide 3% of the US electrical energy consumption.

#### Performance

The grid-connected inverter is the interface between the wind turbine and the grid, it hence has the responsibility of controlling the turbine torque and hence power. The gridconnected inverter should be designed such that power is extracted over a wide range of wind speeds, and that it is reliable and efficient.

- 1. General operation The inverter should maximise turbine power below rated wind speed and maintain rated power above rated wind speed, in accordance with standard turbine practise. Improving the ability to extract power from the wind at low speeds is especially important as small-scale wind turbines suffer poor capacity factors, as they are typically located close to the local power demand. These turbines thus generally see lower average wind speeds compared to large (MW) wind turbines, whose siting is carefully chosen to optimise output power.
- 2. Reliability The inverter should operate autonomously for long periods of time, e.g. years, without regular maintenance. Careful selection of inverter types and hence components may reduce the need for regular maintenance, e.g. a voltage source inverter requires the use of electrolytic (DC link) capacitors to regulate the input voltage. These are often large and bulky, as they have a limited life expectancy; this is mainly attributed to the electrolyte [101]. Current source inverters are generally more reliable than voltage source types as they do not use DC link capacitors.
- 3. Efficiency Internal power losses are generally larger in current source inverters compared to voltage source types, as these inverters require a DC link inductor, which attracts copper losses. It is generally recognised that current source inverters are less efficient than their voltage source counterparts.

While both the current and voltage source inverters should be able to appropriately control a wind turbine, a reliability / efficiency trade-off exists. Despite this, the majority of purchased grid-connected inverters are the voltage source type; 81% of these are current controlled, whilst 19% control the output voltage [62]. Current control is more popular as high power factors can be obtained with simple control circuits, and transient current suppression is possible when disturbances, e.g. voltage changes, occur in the utility power system [62].

## 4.2 Proposed Concept

#### 4.2.1 Inverter Overview

The proposed grid-connected inverter is based on a current source input. This is decided as i) a low-cost constant current source can be easily obtained using a high-inductance generator with a rectifier, and ii) the inverter output current can be linearly controlled using a single-switch converter, and without the need for a high-cost output current sensor or a fast feedback controller.

An equivalent block diagram of the proposed inverter is shown in Figure 4.3, along with the expected input and output currents of the four inverter stages, i.e. the *constant* current source, current-wave shaper, unfolding circuit, and low-pass filter. The inverter operates from an essentially constant input current, delivered by the PM generator and rectifier. The current wave-shaper and unfolding circuit stages are used to produce a unipolar PWM current, which is filtered and fed to the grid, via the low-pass filter. Each inverter stage is further discussed in the following sections.



Figure 4.3: Overview of the proposed grid-connected inverter, showing the: rectifier, current wave-shaper, unfolding circuit and output filter. The output current of each stage is also shown. Note the input (machine) current is three-phase AC, of varying magnitude and frequency, which is speed dependent.

The inverter exploits the use of a high-inductance generator and rectifier to form the constant current source. The common DC link inductor (energy storage element) is therefore avoided, which will further reduce the cost and improve the overall efficiency.

#### 4.2.2 Constant Current Source

The combination of the high-inductance generator and an uncontrolled rectifier, ideally provide the inverter with constant current. This arrangement and the idealised machine line and rectifier output currents are shown in Figure 4.4. This type of constant current source (CCS) is not reported amongst the literature, and forms the basis of the novel grid-connected inverter topology.



Figure 4.4: Constant current source, showing (left) the circuit, and (right) the ideal machine and rectifier output currents.

The rectifier output current,  $I_R$ , is expressed by Equation (4.1), where  $I_L$  is the RMS line current value. For example, a current source based on the same high-inductance PM generator that was used in part I, would deliver a maximum line current of about  $16A_{RMS}$ , which corresponds to a DC (rectifier) output current of approximately 22A.

$$I_R = 1.35 I_L$$
 (4.1)

The lack of DC link inductor implies that the instantaneous inverter input and output powers must be equal. This is easily achieved using a 3ph inverter, as the summation of the instantaneous output powers is constant. However, this project focusses on a single-phase inverter whose instantaneous output power fluctuates at twice the grid frequency (the average power is constant). The inverter hence draws non-constant (pulsating) power from the generator, which causes the generator to vibrate [91] and the inverter input current and voltage to fluctuate at twice the grid frequency.

The fluctuating inverter input current contains harmonic distortion, which is ultimately seen in the output current. Despite this, the high-inductance generator and rectifier act as a constant current source when the rectifier output voltage is much less than its open circuit voltage; a reasonable choice is 25%. This is seen experimentally in Section 4.4.2 and is analysed in Section 5.1. Feed-forward current compensation is examined in Section 5.4 to overcome this limitation. Note that the remaining inverter stages assume an ideal input current.

#### 4.2.3 Current Wave-Shaper

The current wave-shaper (CWS) is essentially a high-speed switch and a reverse blocking diode, that shorts the input current to create a PWM chopped output current. The circuit, along with the ideal input and output currents, are shown in Figure 4.5. The output current,  $I_{out}$ , is the inverse of the PWM duty-cycle (control signal), d, as shown by Equation (4.2), where  $I_{in}$  is the input current.



Figure 4.5: Current wave-shaper, and the input and output currents.

$$I_{out} = I_{in}(1-d) \tag{4.2}$$

The combination of the constant current source and the current wave-shaper is the same as the SMR circuit and high inductance PM generator, previously examined as a DC-DC converter in part I. Both circuits make use of a reverse blocking diode, which prevents shorting the DC link capacitor, for the DC-DC converter case, and the grid for the inverter case. The key physical difference between the circuits is the removal of the DC link capacitor for the inverter application, which is summarised in Figure 4.6.

The modified SMR circuit is also controlled differently i.e. a time-varying duty-cycle is now applied to the high-frequency switch, as opposed to a constant duty-cycle for the DC-DC converter case. A comparison of the constant and time-varying duty-cycles is shown



Figure 4.6: Comparison of (a) the switched-mode rectifier (SMR), and (b) modified SMR circuit, for use as a DC-DC converter and for the inverter, respectively. The SMR circuits are enclosed by the dashed boxes, whilst the dotted boxes of (b) show the constant current source (CCS) and the current wave-shaper (CWS) circuits. The key difference in (b) is the removal of the DC link capacitor.

in Figure 4.7. The combination of the new control signal and the removal of the DC link capacitor allows the proposed inverter to produce an AC output current that resembles that of a rectified unipolar PWM wave. This is converted into a unipolar PWM current by the unfolding circuit, and is then fed to the grid via the low-pass filter.



Figure 4.7: Comparison of the SMR control PWM duty-cycles, showing the (top) constant, and (bottom) time-varying signals, as used for the DC-DC converter and the inverter applications, respectively. The zero position for each waveform is shown by the arrow on the left; the vertical and horizontal scales are 2V and 2ms per division, respectively.

#### 4.2.4 Unfolding Circuit

The unfolding circuit is essentially an H-bridge inverter. Its role is to unfold the input current at twice the grid frequency,  $\omega_g$ , as summarised in Equation (4.3). The combination of this unfolding action and the CWS PWM switching, yield a unipolar PWM output current, whose fundamental frequency matches that of the grid. The unfolding circuit is built using low-cost thyristors and as shown in Figure 4.8, along with a demonstration of its unfolding action. The polarity is alternated by the correct triggering of opposing thyristors, e.g. THY 1 and THY 3 are triggered simultaneously to obtain a positive output current, whilst triggering THY 2 and THY 4 in unison, will cause the current to flow in the opposite direction.

$$I_{out} = I_{in} \cdot \operatorname{sgn}\left[\sin(\omega_g t)\right] \tag{4.3}$$



Figure 4.8: Unfolding circuit converter, and the input and output currents.

Thyristors are primarily selected, based on their low-cost. However, they also eliminate the need for series connected diodes, which are required for a current source H-bridge inverter, to block bi-directional voltages from the load and to allow uni-directional current flow [81, 102]. In addition, thyristors have a fixed voltage drop, and can hence reduce power losses in high voltage applications [103]. The drawback, however, is its inability to commutate (turn-off) when fed by a constant input current. A grid-connected current-fed thyristor H-bridge inverter will only commutate correctly if the load has a leading power factor [104]. To overcome these difficulties, the majority of grid-connected inverters and all PWM inverters force switch commutation. However, this attract additional circuitry, and increases the complexity and hence cost. The advantage of the proposed grid-connected inverter design is that the thyristors commutate naturally due to the nature of the waveshaper output current, i.e. the current sufficiently falls below the thyristor holding current. The drawback of this design is that the thyristors must now be triggered at the PWM switching frequency, which increases the inverter switching losses.

#### 4.2.5 Low-Pass Filter

The low-pass filter is the final stage of the proposed gird-connected inverter. It is used to remove the high frequency PWM current components, whilst allowing the fundamental component to pass through to the grid. Figure 4.9 shows the low-pass filter required for a current source grid-connected inverter, and its expected input and output currents. Note that this capacitive-inductive (CL) filter differs from conventional second-order low-pass, i.e. inductive-capacitive (LC) filters, as the unfolding circuit output is essentially a current source, whilst the grid is effectively a voltage source. The *filter impedance mismatch criteria* states that the filter input impedance must hence be low and that its output impedance must be high, to correctly couple a high impedance current source and a low impedance voltage source (grid) [105].



Figure 4.9: Inverter low-pass CL filter circuit, and the input and output currents.

## 4.3 Test Set-up, Implementation and Simulation

This section describes the inverter test arrangements, the PM generator current source, the power electronics and control hardware implementation, and the simulations used to validate the inverter concept and simulate inverter operation.

#### 4.3.1 Dynamometer Test Arrangement

The dynamometer test-rig was used to drive the PM generator, which consists of a 5kW DC machine that is directly coupled to the PM generator. The PM generator acts as the inverter current source, and was rotated at constant speed to simulate inverter operation under steady-state wind conditions. Turbine speeds of 500 and 1,000rpm were selected as they represent turbine speeds at average and high wind speeds, respectively.

#### **Proof of Concept**

The inverter was initially designed, constructed and tested using a resistive / capacitive load. The circuit was tested first using an ideal current source, i.e. a DC voltage source in series with an inductor, and second using the high-inductance PM generator and rectifier acting as the current source. These test arrangements are summarised in Figure 4.10.



(a) Ideal current source set-up

(b) Dynamometer set-up

Figure 4.10: Preliminary resistive / capacitive load test arrangements, showing (a) the ideal, and (b) the PM generator current source test arrangements.

#### Grid-Connected Testing

Due to the use of a low-voltage generator, the inverter was grid-connected to an artificial (low-voltage) grid. This was achieved experimentally by using a 10A autotransformer, whose secondary voltage was varied between 5 and  $20V_{RMS}$ . In addition, an isolation transformer was connected between the mains and the autotransformer for safety, as

shown in Figure 4.11. This figure shows the inverter arrangement as tested in the laboratory, which includes a resistive load,  $R_L$  and two switches, SW1 and SW2, which are used to make the transition from resistive to grid-connected loading (the synchronisation procedure is explained at the end of this section). The transformers conveniently provide leakage reactance and resistance to assist with filtering. However, these parameters vary with turns ratio and are hence enclosed within the dashed box of Figure 4.11.



Figure 4.11: Grid-connected inverter test arrangement, showing the use of an isolation and autotransformer. The current source is not shown, however, it is achieved using the PM generator and rectifier, using the dynamometer test-rig.

The variable transformer parameters are approximated by an equivalent inductance,  $L_{eq}$ , and resistance,  $R_{eq}$ , which are expressed by Equations (4.4) and (4.5), respectively. Note that  $R_1 n^2$  and  $L_1 n^2$  represent the variable resistance and inductance of the autotransformer, where *n* represents the turns ratio. Similarly,  $R_2$  and  $L_2$  represent the fixed resistance and inductance of the isolation transformer. The equivalent (combined) transformer inductance and resistance is shown in Figure 4.12.

$$L_{eq} = L_1 n^2 + L_2 \tag{4.4}$$

$$R_{eq} = R_1 n^2 + R_2 \tag{4.5}$$

The equivalent transformer inductance and resistance was experimentally measured, for a number of autotransformer turns ratios. These were obtained using a second autotransformer to perform a short-circuit test on the first one. The results are expressed as a function of autotransformer output voltage, and are shown in Figure 4.13. The points represent the calculated data, whilst the lines correspond to fitted curves. Both fitted curves are quadratic, which is expected as both  $L_{eq}$  and  $R_{eq}$  have a variable term that is proportional to  $n^2$ .



Figure 4.12: Equivalent circuit for test arrangement, showing the isolation and autotransformer leakage inductance and resistance inside the dashed box.



Figure 4.13: Measured transformer equivalent (a) resistance, and (b) inductance vs. autotransformer output voltage. Each subfigure includes a fitted curve, and its quadratic equation.

#### Synchronisation Procedure

The inverter was synchronised to the grid using the procedure listed below:

- Close switch 1, open switch 2 before the inverter is powered on.
- Power the inverter and rotate the PM machine at the desired speed.
- Synchronise the inverter to the grid by adjusting the load resistance or the inverter modulation index until the voltage drop across  $R_L$  and the artificial grid is synchronised (both in phase and magnitude).

- Close switch 2 the resistor is effectively connected to the grid and the inverter.
- Open switch 1 such that the inverter is now grid-connected

Note that care must be taken when loading the current source inverter, as one switch must remain closed at all times to provide a current path. In the event that both switches are opened, the inverter output voltage will significantly increase which may damage the inverter components. This voltage increase is caused by both the high generator opencircuit voltage and the voltage boosting action of the un-loaded modified SMR.

#### 4.3.2 Constant Current Source - PM Generator

The test PM machine selected is the same as that investigated in Chapter 2. This machine is again used because of its high inductance, which is due to the use of a concentrated stator winding, and its torque rating, which is consistent with a several hundred watt wind generator. The test machine is shown again in Figure 4.14, for convenience.



Figure 4.14: Photograph of an outer-rotor Fisher & Paykel PM generator.

The generator was tested over a range of speeds using the dynamometer test-rig. A summary of the PM machines's generating capability is shown in Figure 4.15, which shows the measured and simulated DC output current versus DC output voltage characteristics (DC I-V locus). This was measured at various speeds when operating into a resistive load

through a three-phase rectifier. The simulated results are obtained using PSIM<sup>®</sup> and are based on the machine parameters shown in Table 4.2. The calculated results show a good correspondence with the experimental results and give confidence in the simulation model. It is seen that the generator is a reasonable approximation to a constant-current source at high generator speeds and low DC output voltages.



Figure 4.15: Current-Voltage locus of PM generator. The solid lines represent PSIM<sup>®</sup> simulations, whilst the points represent measured experimental data.

Parameter	Value	
Stator Connection	Delta	
Pole Pairs	24	
Maximum Speed Tested	$1000 \mathrm{rpm}$	
SC Line Current (max. speed)	$16 A_{RMS}$	
Back-EMF Constant, $k$	0.0668  V/rpm	
Phase Inductance, $L_{ph}$	$2.87 \mathrm{~mH}$	
Phase Resistance, $R_{ph}$	$0.519 \ \Omega$	

Table 4.2: Measured PM generator properties.

#### 4.3.3 Power Electronics and Control Implementation

The proposed grid-connected inverter was constructed using a line frequency, three-phase uncontrolled rectifier, a high-current MOSFET and high-current thyristors. Note that the generator stator currents are continuous at high speeds, and hence the rectifier does not need a high-frequency switching capability. In addition, the thyristors and MOSFET are rated at 40A, despite an expected maximum rectifier output current of 22A, i.e. this provides some safety margin. Sections of the grid-connected inverter hardware, including the power electronic components and microcontroller are shown in Figure 4.16.



Figure 4.16: Photograph of power electronic components of the grid-connected inverter, showing (from top left and moving clockwise): ① the current wave-shaper, ② the full-bridge 3ph rectifier, ③ the microcontroller, ④ the unfolding circuit, and ⑤ the necessary MOSFET and thyristor gate drivers.

The inverter is controlled by a 16 bit Mitsubishi MSA0654A microcontroller in an open-loop manner. The zero-crossings of the grid voltage are used to generate the current wave-shaper PWM signal and thyristor trigger pulses. The PWM switching frequency is set to 4 kHz, as this is a reasonable trade-off between inverter switching losses and output quality, both of which are proportional to switching frequency. The PWM signal is based on a duty-cycle look-up-table whose pointer is reset at positive grid zero-crossings, which ensures that the unfolding circuit output current is in-phase with the grid voltage. The use

of a look-up-table is recommended in [106], as this improves power quality characteristics of the distribution network. The same reference explains that some inverters sample the grid voltage for their current reference. The resulting output current, of such inverters, hence contains the same distortion found in the grid voltage.

The look-up-table stores a normalised 180° conduction sinusoid, which corresponds to a modulation index,  $m_a$ , of 100%. The inverter output current magnitude and power is controlled by reducing the modulation index, which effectively increases the current waveshaper PWM duty-cycle. The adjusted duty-cycle,  $d_a$ , is expressed by Equation (4.6), where  $d_i$  represents the stored duty-cycles.

$$d_a = 100 - m_a + \frac{(m_a \times d_i)}{100} \tag{4.6}$$

The microcontroller is programmed using the C language, and makes use of both software and hardware interrupts. A simplified overview of the microcontroller hardware and its software flow-chart is shown in Figure 4.17, and a copy of the code can be found in appendix C.2. Note that the thyristors trigger signals are synchronised to the falling edge of the CWS MOSFET control signal, and that the MOSFET is effectively shorted for two pulse widths before the zero-crossing to ensure thyristor commutation.



Figure 4.17: Overview of the Microcontroller hardware, and software flow chart. The large arrows represent control signals that are used to drive the semiconductor switches, whilst the shaded boxes represent microcontroller interrupts.

#### 4.3.4 Inverter Simulation

The power electronic simulation tool PSIM<sup>®</sup>, was used to simulate the grid-connected and resistive load testing of the proposed inverter. The software is again used as it has previously modelled the PM generator loaded by a rectifier and resistive load, and operating with a switched-mode rectifier, with high accuracy. The PSIM<sup>®</sup> model of the grid-connected inverter is shown in Figure 4.18. Various components of the hardware are shown in the figure, including the PM generator, full-bridge rectifier, current wave-shaper, unfolding circuit, low-pass filter and the artificial (low-voltage) grid. Control components, such as the PWM duty-cycle look-up-table generator, modulation index controller and the zero-crossing detector, are also shown. Note that the reverse-blocking diode of the CWS is a carry-over from the SMR design and it is not strictly required due to the reverse-blocking capability of the thyristors. Despite this, the diode is incorporated into the simulation model and the inverter hardware, as it is used for testing purposes. In practise, the diode should be able to be removed without affecting the inverter's performance.



Figure 4.18: PSIM<sup>®</sup> grid-connected inverter model, showing the hardware and control components. Hardware components include the delta-connected PM generator, the full-bridge 3ph rectifier, current wave-shaper, unfolding circuit, low-pass filter and the artificial grid; control components include the PWM duty-cycle look-up-table generator, modulation index controller, and the zero-crossing detector.

The proposed grid-connected inverter is accurately modelled in PSIM<sup>®</sup>. The machine is modelled by three delta-connected voltage sources, with each phase containing series inductance and resistance equal to the values specified in Table 4.2. The back-EMF and machine frequency is set for each turbine / generator speed. The semiconductor devices are also accurately modelled, i.e. the thyristor, reverse blocking diode and full-wave 3ph bridge rectifier voltage drops are included, as is the *on-resistance* of the MOSFET, and the latching and holding currents of the thyristors. This information is summarised in Table 4.3, and together with the machine modelling should result in simulations that closely match the experimentally measured data.

Component Property	Bridge Rectifier	MOSFET	Diode	Thyristors
Manufacturer	SEMIKRON	IR	IR	RS
Part Number	SKD230/16	IRFP150N	70HFLR	262-602
Rating	230A, 1600V	42A, 100V	70A, 1000V	40A, 1200V
Forward Voltage Drop	1V per diode	-	$1.85\mathrm{V}$	1.95V
On Resistance	-	$0.036\Omega$	-	-
Latching Current	-	-	_	0.6A
Holding Current	-	-	-	0.25A

Table 4.3: Semiconductor properties.

Note that IR denotes International Rectifier.

#### **Proof of Concept**

The inverter concept is initially simulated using the PM generator as a current source, and a resistive/capacitive load. The generator is simulated at 500rpm, as this is the expected turbine speed under average wind speed conditions, for an arbitrarily selected load combination of  $0.93\Omega$  and  $2,000\mu$ F. Various inverter current waveforms are shown in Figure 4.19, which prove the simulated inverter concept. The waveforms include the inverter input, and the current wave-shaper, unfolding circuit and filter output currents.

#### Modulation Index Variation

The effect of the inverter modulation index,  $m_a$ , variation is shown in Figure 4.20, for  $m_a$  equal to 100, 50 and 25%. It is seen that the output current magnitude is directly proportional to  $m_a$ , and hence the output power is proportional to  $m_a^2$ . This applies only to the resistive load case, as both the output current and voltage are dependent on  $m_a$ , whereas the power is only proportional to  $m_a$ , for the grid-connected case, as the inverter output (grid) voltage is fixed.



Figure 4.19: Preliminary PSIM<sup>®</sup> simulation proving the grid-connected inverter concept. The waveforms shown are the rectifier, current wave-shaper, unfolding circuit and filter output currents, from top to bottom, respectively. The zero position for each waveform is shown by the small arrow on the left. The vertical and horizontal scales are 20A and 2ms per division, respectively.



Figure 4.20: Simulated effect of modulation index,  $m_a$ , variation on inverter operation with resistive load. The resistive load currents for  $m_a$  equal to 100, 50 and 25% are shown. The vertical and horizontal scales are 5A and 2ms per division, respectively.

## 4.4 Experimental Testing

#### 4.4.1 Proof of Concept - Resistive Loading

Preliminary tests on the inverter were carried out with two power sources. Firstly, a DC power supply and a series-connected inductor was used to simulate an ideal constant current source. Figure 4.21 compares the computer generated (PSIM<sup>®</sup>) simulation and measured results corresponding to an arbitrarily selected resistive / capacitive load. The load resistance is  $0.5\Omega$ , and a filter capacitance of  $1000\mu$ F is selected to obtain a low-pass filter with cut-off frequency equal to 400Hz, i.e. 10% of the PWM switching frequency. The displayed waveforms include the rectifier, current wave-shaper, unfolding circuit and filter output currents, from top to bottom, respectively.

Secondly, the PM generator and rectifier was used as the inverter current source. Figure 4.22 compares the simulated and measured results, for the same resistive / capacitive inverter load. The figure also shows the rectifier, current wave-shaper, unfolding and filter output currents. The measured results show a good correspondence with the predicted waveforms and give confidence in the simulation model.

The current waveforms shown in Figures 4.21 and 4.22 illustrate the key principles of the inverter, i.e. the power source acts as a current source, and the wave-shaper acts as a PWM current chopper which produces a waveform whose fundamental component is a full-wave rectified sinewave. The action of the H-bridge inverter is illustrated, as is the low-pass filter's ability to sufficiently remove the high-frequency PWM components. The inverter output current is sinusoidal and fluctuates at the grid (mains) frequency.

Inspection of the PM generator's rectified output current shows that a high frequency ripple component, due to the bridge rectifier, is present. A low frequency 100Hz ripple component is also shown. This is caused by the inverter output voltage variation, which causes the inverter input current to vary (see the generator I-V locus in Figure 4.15). This 100 Hz input current variation ultimately affects the fundamental magnitude of the inverter output current, and hence should be limited to reduce output current distortion. The rectifier current ripple is less problematic, as it occurs at frequencies higher than the filter cut-off, and is hence not seen in the inverter output current. Despite the input current distortion from the rectifier ripple and power fluctuations, the simulated and measured results indicate that the inverter output current and its THD are not greatly affected by this, for this load condition. This is also summarised in Table 4.4, which shows a detailed comparison of the two current source cases.



Figure 4.21: Proof of inverter concept, showing (a) PSIM<sup>®</sup> simulation, and (b) measured results, using an ideal current source. The waveforms shown are the power supply, current wave-shaper, unfolding circuit and filter output currents, from top to bottom, respectively. The zero position for each waveform is shown by the small arrow on the left. The vertical and horizontal scales are 2A and 5ms per division, respectively.



Figure 4.22: Proof of inverter concept, showing (a) PSIM<sup>®</sup> simulation, and (b) measured results, using the PM generator as current source. The waveforms shown are the rectifier, current wave-shaper, unfolding circuit and filter output currents, from top to bottom, respectively. The vertical and horizontal scales are 20A and 5ms per division, respectively.
	DC Powe	er Supply	PM Generator (500rpm)		
	Simulated	Measured	Simulated	Measured	
Input current (A)	2.05	2.03	19.9	21	
Input current ripple $(A_{pk-pk})$	0.06	0.10	8.15	8.80	
Output current (A)	1.43	1.26	13.0	12.8	
Output voltage (V)	0.71	0.63	6.50	6.40	
Output power (W)	1.02	0.79	84.5	81.9	
Output current THD $(\%)$	3.40	3.70	5.97	4.00	

Table 4.4: Simulated and measured resistive load inverter performance.

The inverter concept has been successfully demonstrated, using both an ideal and the PM generator current sources. It is also tested with various resistive / capacitive loads to investigate the constant current assumption previously made, as discussed below.

## 4.4.2 Constant Current Assumption

For high quality inverter output current waveforms it is necessary that the input current be relatively constant even at the peak output voltage. This condition is satisfied when the peak output voltage is much less than the rectified open-circuit voltage. This is investigated for three machine speed / load cases, as highlighted on the I-V locus of Figure 4.23. The experimental data from each case is summarised in Table 4.5. Note that that the inverter modulation index is maintained at 100% throughout the testing, and that cases 2 and 3 are bench-marked against case 1.

#### Case 1 $(n = 500 \text{ rpm}, R_L = 0.5 \Omega)$

The calculated I-V locus indicates that a minimum peak current of about 19A occurs under this condition, however this will be slightly less, due to the rectifier ripple. Hence the inverter input current will fluctuate between the short-circuit current (21.6A) and about 18A. This is verified by Figure 4.24, which shows the simulated and the measured experimental inverter input current, as well as the resulting resistive load current. The load current contains very little distortion as a filter capacitance of  $2,000\mu$ F is used. This value results in a filter cutoff frequency of approximately 160Hz, which corresponds to 4% of the PWM switching frequency.



Figure 4.23: Calculated generator I-V locus for 500 and 1,000 rpm. The dashed lines represent resistive load lines of 0.5 and 5 $\Omega$ , and the circles correspond to three test cases.



Figure 4.24: Simulated (a), and measured experimental (b) inverter currents, for case 1. The waveforms shown are the rectifier (top) and filter (bottom) output currents. The vertical and horizontal scales are 10A and 5ms per division, respectively.

#### Case 2 $(n = 1000 \text{ rpm}, R_L = 0.5 \Omega)$

The inverter load remains unchanged, however the generator speed has doubled. Although the open-circuit voltage has doubled (from 45V to 90V), the I-V locus indicates that the peak load current will only slightly increase from 18 to 20A. Hence, the inverter input current is expected to fluctuate between the short-circuit (21.6A) current and about 20A (allowing for the rectifier ripple). This is verified by Figure 4.25, which shows the simulated and experimentally measured inverter input and load currents. The increased speed clearly makes the input current more constant, by reducing the magnitude of the 100Hz ripple. This slightly increases the fundamental of the load current, and significantly improves the measured total harmonic distortion (THD), from 4.0% to 3.0%. Note that the increase in speed also increases the rectifier ripple frequency.



Figure 4.25: Simulation (a), and measured experimental (b) inverter currents, for case 2. The waveforms shown are the rectifier (top) and filter (bottom) output currents. The vertical and horizontal scales are 10A and 5ms per division, respectively.

#### Case 3 $(n = 500 \text{ rpm}, R_L = 5\Omega)$

The generator speed is reduced to 500rpm. The load resistance is increased ten-fold (to  $5\Omega$ ), whilst the filter capacitance is reduced by a factor of 10, to maintain the filter time-constant. With an ideal constant current source input, the peak output voltage should increase by a factor of ten to about 98V. However at 500rpm, the rectified opencircuit voltage is 45V, and so the constant current assumption breaks down. The locus also suggests that the inverter input current will fluctuate between the short-circuit current and about 5.7A (allowing for the rectifier ripple). This is verified by the simulated and measured inverter input and load currents, shown in Figure 4.26. The inverter input current fluctuates significantly and the subsequent load current is highly distorted, i.e. its THD is 35.8%, compared to 4.0% for case 1. Note that the open-loop control algorithm causes the output current. In addition, the peak input current now exceeds the SC current. This is because the SC current seen in the I-V locus is that under steady-state conditions, whilst the inverter input currents seen here are transient waveforms.



Figure 4.26: Simulation (a), and measured experimental (b) inverter currents, for case 3. The waveforms shown are the rectifier (top) and filter (bottom) output currents. The vertical and horizontal scales are 10A and 5ms per division, respectively.

Table 4.5: Summary of simulated (SIM) and measured (MEAS) inverter performance, for cases 1–3.

	Case 1		Case 2		Case 3	
	SIM	MEAS	SIM	MEAS	SIM	MEAS
Input current (A)	20.0	21.0	21.2	22.0	15.4	15.1
Input current ripple $(A_{pk-pk})$	8.25	8.80	4.68	5.10	24.1	22.2
Output current (A)	12.5	12.8	14.1	13.8	5.88	5.80
Output voltage (V)	6.49	7.55	7.33	8.20	29.4	31.0
Output power (W)	81.2	85.0	103	98.8	173	168
Output current THD $(\%)$	4.10	4.00	1.90	3.00	35.8	35.8

Comparison of the above input current waveforms shows that it contains two forms of ripple. These are referred to as the low and medium-frequency ripples (relative to the high-frequency PWM frequency). The medium-frequency ripple is caused by the rectifier and is proportional to the generator speed. It does not contribute much to the output current THD, as these frequencies are beyond the filter cutoff frequency, i.e. the filter attenuates these harmonics. In contrast, the low-frequency ripple occurs at frequencies below the cutoff and hence are not attenuated by the low-pass filter.

The low-frequency input current ripple is caused by the low-cost inverter design, which avoids a DC link inductor (energy storage device) to improve the overall efficiency. The consequence of this is that the instantaneous input and output powers must be equal, which implies that the inverter input current and voltage must fluctuate at twice the grid frequency (100 Hz). The magnitude of the peak-to-peak inverter input current ripple is shown to be related to the ratio of the load voltage to the open-circuit voltage (recall Figure 4.23). It is seen that the 100Hz component of the inverter input current ripple decreases as the voltage ratio approaches zero. This implies that the inverter output current distortion can be reduced if the load voltage is much less than the rectifier open-circuit voltage, i.e. the generator operates in the constant current region of its I-V locus. The sources of input and output current harmonics are further discussed in Chapter 5.

## 4.4.3 Grid-Connected Testing

#### Proof of Concept - Grid-Connected Operation

Once the resistive load and grid voltages are synchronised, switch SW2 is closed and the artificial grid (autotransformer output) is connected in parallel with the resistive load (refer to the synchronisation procedure in Section 4.3.1). In this case, ideally the grid should provide zero current and the output power of the inverter should still be absorbed by the load resistance. The currents for this case are shown in Figure 4.27 and include the inverter output, grid drawn and resistive load currents. Note that the resistive load current is simply the summation of the inverter and grid drawn currents.



Figure 4.27: Simulated (a), and measured experimental (b) inverter currents, for the intermediate grid-connected stage, i.e. resistive / capacitive load in parallel with the grid. The horizontal and vertical scales are 10A and 5ms per division, respectively.

The pure grid-connected case is obtained by removing the load resistance (opening switch SW1 of Figure 4.12), which causes the inverter current to flow through the autotransformer's leakage inductance into the mains. The simulated and experimentally measured inverter output current and artificial grid voltage waveforms, for an autotransformer output voltage of  $15 V_{RMS}$ , are shown in Figure 4.28. Note that the simulation is based on the equivalent transformer model shown in Figure 4.12 with equivalent resistance and inductance values taken from Figure 4.13.



Figure 4.28: Simulated (a), and measured experimental (b) : inverter output current and artificial grid voltage waveforms, for the purely grid-connected stage. The voltage and current vertical scales are 5V and 10A per division, respectively.

The inverter output current above is shown to lag the grid voltage by about 27°, which equates to a power-factor of approximately 0.89. However, according to the Australian Standards, the inverter must be considered as a load of the grid, hence the inverter current is actually <u>leading</u> the grid voltage. This can also be seen from Figure 4.9, which shows that in the absence of the inverter, the grid feeds an LC network. The high capacitive reactance hence draws a leading current from the grid. Therefore, the inverter is shown to comply with the power factor requirements of the Australian Standards, i.e. 0.95 lag to 0.8 lead.

In contrast, the inverter output current of Figure 4.28 has a THD of about 12.5%, which exceeds the grid THD requirements of 5%. The output current contains more distortion than that for the resistive load case, this is due to i) resonance between the filter capacitor and transformer leakage reactance, and ii) a higher cutoff frequency. The output current THD must be reduced before the inverter can be made commercially

available. Two possible techniques to reduce the output THD are hence investigated, these include the reduction of i) the artificial grid voltage, and ii) the inverter modulation index. Both techniques should push the generator closer to the constant current region of the I-V locus, and hence reduce the input current ripple and output current distortion.

#### Effect of Grid Voltage Variation

The inverter current was monitored as the artificial grid voltage was varied between RMS voltages of 5 and 20V. This test was performed twice. Initially a fixed filter capacitance of  $600\mu$ F was used, which caused high amounts of distortion. The test was repeated and the capacitance was varied with grid voltage, in an effort to reduce the output current THD. The filter capacitance ranged from 200 to  $2,000\mu$ F, and was varied for each grid voltage such that the output current THD was minimised. This was determined by visual inspection of the inverter output current waveform. Examples of simulated and experimentally measured inverter output currents for the three artificial grid voltages are shown in Figure 4.29, for the variable capacitance case. The corresponding values of filter capacitance is shown in Figure 4.30. It is seen in Figure 4.29 (by inspection) that with increasing grid voltage, the output current decreases in magnitude while the THD increases. This occurs as the generator is shifted toward the voltage source region, where the constant current assumption begins to break-down, see Figure 4.23.



Figure 4.29: Simulated (a), and measured experimental (b) grid-connected inverter currents, for various grid voltages. The currents shown correspond to RMS grid voltages of (top) 6.5V, (middle) 10V, and (bottom) 15V. The vertical and horizontal scales are 10A and 5ms per division, respectively.



Figure 4.30: Variation of filter capacitance with artificial grid voltage, for both the fixed and varied capacitance tests.

The autotransformer turns ratio and hence its equivalent leakage inductance and resistance increase with grid voltage. These parameters are important as they affect the filter resonant frequency,  $f_{res}$ , and its gain at the resonant frequency (quality factor, Q). Therefore harmonics present at the resonant frequency may be potentially amplified, which will increase the overall total harmonic distortion. A summary of the effect of grid voltage variation on the resonant frequency and quality factor is shown in Figure 4.31. Note that the rectifier ripple frequency is 1,200 Hz in this case, and hence the rectifier ripple is attenuated by the filter rather than amplified by the filter resonance.

Figures 4.31(a) and 4.31(b) indicate that the resonant frequency decreases, while the quality factor remains relatively constant for the fixed capacitance case. In contrast, the resonant frequency and quality factor both increase with grid voltage for the variable capacitance case. Both sets of observations indicate that the THD will increase with grid voltage, due to the effects resonance. In addition, the THD will also increase as the constant current assumption breaks down (grid voltage increases), due to increased low order harmonics. This coupled with a quality factor greater than 1 and the decreasing resonant frequency, suggests that the THD for the fixed capacitance case should be greater than that for the variable capacitance case. This is verified by Figure 4.32, which shows the



Figure 4.31: Calculated (a) resonant frequency, and (b) quality factor vs. grid voltage, for both the varied and fixed capacitance cases.

simulated and experimentally measured THD for the four different grid voltages mentioned above. Despite the discrepancies between the simulated and measured data, they both verify that the inverter output current THD increases with grid voltage.



Figure 4.32: Inverter output current THD vs. grid voltage, for both fixed and variable filter capacitance cases. The solid lines represent simulated THD, whilst the points correspond to experimentally measured data.

The discrepancy between the measured and simulated THD data is likely caused by the equivalent transformer modelling. The transformer model is an approximation only, which ignores the magnetising reactance and core loss resistance. Note that this will have an effect on the quality factor, i.e. the gain at the resonant frequency. This appears somewhat valid as the simulated current waveforms show a slight increased effect of resonance on the output current THD, compared to those captured experimentally. Despite this, the close matching simulated and experimentally captured current waveforms of Figure 4.29 give confidence in the PSIM<sup>®</sup> model.

#### Effect of Modulation Index Variation

The PWM modulation index was varied for the pure grid-connected case. The artificial grid voltage was set to  $15V_{RMS}$  and a filter capacitance of  $400\mu$ F was selected to minimise the output current THD, for a  $m_a$  of 100%. The effect of modulation index variation on the simulated and measured output current waveforms is shown in Figure 4.33, for modulation indices of 100%, 80% and 60%. The simulated and measured current waveforms indicate that the inverter output current, and hence output power, is linearly proportional to  $m_a$  for a fixed grid voltage. This differs from the resistive load case, where the inverter output power is proportional to  $m_a^2$ .



Figure 4.33: Simulated (a), and measured experimental (b) grid-connected inverter currents, for various modulation indices. The currents shown correspond to modulation indices of (top) 100%, (middle) 80%, and (bottom) 60%. The vertical and horizontal scales are 10A and 5ms per division, respectively.

The inverter output current is expected to vary linearly with modulation index, if supplied by an ideal constant current source and loaded by a resistance. However, the inverter is loaded by the autotransformer secondary winding and is fed by the PM generator and rectifier current source. The (PSIM<sup>®</sup>) simulated and measured output currents and voltages are shown in Figure 4.34, over a wide range of modulation indices.



Figure 4.34: Inverter output (a) current, and (b) voltage for the grid-connected case, showing the effect of modulation index variation. The solid lines represent PSIM<sup>®</sup> simulations, whilst the points correspond to experimentally measured data.

Figure 4.34(a) shows that the inverter output current does not approach zero at low values of  $m_a$ . This is the result of the grid-connected filter, which draws a current from the grid when  $m_a$  is equal to 0%; the magnitude and power factor depend on the filter reactance. The output current appears to increase linearly at low values of  $m_a$ , and then saturate as  $m_a$  approaches 100%. This is caused by the non-ideal constant current source, i.e. the inverter pushes the generator further from the constant current region of the I-V locus as the  $m_a$  increases. The degree of the saturation varies with grid voltage. For example, if the generator remains in the constant current region for a  $m_a$  of 100%, the inverter output current will be linearly proportional to the modulation index over the entire range of  $m_a$ , except at low values, where the output current will approach that drawn by the filter.

Figure 4.34(b) shows how the inverter output voltage varies with modulation index. The inverter has a non-zero output voltage for  $m_a$  equal to 0%, due to the voltage drop across the (grid-connected) filter capacitance, and the output voltage is shown to increase with modulation index. This is due to the autotransformer regulation, i.e. the voltage drop across the transformer equivalent inductance increases with the inverter output current (recall Figure 4.12). Therefore, the inverter output voltage must increase with modulation index as this action increases the inverter output current.

It was seen above that the inverter output current saturates as the modulation index approaches 100%, as the PM generator is moved away from the constant current region of the I-V locus. It is hence expected that as the  $m_a$  increases, the input current distortion and hence output current distortion will also increase. This is verified by the simulated output current THD in Figure 4.35. The figure also includes experimentally measured THD data, which does not agree with the simulations over the entire range of  $m_a$ . The measured THD does decrease with modulation index, until  $m_a$  is equal to about 60%, the THD then increases as  $m_a$  approaches 0%. The harmonic analysis of the measured current waveforms indicate that resonance causes the increased THD. The measured resonant frequency is 450Hz. PSIM<sup>®</sup>, however, predicts a resonant frequency of 411Hz, and is likely due to the approximation of the transformer parameters. Despite the discrepancy between the simulated and measured THD, which could be further investigated, the simulated current waveforms closely match those measured throughout this chapter and give confidence in the PSIM<sup>®</sup> model.



Figure 4.35: Inverter output current THD vs. modulation index. The solid line represents the simulated THD, whilst the points represent the measured data.

# 4.5 Chapter Summary

The proposed novel grid-connected current-source inverter topology, based on the highinductance PM generator, uncontrolled rectifier, single-switch current wave-shaper, unfolding circuit (H-bridge inverter) and low-pass filter, has been designed and simulated in PSIM<sup>®</sup>. The inverter concept was successfully demonstrated using a 150W prototype that was experimentally tested using a dynamometer, and controlled in an open-loop manner. The current wave-shaper switch was modulated to create a rectified sinusoidal output current, which was unfolded and filtered by the H-bridge inverter and low-pass filter, respectively. The measured inverter output current waveforms, for both resistive and grid-connected loads, closely matched the simulation results, which gave confidence in the model.

The use of an open-loop control method meant that variations in the rectifier output current appeared in the inverter output current. The rectifier output current was modulated by the rectifier ripple and the 100Hz instantaneous power fluctuations due to the single-phase output and the lack of input DC link energy storage. The magnitude of the rectifier ripple was fixed and its varied with generator speed. In contrast, the lack of DC link energy storage caused the input current to fluctuate at twice the grid frequency; its ripple magnitude varied with inverter output power. This was experimentally verified by firstly varying the resistance and capacitance values of three RC load combinations (for a fixed load time-constant), and secondly by increasing the effective (artificial) grid voltage seen by the inverter during the grid-loaded testing stage.

The inverter low-pass filter, which utilised the grid transformer leakage reactance and resistance, demonstrated its ability to attenuate the high-frequency current harmonics caused by the unipolar PWM switching. Inspection of the output current, however, revealed a significant ninth-order harmonic, which was caused by resonance between the filter capacitor and transformer inductance. This prevented the inverter from meeting the THD grid requirement.

# Chapter 5

# **Inverter Analysis and Control**

This chapter analyses the proposed grid-connected inverter. The non-ideal input current source, and its effect on the inverter output current is studied. The harmonic distortion of the non-ideal current source and that of the PWM switching is examined. Four techniques to attenuate unwanted harmonics are suggested and briefly examined. Of these, the low-pass filter and feed-forward controller are further discussed. The low-pass filter is analysed and design trade-offs are shown. The feed-forward controller is also designed and its ability to significantly improve the output current THD is demonstrated.

## 5.1 Analysis of Non-Ideal Constant Current Source

The grid-connected inverter concept assumes that the rectifier ideally provides constant current to the inverter, however, the previous chapter showed that the inverter input current was non-ideal. An example of the ideal and non-ideal inverter input currents is shown in Figure 5.1(a). The figure indicates that the harmonics in the output current are related to the harmonics found in the input current. The large second-order input current harmonics become large third-order output current harmonics, as a result of the sinusoidal modulation.

The input current shows two distinct forms of ripple. The higher frequency ripple is caused by the use of the uncontrolled three-phase rectifier, whilst the low-frequency ripple (twice that of the mains) is caused by the power fluctuations associated with the single-phase inverter design. The lack of an energy storage element implies that the instantaneous inverter input and output powers must be equal. Assuming unity power factor operation, the inverter delivers a time-varying (fluctuating) power that is the square



Figure 5.1: Normalised inverter (a) input, and (b) output currents for ideal and non-ideal inverter input currents.

of a sinusoid, whose average power is half that of its peak. This fluctuating input power will cause the generator to vibrate [91] and the inverter input current to vary. Each source of distortion is further discussed below. Note that the effect of the PWM switching and that of the low-pass filter is ignored in this section. In addition, the inverter input and output currents and voltages are expressed as normalised quantities throughout this chapter, i.e. they have peak values of 1pu.

## 5.1.1 Fluctuating Input Current

Part I showed that the PM generator is able to provide constant current, and hence power, to a battery or resistive load. However, the proposed single-phase inverter delivers a timevarying output power. This, coupled with the energy storage-less inverter design, implies the inverter input power must also vary with time (fluctuate). It follows that the inverter input voltage varies with time. For a practical generator, this voltage variation produces an output current variation. This variation is analysed by considering the time-varying rectifier output (inverter input) voltage,  $v_R(t)$ , in terms of the current wave-shaper output and the grid voltage,  $v_{ws}(t)$  and  $v_q(t)$ , respectively.

Figure 4.29 previously showed that the inverter output current THD increased with the artificial grid voltage for a fixed modulation index. This occurred as the generator was operated closer to the variable voltage region of the I-V locus, which reduces the machine's ability to provide constant current. This is further analysed in this section. Note that all voltages hereafter are normalised relative to the rectifier open-circuit voltage. Consider the time-varying inverter output (grid) voltage. Its instantaneous value, relative to that of the rectified open-circuit voltage, is defined as  $\alpha_0$ ; e.g. a peak  $\alpha_0$  of 0.5pu corresponds to a rectified open-circuit voltage of 679V. The normalised grid voltage is hence expressed in terms of peak  $\alpha_0$  (i.e.  $\hat{\alpha}_0$ ), as shown in Equation (5.1), where  $\omega_g$ represents the grid angular frequency. The unfolding circuit input voltage can also be expressed as a function of  $\hat{\alpha}_0$ , as it is simply the rectified equivalent of its output voltage. This is shown in Equation (5.2). Note that  $\hat{\alpha}_0$  is a scalar that ranges between 0 and 1.

$$v_g(t) = \hat{\alpha}_0.\sin(\omega_g t) \quad \text{(pu)} \tag{5.1}$$

$$v_{ws}(t) = |v_g(t)| = \widehat{\alpha}_0 |\sin(\omega_g t)| \quad (pu)$$
(5.2)

The rectifier output voltage can also be expressed in terms of  $\hat{\alpha}_0$ , and is derived from the time-varying ideal boost equation shown in Equation (5.3). Note that the duty-cycle, d, was previously kept constant for a battery load. However, it is now time-varying as shown in Equation (5.4), where  $m_a$  represents the inverter modulation index. The rectifier voltage is hence expressed by Equations (5.5) and (5.6).

$$v_R(t) = [1 - d(t)] v_{ws}(t)$$
 (pu) (5.3)

$$d(t) = 1 - m_a |\sin(\omega_g t)| \quad (pu)$$
(5.4)

$$v_R(t) = m_a |\sin(\omega_g t)| \cdot v_{ws}(t) \tag{5.5}$$

$$= m_a.\widehat{\alpha}_0.\sin^2(\omega_g t) \quad (pu) \tag{5.6}$$

A graphical summary of the normalised grid, (current) wave-shaper and rectifier output voltages (Equations (5.1), 5.2 and (5.6), respectively) is shown in Figure 5.2. The figure shows that the grid voltage and wave-shaper output voltages are sinusoids with a peak value equal to  $\alpha_0$ . The peak rectifier voltage, however, is proportional to the inverter modulation index, and the figure shows normalised voltages for cases where  $m_a$  is equal to 100 and 50%. Note that the grid and current wave-shaper output voltage waveforms are independent of  $m_a$ , i.e. their magnitudes are set by the grid voltage.

The inverter input current can be derived from the normalised rectifier I-V locus, which is shown in Figure 5.3 along with the locus of an ideal current source. The figure shows both the ideal and the measured (PM machine) loci, where  $\alpha$  and  $\beta$  represent the



Figure 5.2: Output voltage of various inverter stages, showing the grid, wave-shaper and rectifier output voltages. The rectifier output voltage is a sinusoid squared, and is shown for modulation indices,  $m_a$ , of 100 and 50%.

normalised rectifier voltage and current, respectively. Therefore,  $\alpha = 1$  pu represents an open-circuit, whilst  $\beta = 1$  pu represents a short-circuit. The ideal locus neglects the effect of the rectifier (as described in Section 2.3.1), hence the ideal current,  $\beta_{(id)}$ , is given by Equation (5.7), which is based on the unit circle equation, i.e.  $x^2 + y^2 = r^2$ . In contrast, the dashed line, which represents a fitted curve corresponding to the measured data,  $\beta_{(exp)}$ , is expressed as a quintic function of  $\alpha$ , as shown in Equation (5.8).



Figure 5.3: Normalised ideal and fitted experimental rectifier I-V loci, where  $\alpha$  and  $\beta$  represent the voltage and current, respectively. The points represent measured data.

$$\beta_{(id)} = \sqrt{1 - \alpha^2} \tag{5.7}$$

$$\beta_{(exp)} = 3.86\alpha^5 - 7.47\alpha^4 + 4.27\alpha^3 - 1.53\alpha^2 - 0.116\alpha + 1 \tag{5.8}$$

The normalised rectifier current is determined by tracing the time-varying normalised rectifier voltage (from Figure 5.2) on to the rectifier I-V locus, and plotting the corresponding current. This process is demonstrated using the ideal rectifier I-V locus in Figure 5.4 for a peak normalised rectifier voltage ( $\hat{\alpha}$ ) of 0.8 and 0.6 pu, which results in a fluctuating rectifier current,  $i_R(t)$ , with minima of 0.6 and 0.8, respectively. The experimental inverter input current is obtained similarly and is expected to contain greater harmonic distortion than the ideal case, as it is less effective as a constant current source.



Figure 5.4: Normalised ideal inverter input (rectifier output) current, derived from the rectifier output voltage and the normalised rectifier I-V locus. The inverter input currents shown correspond to the cases where  $\hat{\alpha} = 0.6$  and 0.8 pu.

The ideal boost equation, shown in Equation (5.9), is used to determine the waveshaper output current,  $I_{ws}$ . Consider the ideal I-V locus case. Using Equations (5.7) and (5.8) and substituting the terms (1-d),  $I_R$  and  $I_{ws}$  for their time-varying equivalents, allows the ideal wave-shaper output current,  $i_{ws \ (id)}(t)$ , to be expressed by Equation (5.10). The ideal inverter output current,  $i_{out \ (id)}(t)$ , is simply the unfolded equivalent of the waveshaper output current, as given below in Equation (5.11). Examples of the inverter input and subsequent output currents, using both the ideal and experimental loci over a wide range of  $\hat{\alpha}_0$ , is shown in Figure 5.5. The  $m_a$  is 100% for all cases.

$$I_{ws} = (1 - d) \ I_R \tag{5.9}$$

$$i_{ws \ (id)}(t) = m_a |\sin(\omega_g t)| \sqrt{1 - (m_a \cdot \hat{\alpha}_0 |\sin^2(\omega_g t)|)^2}$$
 (pu) (5.10)

$$i_{out \ (id)}(t) = m_a . \sin(\omega_g t) \ \sqrt{1 - \left(m_a . \widehat{\alpha}_0 |\sin^2(\omega_g t)|\right)^2} \quad (\text{pu})$$
(5.11)

Figure 5.5 shows the inverter input and output currents for various values of  $\hat{\alpha}_0$   $(m_a = 100\%)$  using both the ideal and experimental I-V loci. The importance of a constant input current source is clearly seen. The fluctuating input current is caused by the low-cost single-phase inverter design, and could be reduced by modifying the inverter design, e.g. introducing an energy storage element or outputting three-phase power.

#### 5.1.2 Rectifier Ripple

The uncontrolled full-wave three-phase rectifier is used to convert the generator's AC output currents to a DC current. The drawback is that the rectifier output (inverter input) current contains ripple. This time-varying rectifier ripple, rect(t), fluctuates at six times the generator frequency and has a magnitude of 13.34%  $(1 - \sqrt{3}/2)$  of the peak AC line current. This is expressed in Equation (5.12), where  $\omega_m$  is the generator electrical frequency. An example of the effect of the rectifier ripple is shown in Figure 5.6, which shows the distorted input and subsequent inverter output current. Note that only the effect of the rectifier ripple is shown (i.e.  $\hat{\alpha}_0$  is zero).

$$rect(t) = \frac{|\sin(\omega_m t)| + |\sin(\omega_m t + \frac{2\pi}{3})| + |\sin(\omega_m t + \frac{4\pi}{3})|}{2}$$
(5.12)



Figure 5.5: Normalised inverter input (top) and output (bottom) currents, using the ideal (left) and experimental (right) I-V loci. The currents are displayed for  $\hat{\alpha}_0$  ranging from 0 to 1pu, in 0.25 increments. The output current contains the distortion found in the input current. There is greater distortion using the experimental I-V locus.



Figure 5.6: Normalised inverter (a) input and (b) output currents, showing the effect of the rectifier ripple ( $\hat{\alpha}_0 = 0$ ) for a machine frequency of 200Hz (rectifier ripple frequency of 1.2 kHz).

#### 5.1.3 The Resulting Input Current

The inverter input current,  $i_R(t)$ , contains distortion from the rectifier ripple and the fluctuating inverter input power, and is expressed by Equation (5.13). Substituting  $\beta$  with  $\beta_{(id)}$  (Equations (5.6) and (5.7)), and the term rect(t) (Equation (5.12)), allows the ideal inverter input current,  $i_{R(id)}(t)$ , to be represented by Equation (5.14). The resulting (simplified) ideal output current,  $i_{out}(id)(t)$ , is shown in Equation (5.15).

$$i_R(t) = rect(t) \cdot \beta$$
 (pu) (5.13)

$$i_{R\ (id)}(t) = \frac{|\sin(\omega_m t)| + |\sin(\omega_m t + \frac{2\pi}{3})| + |\sin(\omega_m t + \frac{4\pi}{3})|}{2} \sqrt{1 - (m_a \cdot \widehat{\alpha}_0 |\sin^2(\omega_g t)|)^2} \tag{pu}$$
(5.14)

$$i_{out \ (id)}(t) = m_a . \sin(\omega_g t) . i_{R \ (id)}(t)$$
 (5.15)

An example of the inverter input and its subsequent output current is shown in Figure 5.7, for the ideal locus,  $\hat{\alpha}$  value of 0.7pu ( $\hat{\alpha}_0 = 0.7, m_a = 100\%$ ) and a machine frequency of 200Hz. Note that these current waveforms are identical to those in Figure 5.1. The input current is mainly caused by the fluctuating input power, however, the rectifier ripple also contributes. It is expected that the rectifier associated distortion will dominate at low values of  $\hat{\alpha}$ . This is shown in Figure 5.8, which shows the inverter input and output current for two cases of  $\hat{\alpha} = 0.20$ pu ( $\alpha = m_a.\alpha_0$ ). Case one has  $\hat{\alpha}_0 = 0.20$  and  $m_a = 100\%$ ,



Figure 5.7: Normalised ideal inverter (a) input and (b) output currents, showing the combined effect of the rectifier ripple ( $f_m = 200$ Hz) and the fluctuating input current ( $\hat{\alpha}_0 = 0.7$ pu).

whilst case two has  $\hat{\alpha}_0 = 0.50$  and  $m_a = 40\%$ . Both input currents are identical. The output currents waveforms are identical in shape, however, differ in magnitude. This highlights that the input current depends on  $\hat{\alpha}$ , while the ratio of the output to the input current depends on the  $m_a$ .



Figure 5.8: Normalised ideal inverter (a) input and (b) output current, for a generator frequency of 200Hz and two cases of  $\hat{\alpha}$  equal to 0.2pu ( $\hat{\alpha} = 0.2$ pu,  $m_a = 100\%$ , and  $\hat{\alpha} = 0.5$ pu,  $m_a = 40\%$ ).

The inverter output current distortion is quantified in the next section.

## 5.2 Total Harmonic Distortion

The amount of distortion in a waveform is quantified by the means of an index called the *total harmonic distortion* (THD) [67]. The THD is a ratio of the total harmonic components,  $h_{tot}$ , to the harmonic magnitude of the fundamental frequency,  $h_1$ , as shown in Equation (5.16), where  $h_m$  specifies the harmonic magnitude found at integer multiples of the fundamental frequency, e.g.  $h_3$  and  $h_{11}$  represent the third and  $11^{th}$ -order harmonics, respectively.

THD (%) = 
$$100 \times \frac{h_{tot}}{h_1} = \frac{\sqrt{\sum_{m \neq 1} h_m^2}}{h_1}$$
 (5.16)

Note that an ideal waveform is a sinusoid, which has zero harmonic components and hence zero distortion. The harmonic content is determined by the *fast Fourier trans-form* (FFT), which is an efficient algorithm that computes the *discrete Fourier transform* (DFT) [107]. The DFT is a signal processing technique that is used to identify the magnitude and frequency of harmonics that are embedded within signals.

In addition to the rectifier ripple and fluctuating input current, the unipolar PWM switching of the current wave-shaper significantly contributes to the inverter output current THD. Each of these harmonic sources are individually discussed and analysed in the following sections. As previously stated, Australian Standard 4777.2 requires that the output current of a grid-connected inverter must contain less than 5% THD [96].

#### 5.2.1 Fluctuating Input Power

It was previously discussed that the lack of energy storage causes the inverter input current to fluctuate at twice the grid frequency for a single-phase output, recall Figure 5.5. The figure showed the effect of the fluctuating input power on the inverter output current for  $\hat{\alpha}_0$  ranging from from 0 to 1pu. These output current waveforms suggest that the thirdorder harmonic,  $h_3$ , increases and that the fundamental decreases, with increasing  $\hat{\alpha}_0$ . This is confirmed by Figure 5.9, which shows the ideal locus output currents for  $\hat{\alpha} = 0.50$ and 0.75pu ( $m_a = 100\%$ ), and the effect of varying  $\hat{\alpha}$  on the fundamental, third, fifth and seventh-order harmonics ( $h_1$ ,  $h_3$ ,  $h_5$ ,  $h_7$ ). Although higher frequency odd-order harmonics exist, these are ignored as their magnitudes are much less than those of  $h_{3-7}$ , and do not significantly affect the THD. It is expected that the THD will rise rapidly with increasing  $\hat{\alpha}$  as the harmonic content increases while the fundamental decreases.



Figure 5.9: Normalised inverter output current for  $\hat{\alpha} = 0.50$  and 0.75 pu (left), and the harmonic response to increasing  $\hat{\alpha}$  (right), using the ideal I-V locus. Note that the output current is shown for  $m_a = 100\%$ .

Similarly, Figure 5.10 shows the inverter output currents for the same values of  $\hat{\alpha}$ , and the harmonic response to increasing  $\hat{\alpha}$ , now using the experimental I-V locus. As with the ideal I-V locus, the magnitude of the fundamental decreases, whilst the magnitudes of the third, fifth and seventh-order harmonics increase for increasing  $\hat{\alpha}$ . The key differences are that the fundamental falls more rapidly and the third-order harmonic increases more rapidly than that for the ideal case (locus). The fifth and seventh-order harmonics are similar for the ideal and experimental loci. The only difference is for  $\hat{\alpha}$  between 0.9 and 1pu, where the ideal locus harmonics are slightly higher than those corresponding to the experimental locus. Note that the relatively small magnitudes in  $h_5$  and  $h_7$  have little effect on the resulting THD.



Figure 5.10: Normalised inverter output current for  $\hat{\alpha} = 0.50$  and 0.75 pu (left), and the harmonic response to increasing  $\hat{\alpha}$ , (right) using the experimental I-V locus. The output current is shown for  $m_a = 100\%$ .

The increase in third-order harmonics, together with the decrease in fundamental magnitude, imply that the experimental locus will have a higher value of THD for the entire range of  $\hat{\alpha}$  values. This is confirmed by Figure 5.11, which shows the calculated THD vs.  $\hat{\alpha}$  for both loci, where the dashed line represents the grid THD limit of 5%. Note that the THD values shown here ignore the effect of the rectifier current ripple. The figure suggests that for a DC (fluctuating) input current, the grid THD requirements can be met providing  $\hat{\alpha}$  is limited to 0.52 or 0.34 pu, using the ideal and experimental loci, respectively. Note that the conduction losses of the uncontrolled rectifier cause the experimental I-V locus to differ from the analytical ideal case, and hence show the importance of reducing its losses.



Figure 5.11: Calculated THD vs.  $\hat{\alpha}$  for both the ideal and experiential rectifier I-V loci. The dashed line represents the acceptable THD limit.

#### 5.2.2 Rectifier Ripple

The normalised three-phase rectifier ripple has a fixed magnitude, but varying frequency. This occurs as the turbine operates over a wide range of speeds, and the rectifier ripple frequency is six times the machine frequency,  $f_m$ . The inverter output current and the corresponding harmonic spectrum are shown in Figure 5.12 for a grid frequency of 50Hz and a machine frequency of 200Hz.



Figure 5.12: Normalised inverter output current showing the effect of the rectifier ripple (left), and its calculated harmonic spectrum (right);  $h_f$  represents harmonic frequencies.

The harmonic frequencies,  $h_f$ , are given by Equation (5.17), where  $f_1$  is the fundamental frequency, and m is an integer. The output current THD is calculated from the FFT spectrum and is equal to 4.20%. This THD is independent of the machine frequency and the type of generator locus used.

$$h_f = m \, 6f_m \pm f_1 \tag{5.17}$$

### 5.2.3 The Resulting Input Current

The inverter input current contains distortion from both the rectifier ripple and the fluctuating input current, hence, these should be seen in the output current and the corresponding FFT. This is shown in Figure 5.13 for  $\hat{\alpha} = 0.7$  pu ( $\hat{\alpha}_0 = 0.7$ ,  $m_a = 100\%$ ), using the ideal I-V locus. The harmonics are identified by their source.



Figure 5.13: Normalised inverter output current showing the effect of the rectifier ripple and the fluctuating input power (left), and the corresponding harmonic spectrum (right).

The harmonic spectrum of the output current shows that the harmonics caused by the fluctuating input current can be found at frequencies expressed by Equation (5.18). In addition, the harmonics caused by the rectifier ripple are expressed by Equation (5.19), where m is an integer and n is an odd integer. Although multiple sidebands now appear centred around  $m 6 f_m$ , Figure 5.13 shows only two sidebands centred at 1.2kHz and 1 pair thereafter. Sidebands do not appear in this figure for n > 3 or m > 2, as their magnitudes are negligible, i.e. < 0.1% of the fundamental magnitude.

$$h_f = nf_1 \qquad (n > 1) \qquad (5.18)$$

$$h_f = m \, 6 f_m \pm n f_1 \qquad (m, n \ge 1)$$
 (5.19)

The use of a rectifier causes the inverter output current to contain a minimum of 4.19% THD. The grid limit of 5% hence allows for a 2.71% margin ( $\sqrt{5^2 - 4.2^2}$ ). Therefore, the inverter is able to meet the grid THD requirement, if  $\hat{\alpha}$  is limited to 0.39 and 0.23pu, using the ideal and experimental I-V loci, respectively (see Figure 5.11). Recall that  $\hat{\alpha} = \hat{\alpha}_0 m_a$ . This implies that as the machine speed decreases, so too does the induced voltage and hence  $\hat{\alpha}_0$ . Therefore, the modulation index should be reduced with speed to maintain a constant output current THD. This concept is shown in Figure 5.14 which shows the output current THD as a function of  $m_a$  for various values of  $\hat{\alpha}_0$  (pu).



Figure 5.14: Output current THD vs. modulation index and  $\hat{\alpha}_0$  (pu), for the (a) ideal, and (b) experimental I-V loci. The dashed line represents the grid THD limit of 5%.

The THD information seen in Figure 5.14 can be rearranged and presented as a contour plot, as shown in Figure 5.15 for THD contours of 4.2, 5 and 10%. These contours represent just over the minimum output current THD (caused by the rectifier ripple), the acceptable grid limit and twice the acceptable limit. The THD contours correspond to fixed values of  $\hat{\alpha}$ , e.g. an output current THD of 5% corresponds to  $\hat{\alpha}$  values of 0.394 and 0.231 pu using the ideal and experimental loci, respectively. The figure shows that a constant THD can be maintained by reducing  $m_a$  in proportion with the generator speed ( $\hat{\alpha}$ ). Note that the rectifier ripple distortion occurs for all generator speeds, hence the THD for small values of  $\hat{\alpha}$  is 4.19%, as indicated by the shaded regions.



Figure 5.15: THD vs.  $\hat{\alpha}_0$  and modulation index contour plot, for the (a) ideal, and (b) experimental I-V loci. The shaded regions represents the minimum output THD, caused by the rectifier, while the dashed line represents the grid THD limit of 5%.

Figures 5.15 and 5.14 indicate that the inverter must reduce  $m_a$  as the generator speed decreases ( $\hat{\alpha}_0$  increases) to maintain a constant THD. The inverter may reduce the modulation index automatically as the turbine speed and hence power decrease with wind speed. This is further investigated in the following subsection.

#### Modulation Index and $\widehat{\alpha}_0$ Variation with Wind Speed

Recall that  $\alpha_0$  represents the ratio of the peak grid to rectified open-circuit voltage, and that the peak turbine speed occurs at rated wind speed. This implies that the minimum  $\hat{\alpha}_0$  will occur at rated wind speed and that  $\hat{\alpha}_0$  will be larger at all other wind speeds. This is shown in Figure 5.16, which shows the turbine speed and the resulting  $\hat{\alpha}_0$  as a function of wind speed. Note that Figure 5.16(a) is the normalised equivalent of Figure 1.5(c).



Figure 5.16: Turbine speed (a) and the resulting  $\hat{\alpha}_0$  (b), as a function of wind speed. Note that  $\hat{\alpha}_0$  is equal to 0.23pu at rated wind speed.

Consider wind speeds below rated. The turbine and inverter output power, and hence the required modulation index increases with the cube of wind speed. Therefore,  $\hat{\alpha}$  must increase with the square of wind speed, as  $\hat{\alpha}_0$  is inversely proportional to wind speed (recall that  $\hat{\alpha} = \hat{\alpha}_0 \cdot m_a$ ). A summary of the modulation index and the resulting  $\hat{\alpha}$ is shown in Figure 5.17, over a wide range of normalised wind speeds. Note that the modulation index is 100% above rated wind speed as the inverter aims to deliver rated power to grid, and the resulting  $\hat{\alpha}$  is identical to  $\hat{\alpha}_0$  for this range of wind speeds.



Figure 5.17: Modulation index (a) and the resulting  $\hat{\alpha}$  (b), as a function of wind speed. Note that  $\hat{\alpha}$  is equal to 0.23pu at rated wind speed.

The open-loop inverter output current THD is determined from the above  $\hat{\alpha}$  curve, and the THD vs.  $\hat{\alpha}$  curve (see Figure 5.14(b)). The resulting output current THD is shown in Figure 5.18, as a function of normalised wind speed. The THD is shown to increase from the 4.19% at low wind speeds, which is solely caused by the rectifier ripple, up to the grid requirement at rated wind speed. The THD then increases above the 5% grid requirement, with wind speed, at rate proportional to  $\hat{\alpha}$ . The figure highlights that the inverter is able to meet the grid THD requirements below rated wind speed, providing a suitable generator is used ( $\hat{\alpha}_0$ ). However, the effects of resonance, i.e. input current harmonic amplification, are not considered here. This is examined in Section 5.3.9.



Figure 5.18: Open-loop inverter output current THD vs. wind speed. The dashed line represents the grid THD requirement, whilst the dotted line corresponds to the THD caused by the rectifier ripple.

#### 5.2.4 PWM Switching Schemes

In addition to the input current distortion caused by the non-ideal current source, the PWM switching of the current wave-shaper significantly contributes to the inverter output current THD. Two varieties of PWM switching exist, these are the bipolar and unipolar schemes. An example of each waveform is shown in Figure 5.19, for a modulation index of 100%. Their corresponding harmonic spectra are shown in Figure 5.20.



Figure 5.19: Normalised PWM switching scheme waveforms, showing the (a) Bipolar, and (b) Unipolar varieties, for a switching frequency of 1kHz.



Figure 5.20: Harmonic spectra of the (a) bipolar, and (b) unipolar PWM switching scheme waveforms. Note: the harmonic magnitudes now have linear scales, which were logarithmic in earlier figures.

The unipolar PWM waveform has lower harmonics than the bipolar waveform, as it changes polarity at twice the inverter frequency, as does a sinusoid, whereas the bipolar PWM waveform changes polarity at the inverter switching frequency,  $f_{sw}$ . The harmonic spectra shows that the harmonics appear as sidebands that are centred at integer multiples of the PWM switching frequency,  $f_{sw}$ . These harmonic frequencies,  $h_f$ , are expressed by Equation (5.20), where  $f_1$  is the fundamental frequency, m is an integer, and n is an odd integer. Comparison of the harmonic spectra indicate that the unipolar harmonics centred at  $m f_{sw}$  have the same amplitude as the bipolar harmonics centred at  $2m f_{sw}$ . The unipolar waveform does not contain the bipolar harmonics that are centred at odd multiples of  $f_{sw}$ , which is verified in [67]. The reduced harmonic content implies that the unipolar waveform contains less THD than the bipolar equivalent. Comparison of the calculated THDs confirms this, i.e. the bipolar PWM waveform THD is 96.2%, whilst the unipolar waveform THD is 49.5%. Despite the significant increase in THD, the bipolar PWM switching scheme is used more often than the unipolar alternative [108].

$$h_f = m f_{sw} \pm n f_1 \tag{5.20}$$

#### Effect of Modulation Index Variation

The bipolar PWM waveform harmonics are further analysed in [67], where the effect of varying the inverter modulation index is also taken into consideration. However, only half of this information applies to the unipolar PWM case, and as such, a detailed harmonic analysis of the unipolar PWM waveform is presented here. Note that the PWM switching frequency is set to 4kHz, as this corresponds to that used experimentally in Chapter 4.

The unipolar PWM waveform was simulated and analysed, over a wide range of modulation indices. It was found that the total harmonic components,  $h_{tot}$ , varied somewhat quadratically with modulation index, whilst the fundamental magnitude was directly proportional to  $m_a$ . The resulting THD is hence inversely proportional to the inverter modulation index, as shown in Figure 5.21, where the dashed line represents the grid THD limit of 5%. It is clearly seen that the output current fails to meet the THD requirement over the entire range of modulation index.



Figure 5.21: Calculated total harmonic amplitude (left), and distortion (right) vs. inverter modulation index, for a unipolar PWM waveform. Note that the fundamental magnitude is also shown with harmonic amplitude, as the THD is inversely proportional to it. The dashed line represents the grid THD limit of 5%.

The harmonic information presented in Figure 5.21 is summarised in Table 5.1. The table identifies the harmonic frequencies, their magnitudes, and the total harmonic distortion for various modulation indices, using an ideal DC input current ( $\hat{\alpha}_0 = 0$ ).

Table 5.1: Harmonic information regarding the unipolar PWM waveform, for various modulation indices. The harmonic frequencies are expressed as functions of the fundamental and inverter switching frequencies,  $(f_1 \text{ and } f_{sw})$ , whilst the harmonic magnitudes are expressed as percentages relative to the peak of the PWM waveform.

	1				
$m_a$ $h_f$	100%	80%	60%	40%	20%
h <sub>1</sub>	100	80	60	40	20
$f_{sw} \pm f_1$	18.15	31.45	37.01	32.58	18.95
$f_{sw} \pm 3f_1$	21.19	13.92	7.06	2.34	
$f_{sw} \pm 5f_1$	3.34	1.24			
$2\mathbf{f}_{sw} \pm f_1$	6.75	10.50	0.79	15.75	16.26
$2\mathbf{f}_{sw} \pm 3f_1$	1.07	11.48	13.21	6.93	1.15
$2f_{sw} \pm 5f_1$	11.88	8.38	3.40	0.64	
$2f_{sw} \pm 7f_1$	5.08	1.73			
$2f_{sw} \pm 9f_1$	0.88				
$3f_{sw} \pm f_1$	3.77	3.04	6.98	0.60	12.33
$3f_{sw} \pm 3f_1$	1.78	5.58	4.57	8.79	2.31
$3f_{sw} \pm 5f_1$	3.00	5.91	7.14	2.23	
$3f_{sw} \pm 7f_1$	6.20	6.07	2.09	0.22	
$3f_{sw} \pm 9f_1$	5.34	1.93	0.29		
$3f_{sw} \pm 11f_1$	1.78	0.35			
$3f_{sw} \pm 13f_1$	0.39				
$4\mathbf{f}_{sw} \pm f_1$	2.46	0.50	2.53	5.25	7.89
$4f_{sw} \pm 3f_1$	1.53	1.12	4.17	5.77	3.43
$4f_{sw} \pm 5f_1$	0.72	3.79	4.40	4.22	0.29
$4f_{sw} \pm 7f_1$	3.55	3.35	4.57	0.87	
$4f_{sw} \pm 9f_1$	2.39	4.70	1.46		
$4\mathbf{f}_{sw} \pm 11f_1$	4.67	2.00	0.23		
$4f_{sw} \pm 13f_1$	2.51	0.50			
$4f_{sw} \pm 15f_1$	0.78				
$4f_{sw} \pm 17f_1$	0.14				
÷	:	:	:	:	
$mf_{sw} \pm nf_1$					
$h_{tot}$	50.54	59.98	62.00	57.51	44.31
THD	50.54	74.97	103.34	143.78	221.53

#### 5.2.5 The Inverter Output Current

This section examines the combined effect of all three harmonic sources on the output current. An example of this is shown in Figure 5.22, which shows the distorted inverter input current and the resulting inverter output current. The input current shows the rectifier ripple and fluctuating input current, for a generator frequency of 200Hz, and  $\hat{\alpha}_0 = 0.8$ pu, respectively. The unipolar PWM switching operates at 4kHz, with a modulation index of 100%. The FFT of the inverter output current reveals the harmonic spectrum shown in Figure 5.23. The harmonics are grouped and identified by their source.



Figure 5.22: The normalised inverter input (left), and output (right) current, showing the effects of the rectifier ripple, fluctuating input power, and the unipolar PWM switching for  $\hat{\alpha} = 0.8$  pu ( $\hat{\alpha}_0 = 0.8$ , and  $m_a = 100\%$ ).



Figure 5.23: Harmonic spectrum of the inverter output current, identifying the harmonics caused by the rectifier, fluctuating input power, and the unipolar PWM switching.

Figures 5.22 and 5.23 indicate that the majority of the output current distortion is associated with the PWM switching. The harmonic spectrum also shows that the harmonics caused by fluctuating input power, rectifier, and PWM switching, occur at low, medium and high frequencies, respectively. It was previously shown, in Section 5.2.3, that the low-frequency harmonics could be reduced by reducing the modulation index. This action could be applied again, however, it will also reduce the fundamental magnitude and hence increase the overall THD. The effect of modulation index variation on the inverter output current is summarised in Figure 5.24, where the calculated THD includes the harmonic distortion caused by the rectifier, fluctuating input power and the unipolar PWM switching. The reduction of modulation index clearly increases the output current THD, with the exception of  $\hat{\alpha}_0$  greater than 0.75pu, where the output THD can be reduced for modulation indices above about 70%.



Figure 5.24: Calculated total harmonic distortion vs. modulation index for various values of  $\hat{\alpha}_0$ . The THD can be minimised, via modulation index manipulation, for  $\hat{\alpha}_0 > 0.75$ pu, however it greatly exceeds the acceptable limit (dashed line).

## 5.2.6 Reducing Harmonic Distortion

The previous section showed that the inverter output current is unable to meet the grid THD requirement, without some form of harmonic reduction. It is anticipated that the low-pass filter will sufficiently attenuate the PWM switching harmonics, however, the low-frequency harmonics, caused by the fluctuating input power, will be difficult to eliminate, as they occur in the vicinity of the fundamental. In addition, the medium-frequency
harmonics may be difficult to eliminate, as the turbine operates at various speeds, which causes the harmonic frequencies to also vary. This section briefly discusses several techniques to attenuate / eliminate harmonics, these include using: low-pass and notch filters, reducing  $\hat{\alpha}$ , and using a feed-froward controller.

#### Low-Pass Filter

The low-pass filter is extensively used in electronic and communication systems, to attenuate high frequency components whilst allowing low frequencies through. For this application, a second-order CL type low-pass filter is required as this sufficiently couples the current source inverter to the voltage source grid, according to the *impedance mismatch criteria* [105]. An example of this type of filter is shown in Figure 5.25, where  $R_d$ is the damping resistor.



Figure 5.25: CL type low-pass filter, as used to couple the current source inverter to the grid.  $R_d$  represents the damping resistor used to control the filter quality factor.

The filter cutoff frequency is commonly referred to as the resonant frequency,  $f_{res}$ , as the filter has an infinite gain at that frequency if there is no damping resistance. The gain at the resonant frequency is known as the quality factor, Q, and can be reduced by using a series damping resistor,  $R_d$  (seen in the above figure). The resonant frequency and quality factor are given by Equations (5.21) and (5.22), respectively, whilst Figure 5.26 shows the effect that varying the damping resistance has on the quality factor. Note that the filter gain represents the ratio of output to input current, i.e.  $I_{out} / I_{in}$ , and that the gain is presented in decibels (dB).

$$f_{res} = \frac{1}{2\pi\sqrt{LC}}\tag{5.21}$$



 $Q = \frac{1}{R_d} \sqrt{\frac{L}{C}} \tag{5.22}$ 

Figure 5.26: Normalised low-pass filter frequency response, showing the effect of varying the filter quality factor, Q.

The CL filter is able to substantially attenuate the high-frequency PWM harmonics if the resonant frequency is selected to be small compared to the PWM switching frequency. Despite this, there are two drawbacks of using this type of low-pass filter. Firstly, the harmonics will be amplified if they occur at the resonant frequency and if the quality factor exceeds 1. The filter will have a fixed resonant frequency and may correspond to the frequency of the rectifier ripple, as the turbine operates over a wide range of speeds. This may amplify the harmonics and increase the output current THD. Secondly, the lowpass filter is unable to attenuate the low-frequency harmonics caused by the fluctuating input power, as these occur close to the fundamental frequency.

### Notch Filters

Notch filters are used to eliminate specific frequencies, whilst passing all others. They comprise of a series LC resonant network in a voltage divider configuration [109]. An example of a notch filter and its position, relative to the proposed grid-connected inverter,

is shown in Figure 5.27. The figure includes a series connected damping resistor,  $R_d$ , which is used to control the filter quality factor. The effect of varying the quality factor on the normalised filter gain is shown in Figure 5.28.



Figure 5.27: Notch filter (left), and its position in the proposed grid-connected inverter system (right).



Figure 5.28: Normalised notch filter frequency response, showing the effect of increasing the quality factor, Q.

The above figure shows that the filter shunt impedance approaches zero as the filter input frequency approaches the resonant frequency. This provides a path to ground, which diverts the input current through the LC network and hence creates the *notch* in the filter frequency response. The notch filter resonant frequency and quality factor are identical to those for the above low-pass filter, see Equations (5.21) and (5.22).

Although notch filters can be used to eliminate fixed frequency harmonics, e.g. the third and fifth-order harmonics of the fundamental, they can not be used to eliminate the harmonics caused by the rectifier ripple, as this harmonic frequency varies with the generator speed. The main drawback of notch filters is that they can potentially become grid harmonic sinks. That is, if tuned to the grid frequency harmonics, notch filters will draw large harmonic currents from the grid, which is undesirable. The Australian Standards also state that the inverter should not significantly sink frequencies used for ripple control (which allows remote grid load optimisation [110]) by the local electrical distributor [96]. Notch filters also attract cost, and hence, are no longer considered.

### Reduction of $\widehat{\alpha}$

Section 5.2.1 showed that low frequency harmonics, i.e.  $h_{3-7}$ , can be reduced by reducing  $\hat{\alpha}$ . This is achieved by reducing either the  $m_a$  or  $\hat{\alpha}_0$ . The modulation index is easier to adjust, as this is handled by the controller. The drawback is that the output current magnitude is linearly proportional to the  $m_a$ , i.e. the output current distortion improves at the expense of the inverter output current magnitude. Hence a distortion / power trade-off exists.

In contrast, the reduction of  $\alpha_0$  does not affect the magnitude of the output current, however, this is more difficult to achieve in practise, as this implies that the generator open-circuit voltage must increase. This can be achieved by:

- increasing the machine speed It was seen in part I that the turbine speed may be controlled by adjusting the electromagnetic torque, however, to increase its speed the turbine may not operate at its optimal  $c_p$  which will reduce the turbine and hence the inverter output power.
- modifying the machine characteristics The back-EMF is dictated by the electrical frequency,  $\omega_e$ , and the back-EMF constant, k, hence increasing either k or the number of pole-pairs will reduce  $\hat{\alpha}_0$ . The drawback is that the machine must be physically altered for such changes to take effect, e.g. the permanent magnets may be replaced those with a higher energy density.

Although the reduction of  $\hat{\alpha}$  is required to minimise the input current ripple, the drawback of this is that the PM generator operates more in the constant current region of its I-V locus. The higher currents ( $\beta \rightarrow 1$ ) cause higher stator copper losses, and hence reduce the generator efficiency.

### Advanced Control

The inverter is presently controlled in an open-loop manner, and hence the output current contains the harmonics found in the input current. The use of a more sophisticated control system, such as a feedback or feed-forward controller would improve the output current waveform. The feedback controller would adjust the current-wave shaper duty-cycle such that the output current would be sinusoidal. This type of controller, however, requires an expensive isolated output current sensor. In contrast, a feed-forward controller would adjust the wave-shaper duty-cycle, based on the sampled input current and the known simple relationship between the duty-cycle and input and output currents. This attracts a lower cost, as the current sensor is not required to be isolated, however, both systems require a fast processor to rapidly sample their respective currents.

In summary, a low-pass filter and a feed-forward controller should allow the gridconnected inverter to deliver high quality power to the grid. These are further discussed in Sections 5.3 and 5.4, respectively.

# 5.3 Low-Pass Filter Analysis

The low-pass filter is situated between the unfolding circuit and the grid. Its primary role is to attenuate the high-frequency PWM current harmonics, such that the output current THD meets the grid requirements, i.e. THD < 5%. In addition, the choice of filter component can also affect the inverter power factor. This is examined in Section 5.3.6, which emphasises that the inverter power factor must lie between 0.8 leading and 0.95 lagging, for inverter outputs between 20% and 100% of rated volt-amperes.

It was previously mentioned in Section 4.2.5 that the low-pass filter configuration is determined by applying the filter *impedance mismatch criteria* to the inverter input type, i.e. current or voltage source. Voltage source inverters (VSI) make up the majority of commercially available inverters, and use either a single inductive (L) or inductivecapacitive-inductive (LCL) type low-pass filter. The L filter is the simplest and most common filter type amongst VSI [61], and is well suited to high-frequency PWM converters [111]. The drawback, however, is that L type filters attenuate high-frequencies at only 20dB/decade, and so its use is restricted in high-powered low-frequency converter applications [111, 112]. The LCL filter addresses these problems, as it attenuates frequencies at 60dB/decade [61], beyond its resonant frequency, and it so achieves reduced levels of harmonic distortion at lower switching frequencies and with less total inductance [111,112] and hence less losses. In contrast, a capacitive-inductive (CL) type filter is used with current source inverters (CSI), and attenuates frequencies at 40dB/decade, beyond its resonant frequency. A summary of these low-pass filter configurations is shown in Figure 5.29.



Figure 5.29: Typical grid-connected inverter low-pass filter varieties, showing the (a) L, (b) LCL and (c) CL types used. The L and LCL filters are used to couple voltage source inverters to the grid, whilst the CL type couples a current source inverter to the grid.

The main drawback of the LCL and CL type filters is resonance, which can potentially increase harmonics if the filter gain at the resonant frequency is greater than 1 and the resonant frequency,  $f_{res}$ , coincides with any low or medium-frequency harmonics. This was seen in the experimental testing stage (Section 4.4.3) where the filter resonated at 450Hz, i.e. the ninth-order harmonic. The effect of resonance can be reduced by limiting the filter gain at the resonant frequency. This, however, requires an understanding of the filter and how it responds to various input frequencies.

#### 5.3.1 Filter Response

The filter response is determined by its transfer function, H(s), which represents the complex ratio of the frequency dependent output to input, where  $s = j\omega$ . Note that H(s) represents the ratio of the output to input voltage, for an L or LCL filter, and the ratio of output to input current, for a CL type filter. An example of an undamped CL low-pass filter transfer function is shown in Equation (5.23).

$$H(s) = \frac{1}{s^2 L C + 1} \tag{5.23}$$

The filter gain, A, is the absolute value of the transfer function, and the phase delay,  $\phi$ , is given by the argument of the transfer function, as summarised by Equations (5.24), and (5.25), respectively.

$$A = |H(s)| \tag{5.24}$$

$$\phi = \arg\left(H(s)\right) \tag{5.25}$$

The gain at the resonant frequency determines the extent of resonance and is commonly referred to as the quality factor, Q. Consider the undamped CL filter. The filter has an infinite quality factor, when the denominator of Equation (5.23) is zero. The frequency corresponding to this case, is known as the resonant frequency,  $f_{res}$ , and is given by Equation (5.26). The infinite gain would amplify harmonics if found at the resonant frequency, and hence increase the filter output THD. The need to limit the filter quality factor is clearly seen. In practise, this is achieved by adding damping to the filter, and is discussed in the following section. Note that  $f_{res}$  is also referred to as the cutoff frequency,  $f_c$ , and that both terms are used throughout this chapter.

$$f_{res} = \frac{1}{2\pi\sqrt{LC}}\tag{5.26}$$

# 5.3.2 Filter Damping

The CL low-pass filter can be damped by introducing a damping resistor,  $R_d$ , to the circuit. The damping resistor can be connected in one of four ways, as listed below and summarised in Figure 5.30. Note that configuration 4 was previously seen in Section 5.2.6.

- 1) connected in series with the capacitor,
- 2) connected in parallel to the capacitor,
- 3) connected in parallel to the inductor, or
- 4) connected in series with the inductor.



Figure 5.30: Damped CL low-pass filter, showing four configurations.

The quality factor is a function of the filter characteristic impedance,  $Z_0$ , the damping resistance and the filter configuration;  $Z_0$  is expressed by Equation (5.27). The filter is said to be *over*, *critically* and *under*-damped if Q is less than, equal to and greater than 0.71  $(1/\sqrt{2})$ , respectively. Although a quality factor of less than 1 prevents harmonic amplification, it is generally acceptable for a switched-mode power supply filter to be designed with a Q between 2 and 4 [105].

$$Z_0 = \sqrt{\frac{L}{C}} \tag{5.27}$$

# 5.3.3 Damped Filter Response and Configuration Comparison

Examples of damped low-pass filter responses are shown in Figure 5.31, for each of the four damped filter configurations (FC) 1–4. Figures 5.31(a) and 5.31(b), show the gain and phase delay as a function of input filter frequency, for damping resistances equal to  $Z_0 \Omega$  and  $2 Z_0 \Omega$ , respectively. Note that the filter input frequency is shown on a logarithmic scale, ranges from 0.01 to 100pu and is normalised relative to the resonant frequency.

The gain and phase delay plots of FC 1 and FC 3 are identical for a damping resistance equal to the characteristic impedance, as are those of FC 2 and FC 4 (Figure 5.31(a)). As the damping resistance increases, the gain and phase delay plots become unique (see Figure 5.31(b)). In general, FC 1 and FC 3 behave similarly, as do FC 2 and FC 4. Generally, the gain of an  $n^{th}$ -order low-pass filter rolls-off at  $-20 n \, dB/decade$  and the phase delay approaches  $-90 n^{\circ}$  at high frequencies. This indicates that FC 1 and FC 3 behave as firstorder low-pass filters, whilst FC 2 and FC 4 act like second-order low-pass filters. This implies that FC 2 and FC 4 will yield a cleaner (lower THD) output current than FC 1 and FC 3.



Figure 5.31: Normalised CL low-pass filter gain and phase margin comparison, for (a)  $R_d = Z_0 \Omega$  and (b)  $R_d = 2 Z_0 \Omega$ . Filter configurations (FC) 1 and 3, behave as first-order filters, whilst FC 2 and 4 behave like second-order filters. Note that the frequency is normalised relative to the resonant frequency,  $f_{res}$ .

The above figure showed that the filter quality factor and phase delay at the resonant frequency, varies with damping resistance and the selected filter configuration. This is further examined and verified in Figure 5.32, for various values of damping resistance, however, the phase delay of FC 2 and FC 4 is independent of the ratio of  $R_d$  to  $Z_0$ . This information and other key filter features is summarised in Table 5.2. Note that the damping resistance is normalised, i.e. it is expressed relative to the characteristic impedance.



Figure 5.32: Filter gain (a) and phase delay (b), at the resonant frequency,  $f_{res}$ , of each filter configuration, as a function of damping resistance,  $R_d$ . Note that the damping resistance is normalised, i.e. it is expressed relative to the characteristic impedance,  $Z_0$ .

#### Filter Induced Phase Delay

Although the above figures and table shows that the low-pass filter introduces a phase delay, it is only the phase delay at the fundamental frequency ( $\phi$  at  $f_1$ ) that is of significance, as this can affect the inverter power factor. Providing the filter Q is greater than 1 and the resonant frequency is significantly greater than the fundamental frequency, the  $\phi$ at  $f_1$  is small and hence negligible. This is shown in Figure 5.33, where the filter delay (at the fundamental frequency) is plotted as a function of Q, for resonant frequencies equal to  $3 f_1$ ,  $5 f_1$  and  $10 f_1$ . It is seen that the filter delay decreases as Q increases, and as the  $f_{res}$  increases.

Recall that FC 1 and FC 3 have quality factors greater than one, for all values of  $R_d$ . This information, together with the knowledge of low-frequency harmonics,  $h_{3-7}$  (from Section 5.2.1), implies that the  $f_{res}$  should be designed to be at least 9 times the  $f_1$ , to

FC	Circuit	Transfer	Quality	Phase delay	Roll-off rate
		function, $H(s)$	Factor, $Q$	at $f_{res}$ (°)	beyond $f_{res}$
1		$\frac{sCR_d+1}{s^2CL+sCR_d+1}$	$\sqrt{\left(\frac{Z_0}{R_d}\right)^2 + 1}$	$90 - \tan^{-1}\left(\frac{R_d}{Z_0}\right)$	-20dB/decade
2		$\frac{R_d}{s^2 CLR_d + sL + R_d}$	$rac{R_d}{Z_0}$	-90	-40dB/decade
3		$\frac{sL+R_d}{s^2CLR_d+sL+R_d}$	$\sqrt{\left(\frac{R_d}{Z_0}\right)^2 + 1}$	$-\tan^{-1}\left(rac{R_d}{Z_0} ight)$	-20dB/decade
4		$\frac{1}{s^2 CL + s CR_d + 1}$	$rac{Z_0}{R_d}$	-90	-40dB/decade

Table 5.2: Comparison of the key low-pass filter properties, for each of the four damped filter configurations (FC).



Figure 5.33: Filter delay at the fundamental frequency ( $\phi$  at  $f_1$ ) vs. quality factor for filter configurations (a) 1 and 3, and (b) 2 and 4. The curves represent cutoff frequencies  $(f_c)$  equal to 3, 5 and 10 times that of the fundamental.

avoid low-frequency harmonic amplification (as Q > 1). Therefore, as a general guideline, the filter resonant frequency should be selected to be at least one decade above the fundamental, when using the PM generator current source. Using this recommendation, the phase delay at  $f_1$  for FC1 and FC3 will be very small, i.e.  $< 0.25^{\circ}$ , and is hence negligible. In contrast, the  $\phi$  at  $f_1$  caused by FC2 and FC4 will be between 3° and 1° for a Q between 2 and 4, respectively. Despite this, the phase delay of FC2 and FC4 is also considered insignificant, i.e. from this point on, it is assumed that the fundamental of the filtered unipolar PWM waveform is in phase with the grid voltage.

# 5.3.4 Design Considerations

It is well known that a GCI is required to provide high quality (low THD) power to the grid, whilst meeting the necessary power factor. Despite this, the importance of the low-pass filter design is often overlooked. The low-pass filter design influences the harmonic attenuation and affects the inverter power factor. In addition, the filter will have copper losses and hence reduce the overall inverter efficiency. To maintain a high efficiency, a maximum damping resistor power loss,  $P_d$ , of 5% rated power is arbitrarily selected. This should also be taken into account when designing a GCI low-pass filter.

# Filter Variables

The filter itself has four variables, three of which are circuit parameters, i.e. the capacitance, the inductance and the damping resistance; the remaining variable is the filter configuration. A summary of these filter variables, and the key filter parameters they affect, is summarised in Table 5.3.

Parameter	Resonant	Quality	Harmonic	Damping Resistor	
Variable	Frequency, $f_{res}$	Factor, $Q$	Attenuation	Power Loss, $P_d$	
Inductance, $L$	✓	$\checkmark$	1	$\checkmark$	
Capacitance, $C$	✓	1	1	✓	
Damping	×				
Resistance, $R_d$	~	v	v	v	
Filter	×				
Configuration (FC)	~	v	v	<b>▼</b>	

Table 5.3: Low-Pass filter variables, and the key parameters they affect.

#### **Frequency and Filter Component Normalisation**

The filter cutoff (or resonant) frequency and filter components are normalised, as this greatly simplifies the analysis and allows the designer to easily select component values for any sized (rated) system. The normalised cutoff frequency,  $\omega_{cn}$  is expressed relative to the base frequency,  $\omega_B$ , these are expressed by Equations (5.28) and (5.29), respectively, where  $f_1$  represents the grid (inverter fundamental) frequency.

$$\omega_{cn} \equiv \frac{\omega}{\omega_B} \tag{5.28}$$

$$\omega_B = 2 \pi f_1 \tag{5.29}$$

Consider an inverter designed to deliver rated power,  $P_B$ , in to a grid of rated voltage  $V_B$ . The resulting base impedance,  $Z_B$ , is expressed by Equation (5.30), and is later used to normalise the filter components. Note that the base impedance is defined as the quotient of the base voltage and base current,  $I_B$ .

$$Z_B \equiv \frac{V_B}{I_B} = \frac{V_B^2}{P_B} \tag{5.30}$$

The normalised filter capacitance,  $C_n$ , is shown in Equation (5.31), where  $C_B$  represents the base capacitance, which itself is expressed in Equation (5.32).

$$C_n = \frac{C}{C_B} \tag{5.31}$$

$$C_B = \frac{1}{\omega_B Z_B} \tag{5.32}$$

Similarly, the normalised filter inductance,  $L_n$ , is expressed relative to the base inductance,  $L_B$ ; these are summarised in Equations (5.33) and (5.34), respectively.

$$L_n = \frac{L}{L_B} \tag{5.33}$$

$$L_B = \frac{Z_B}{\omega_B} \tag{5.34}$$

The filter resonant frequency, which was previously shown in Equation (5.21), is also normalised and simplified. Substitution of  $C_n$  and  $L_n$ , in to this expression, shows that the product is inversely proportional to the square of filter resonant frequency; this is summarised by Equation (5.35) and Figure 5.34(a), which shows  $\omega_{cn}$  contours of 1, 5, 10 and 20pu. Similarly, Figure 5.34(b) shows normalised characteristic impedance,  $Z_{0n}$ , contours of 0.5, 1, 1.5, 2 and 2.5pu;  $Z_{0n}$  is expressed by Equation (5.36).

$$L_n C_n = \frac{1}{\omega_{cn}^2} \tag{5.35}$$



$$Z_{0n} = \sqrt{\frac{L_n}{C_n}} = \frac{Z_0}{Z_B} \tag{5.36}$$

Figure 5.34: Normalised capacitance,  $C_n$ , vs. inductance,  $L_n$ , plots, showing (a) cutoff frequency,  $\omega_{cn}$ , and (b) characteristic impedance, , contours. The  $\omega_{cn}$  contours are equal to 2, 5, 10 and 20pu, whilst those for  $Z_{0n}$  are equal to 0.5, 1, 1.5 and 2pu.

#### **Practical Limitations**

Although Figure 5.34 suggests that a low-pass filter can be easily designed, based on a desired cutoff frequency or characteristic impedance, it should be noted that components may face practical limitations, such as size, cost, weight etc. For example, consider a 1kW GCI: an inductance of 0.6pu corresponds to approximately 110mH. An inductor this size for this application would be heavy, bulky and costly, due to the inverter current rating (4.2A). Hence, the size, cost and weight of the filter (and hence inverter) can be reduced

by selecting a smaller value of  $L_n$ ; conversely, this would require a larger capacitance for a given cutoff frequency.

A small capacitance and small inductance can be selected, however, this produces a high cutoff frequency, which requires the inverter to operate at an even higher switching frequency,  $f_{sw}$ , to allow the filter to remove the PWM harmonics. This leads to higher inverter switching losses,  $P_{SW}$ , as  $P_{SW}$  is proportional to the  $f_{sw}$ . Hence upper switching and cutoff frequency limits are required. For a 50Hz GCI, a maximum switching frequency of 10kHz is selected (200pu). The upper cutoff frequency depends on the filter configuration, required harmonic attenuation, and the inverter switching frequency. This further investigated in Section 5.3.5, whilst Sections 5.3.6 and 5.3.7 examine the power factor requirements and damping resistor power loss, respectively. Section 5.3.8 summarises with a power factor, power loss and output current THD trade-off.

### 5.3.5 Harmonic Attenuation and Distortion

This section examines the filter's ability to attenuate harmonic components and hence reduce the THD of an unipolar PWM current. Consider Figure 5.35, it shows the harmonic spectrum of a 4kHz unipolar PWM waveform, and filter gain of FC1 - 4. The cutoff frequency,  $f_c$ , and quality factor are arbitrarily selected as 1kHz and 2, respectively. Note that the filter gain of FC1 and 3 are indistinguishable, as such, the resulting THD of these filters will be identical. Similarly, the gain and hence THD of FC2 and 4 are identical.



Figure 5.35: Harmonic spectrum of a unipolar PWM waveform, and the gain of filter configurations 1 and 3, and 2 and 4, which represent first and second-order filter responses, respectively. Each filter gain is shown for a quality factor of 2 and a cutoff frequency of 1kHz. Note that the harmonic spectrum is shown for a switching frequency of 4kHz.

Figure 5.35 indicates that filter configurations 2 and 4 offer a greater rate of attenuation than FC 1 and 3. This is expected as FC 1 and 3 behave as first-order filters, i.e. their gains roll-off at -20 dB/decade, whilst the gains of FC 2 and 4 roll-off at -40 dB/decade, as they are second-order low-pass filters. In addition to order, other aspects such as quality factor and cutoff frequency also affect the filter gain and hence output current THD. The effect of varying the normalised cutoff frequency,  $f_c$ , on the filter gain is shown in Figure 5.36 for a Q of 2, whilst Figure 5.37 shows the effect of varying the quality factor, for a fixed  $f_c$  of 800Hz. Note that the PWM switching frequency is maintained at 4kHz.



Figure 5.36: Filter gain on a 4kHz unipolar PWM harmonic spectrum, showing the effect of varying the filter cutoff frequency, for filter configurations (a) 1 and 3, and (b) 2 and 4. The cutoff frequencies are normalised relative to the switching frequency and are shown equal to 0.1, 0.2, 0.5 and 1pu. Note that the PWM harmonics are now only shown for only two multiples of the switching frequency.

Note that the normalised cutoff frequency,  $\omega_{cn}$ , seen in Figure 5.34(a), is expressed relative to the fundamental frequency,  $f_1$ , whilst the normalised cutoff frequency,  $f_{cn}$ , shown in Figures 5.36 and 5.37 are expressed relative to the inverter switching frequency,  $f_{sw}$ .

Figure 5.36 implies that the low-pass filters becomes less effective as  $f_c$  approaches  $f_{sw}$ , as higher magnitude high-frequency components pass through the filter, which ultimately increases the output current THD. In addition, Figure 5.37 suggests the quality factor of FC 1 and 3 also affect the harmonic attenuation and hence the resulting THD. For



Figure 5.37: Filter gain on a 4kHz unipolar PWM harmonic spectrum, showing the effect of varying the filter quality factor, for filter configurations (a) 1 and 3, and (b) 2 and 4. Quality factors, of 1, 2 and 5 are shown for each filter configuration; in addition the critically damped case,  $Q = 0.71 \left(\frac{1}{\sqrt{2}}\right)$ , is shown in (b). Note that the cutoff frequency is 800Hz (0.2pu, relative to the switching frequency).

example a low Q ( $\approx 1$ ) provides little attenuation at high frequencies, whilst a higher value Q (e.g. 5) significantly improves the high-frequency attenuation. In contrast, the Qdoes not affect the output current THD for FC 2 and 4, at low cutoff frequencies, however, it becomes significant as  $f_c$  approaches  $f_{sw}$ .

The above statements are verified in Figure 5.38, which shows the calculated output current THD for each filter, as a function of both cutoff frequency and quality factor; the dashed line represents the grid THD limit of 5%. The figure indicates that FC1 and 3 are effective if the Q is above 2 and the  $f_c$  is between less than about 0.2 to 0.3pu; this may explain why Nave suggests a Q between 2 and 4 [105]. In addition, the figure shows that the THD of FC2 and 4 is independent of Q, for  $f_{cn}$  less than about 0.3pu, and that each filter configuration is able to meet the grid THD requirement for a Q of 5, and a  $f_{cn}$ less than about 0.3pu. Hence, the cutoff frequency is limited to about 0.3pu, such that the filtered current meets the grid THD requirement.

In summary, filter configurations 2 and 4 are able to meet the grid THD specifications over a wider range of Q and  $f_c$ , than FC1 and 3. This knowledge makes the design of the second-order low-pass filters more flexible in terms of cutoff frequency and quality factor, however, the power factor and damping resistor loss should also be considered before selecting a filter configuration.



Figure 5.38: Output current THD vs. filter cutoff frequency and quality factor, for filter configurations (a) 1 and 3, and (b) 2 and 4. The curves represent quality factors, Q, of 1, 2 and 5, for filter configurations 1–4. Note that filter configurations 2 and 4 also include the critically damped case, i.e. a Q of 0.71  $\left(\frac{1}{\sqrt{2}}\right)$ .

# 5.3.6 Power Factor Requirements

This section shows that the inverter power factor is affected by the choice of filter components, and provides a guideline to selecting the filter inductance and capacitance, such that the power factor requirements are met. The Australian Standards consider the inverter as a load from the grid, and requires the inverter to operate between 0.8 lead and 0.95 lag for all output apparent powers between 20 and 100% rated (0.2 to 1 pu) power [96]. A summary of the normalised apparent, real and reactive powers (S, P, and Q, respectively), under these conditions is shown in Table 5.4 and Figure 5.39. Note that it is assumed that the fundamental of the inverter output current is in-phase with the grid voltage.

Table 5.4: Summary of normalised apparent, real and reactive power, and power factor angle  $(\phi)$  under grid power factor requirement extremes.

Power Grid Extremes	Apparent $(S)$	Real $(P)$	Reactive $(Q)$	$\phi$
Power Factor $= 0.80$ lead	$0.20\mathrm{pu}$	0.16 pu	$-0.12\mathrm{pu}$	36.9°
Power Factor = $0.95 \text{ lag}$	1.00 pu	$0.95\mathrm{pu}$	$0.31\mathrm{pu}$	$-18.2^{\circ}$



Figure 5.39: Real, P, reactive, Q, and apparent power, S, triangle, corresponding to the grid power factor limits, showing (a) 0.8 lead and (b) 0.95 lag.

The reactive power, Q, in the above figure represents the net reactive power that is either delivered to or absorbed by the grid. It is the difference between that which is supplied by the capacitor,  $Q_C$ , and that which is absorbed by the inductor,  $Q_L$ , and is hence expressed as  $Q_L - Q_C$ . This reactive power flow, is summarised for an undamped CL low-pass filter in Figure 5.40, which also shows the real power flow, i.e. from the inverter to the grid. Note that the inverter does not produce reactive power as the fundamental output current is assumed to operate at unity power factor, i.e. it is supplied by the capacitor and is absorbed by the inductor and grid.



Figure 5.40: Real, P, and reactive, Q, power flow diagram for an undamped CL low-pass filter. Note that the grid supplies / absorbs reactive power that is equal to  $Q_L - Q_C$ , and that real power is not absorbed in the filter.

The introduction of a damping resistor would simply imply that the net real power, that is fed to the grid, is equal to  $P - P_d$ , where  $P_d$  is the damping resistor power loss. As  $P_d$  is set to 0.05pu (5% rated power) at rated current, it is foreseen that a small reduction in real power will not greatly affect the resulting power factor. For example, consider the 0.95 lagging case: if P is reduced from 0.95 to 0.9pu, whilst Q is maintained at 0.31 pu, the power factor changes from 0.95 lag to 0.945 lag.

#### **Reactive Power**

The reactive power that is absorbed by the inductor, and generated by the capacitor is determined analytically here. Both  $Q_L$  and  $Q_C$  are expressed as a normalised quantities. The normalised inverter output current is assumed be equal to 1pu and 0.2pu, for rated and 20% rated volt-amperes operation, respectively.

The inductor absorbs reactive power,  $Q_L$ , which is proportional to the square of the normalised current  $(I_n)$  and its reactance,  $X_L$ . This is summarised by Equation (5.37), and is simplified to Equation (5.38). Hence, at rated current,  $Q_L$  is equal to  $L_n$  pu, whilst at 20% rated power, it is equal to  $0.04 L_n$  pu.

$$Q_L = I_n^2 X_L \tag{5.37}$$

$$= I_n^2 L_n \quad (pu) \tag{5.38}$$

The capacitor generates reactive power,  $Q_C$ , which is proportional to the square of the capacitor voltage,  $V_C$ , and inversely proportional to the capacitive reactance,  $X_C$ . This is summarised and simplified in Equations (5.39) and (5.40), respectively. Note:  $1/X_C$  simplifies to  $C_n$ , due to the grid frequency.

$$Q_C = \frac{V_C^2}{X_C} \tag{5.39}$$

$$= V_{Cn}^2 C_n$$
 (pu) (5.40)

The capacitor voltage is the phasor summation of the grid voltage,  $V_g$ , and the voltage drop across the inductor,  $V_L$ , as summarised by Equation (5.41). This simplifies to Equation (5.42), as  $V_g$  is assumed to be equal to 1 pu, and  $V_L$  is equal to  $j I_n L_n$  ( $\omega_n = 1$ pu).

$$V_C = V_g - V_L \tag{5.41}$$

$$= 1 - j I_n L_n$$
 (pu) (5.42)

#### **Filter Component Limitations**

The maximum value of  $C_n$  and  $L_n$  can be determined, such that the power factor requirements are met. Consider the <u>leading</u> power factor case, which allows the maximum value of  $C_n$  to be determined. This power factor is most difficult to meet when the inverter operates at 20% rated volt-amperes. Under this condition, the inverter output current is 0.2pu and Q is equal to -0.12pu. As Q is equal to  $Q_L - Q_C$ , the maximum value of  $C_n$  can be calculated as a function of  $L_n$ . Hence, for zero inductance, a normalised capacitance of less than 0.12pu is required to meet the power factor standard. This maximum capacitance value is expected to increase with increasing  $L_n$ , however, only slightly as  $Q_L$  is equal to  $0.04 L_n$  pu (for 20% rated operation). For example, the leading power factor requirement will be met for an inductance of 0.5pu, providing  $C_n$  does not exceed 0.124 pu.

This process is repeated for the <u>lagging</u> power factor case, however, the power factor is set to 0.95 (lag) and the inverter output current is set to 1pu. The normalised capacitance is again determined as a function of  $L_n$ , such that the power factor is met. In the absence of a capacitor, the maximum inductance is equal to 0.31pu. The introduction of capacitance will supply reactive power to the inductor and effectively allow  $L_n$  to increase. Compared to the effect of  $L_n$  on  $C_n$  in the light-load leading power factor case, the effect of  $C_n$  on  $L_n$  under full-load conditions is more significant as the  $V_C$  is greater than 1pu.

A summary of the inductance and capacitance limits, that allow the filter to meet the lagging and leading grid power factor requirements, respectively, is shown by the dashed lines in Figure 5.41. The figure also shows  $\omega_{cn}$  contours of 10, 26, and 65pu, which together with the dashed lines, form the boundary of two design regions; these are shaded for convenience. The smaller region, area A, corresponds to a switching frequency of 4kHz, whilst the larger region, areas A and B, represents the design region for a  $f_{sw}$  of 10kHz. Hence, filter components selected within a specified design region (based on the  $f_{sw}$ ) should yield a sinusoidal output current whose fundamental is within the required power factor range. Note that a minimum  $\omega_{cn}$  of 10pu is selected as this produces a small and insignificant filter phase delay (recall Figure 5.33). Maximum  $\omega_{cn}$  of 26 and 65pu are selected as these correspond to  $f_{cn}$  of about 0.33pu, i.e. the output current waveform will contain less than 5% THD, for switching frequencies of 4 and 10kHz, respectively.

Although the inverter power factor requirements are met by selecting values of  $C_n$  and  $L_n$  within the design region, other aspects such as the filter characteristic impedance, must also be considered. The  $Z_{0n}$  is governed by  $C_n$  and  $L_n$  (recall Equation (5.36)), and also



Figure 5.41: Normalised capacitance vs. inductance, showing various cutoff frequencies and characteristic impedances. The shaded area represents the design region, which limits the inverter switching losses, and meets the required power factor standards.

affects the choice of  $R_d$  and hence the quality factor. The filter power loss, caused by  $R_d$ , is discussed in the following section, whilst the effect of reducing  $C_n$  on both the output current THD and power loss is examined in Section 5.3.8, for all filter configurations.

# 5.3.7 Power Loss

Section 5.3.5 showed that each filter is able to meet the grid THD requirement providing the quality factor is greater than 2 and the cutoff frequency was less than about 0.2pu. The issue is that as Q increases, so too does  $R_d$  for a fixed  $Z_0$ . Intuitively, the power loss associated with the damping resistance,  $P_d$ , should increase with  $R_d$  for FC1 and 4, as these are series connected resistive-inductive-capacitive (RLC) circuits. In contrast,  $P_d$ should decrease for FC2 and 3, as these are parallel damped RLC circuits.

Analytically, the damping resistance power loss is proportional to  $R_d$  and the square of the current that flows through it,  $I_d$ , as shown in Equation (5.43). The total current that flows through  $R_d$  is determined by the theory of *superposition*, i.e. the summation of the currents caused by each source, whilst ignoring the other. The damping resistor current from the inverter,  $I_f$ , is determined by shorting the grid, as it is a voltage source. In contrast, the grid drawn current,  $I_g$ , is calculated by removing the inverter, as it is a current source. This process is depicted in Figure 5.42, for FC 3, and is summarised by Equation (5.44).

$$P_d = I_d^2 R_d \tag{5.43}$$

$$|I_d| = |I_f + I_g| (5.44)$$



(a) Current from inverter,  $I_f$ 

(b) Current from grid,  $I_g$ 

Figure 5.42: Damping resistor current determined by the theory of *superposition*, showing damping resistor current from (a) the inverter,  $I_f$ , and (b) the grid,  $I_g$ . The theory considers the current from each source, whilst ignoring the other. Note that current and voltage sources are replaced with open and short-circuits, respectively.

The current divider rule is used to determine the inverter associated current,  $I_f$ . A summary of these currents, along with those caused from the grid,  $I_g$ , are shown in Table 5.5, as a function of input frequency ( $s = j \omega$ ). The table shows that  $I_f$  of FC 2 and 3 are identical, as are  $I_g$  of FC 1 and 4. Note that the power loss calculation (see Equation (5.43)) uses the RMS value of  $I_f$ , which includes the filtered high-frequency PWM harmonic components up to  $7 f_{sw}$ , which contains 118 harmonic sidebands. In contrast, the grid drawn current considers only the fundamental (grid) frequency, i.e.  $\omega = \omega_B$ .

#### Power Loss Comparison

The power loss of each filter configuration can be determined by examining each circuit of Table 5.5. Consider FC1. The high-frequency PWM harmonics will pass through the capacitor and  $R_d$  as the capacitor is essentially a short-circuit at high-frequencies. Hence, the high-frequency current harmonics and the subsequent high power loss will be Table 5.5: Comparison of damping resistor current, for various low-pass filter configurations. The total current is determined from the principle of superposition, i.e it is the summation of the inverter and grid drawn currents,  $I_{inv}$  and  $I_g$ , respectively.

FC	Circuit	$I_f \ (\omega \ge \omega_B)$	$I_g \ (\omega = \omega_B)$
1		$I_{in} \frac{s^2 CL}{s^2 CL + s CR_d + 1}$	$\frac{V_g  sC}{s^2 CL + sCR_d + 1}$
2		$I_{in} \frac{sL}{s^2 CLR_d + sL + R_d}$	$\frac{V_g}{s^2 CLR_d + sC + R_d}$
3		$I_{in} \frac{sL}{s^2 CLR_d + sL + R_d}$	$\frac{V_g  s^2 C L}{s^2 C L R_d + s L + R_d}$
4		$I_{in} \frac{1}{s^2 CL + s CR_d + 1}$	$\frac{V_g  sC}{s^2 CL + sCR_d + 1}$

dissipated in  $R_d$ . In contrast, the grid-drawn current will be small due to the capacitance limitation, and the resulting power loss will hence be negligible.

Consider FC 2. The high-frequency PWM harmonics currents are assumed to pass through the capacitor. This implies the fundamental current caused by the inverter will pass through  $R_d$ , if it is smaller than the inductive reactance. The grid-drawn current will also be large for small  $R_d$ , as there is a small and insignificant voltage drop across the inductor, i.e. the voltage drop across the resistor is essentially that of the grid. The resistance must hence must be large to limit  $P_d$ . The drawback, however, is that Q is directly proportional to  $R_d$ , i.e. a power loss vs. quality factor trade-off exists.

Consider FC 4. The capacitor again shorts the high-frequency components, implying that the fundamental inverter output current passes through  $R_d$ . The resulting inverter associated power loss is hence proportional to  $R_d$ . The grid-drawn current is once again limited by the capacitance and the resulting power loss is considered negligible.

Consider FC 3. The series-connected capacitor again limits the grid-drawn current to a small amount and hence a negligible power loss. The inverter associated current does not contain high-frequency harmonics and is small predividing  $R_d$  is comparable to (or smaller than) the inductive reactance. Such a case would result in a small Q and hence a large output current THD (recall Figure 5.38(a)). In contrast, a large  $R_d$  will increase Q, which will sufficiently attenuate the high-frequency PWM harmonics, whilst reducing the damping resistor power loss.

The above information suggests that FC 2 is the least favoured filter configuration, due to high power losses. Nave verifies this by stating that power dissipation problems may render this filter impractical [105]. A similar study by Ahmed et al. compares the simulated damping resistor power loss of several LCL filter configurations for a GCI [112]. Although his work compares seven filter topologies, many of which are fourth-order filters, Ahmed does consider LCL versions of FC 1 and 2 for a 10kVA system. Despite the thirdorder nature, the simulations show that the parallel damped arrangement (similar to FC 2) produces copper losses of 1.5kW (0.15pu), compared to 6.5W (0.0065pu) for the series damped capacitor (similar to FC 1). Hence, FC 2 produces a power loss of about 230 times than that of FC 1.

Filter configuration 3 is not considered in Ahmed's study, however, it appears to produce the least power loss. The drawback, however, is that FC 3 is only able to attenuate the high-frequency PWM harmonics as a first-order filter, and hence an output current THD vs. power loss trade-off exists. This is further examined in the following section.

### 5.3.8 Design Trade-Offs - Unipolar PWM Waveform

It is well known that the inverter must meet strict grid power factor and THD requirements in order to operate as a grid-connected inverter. The effect of filter parameter variation, such as cutoff frequency and quality factor, on the both the output current THD and power factor has been examined in Sections 5.3.5 and 5.3.6, respectively. In addition, the damping resistor power loss has also been discussed, albeit separately. The power loss and THD are now considered simultaneously, for a  $f_{sw}$  of 4kHz, and  $C_n$  of 0.12pu.

Figure 5.43 shows the damping resistor power loss vs. filter output current THD for FC 1–4 and  $f_c$  equal to 0.2 and 0.3pu. Simulations over a wider range of  $f_c$  can be found in the appendix (Section A.2.1). The shaded area represents the design region, which is bound by power loss and THD limits of 5%. The power loss is determined using the inverter associated current,  $I_f$ , which includes the filtered harmonic components up to  $7 f_{sw}$ , and the grid drawn current,  $I_g$ . The figure shows that each filter is able to limit power loss to 5%, whilst meeting the THD specification, for some value of  $R_d$  and hence Q.



Figure 5.43: Filter power loss,  $P_d$ , vs. output current THD for cutoff frequencies,  $f_c$ , of (a) 0.2, and (b) 0.3 pu (relative to the inverter switching frequency,  $f_{sw}$ ). The shaded region indicates the design region, where both the THD and power loss is less than 5%.

Although the output current THD, power factor and filter power loss are of prime importance, the filter quality factor should also be known as this has the potential to amplify harmonics, which ultimately influences the output current THD for a non-ideal input current. Figure 5.44 hence shows the quality factor as a function of normalised cutoff frequency, for each filter configuration (FC). Four shaded regions are shown, these correspond to areas where: A) the THD is less than 5%, B) the THD and power loss are both less than 5%, C) the power loss is less than 5%, and D) both the THD and power loss are greater than 5%; the solid lines represent contours of THD and or power loss equal to 5%. Note that the switching frequency is set to 4kHz and that the filter capacitance is maintained at 0.12pu for each FC, such that the power factor standards are met.



Figure 5.44: Power loss and THD contours of 5%, for filter configurations (a) 1, (b) 2, (c) 3 and (d) 4, for a switching frequency of 4kHz. The solid lines represent contours of THD and damping resistor power loss,  $P_d$ , equal to 5%. The resulting shaded regions represent the area where the A) THD, B) THD and damping resistor power loss, and C) power loss, are less than 5%, whilst area D) corresponds to the region where both the THD and power loss exceed 5%. Note that area B is referred to as the *design region*.

The design region allows the inductor to be easily selected, based on a desired cutoff frequency, filter configuration and filter capacitance. Similarly, the damping resistor can also be determined knowing the quality factor and filter configuration. Note that the inverter performance can be improved, i.e. the output current THD and damping resistor power loss can be reduced by increasing the filter quality factor. This is seen in Figure 5.45, which shows THD and power loss contours of 2, 3, 4 and 5% for each filter configuration. Note that the  $f_{sw}$  and  $C_n$  are maintained at 4kHz and 0.12pu, respectively.



Figure 5.45: Power loss and THD contours of 2, 3, 4 and 5%, for filter configuration (FC) (a) 1, (b) 2, (c) 3 and (d) 4, for a switching frequency of 4kHz and a capacitance of 0.12pu. The THD and power loss,  $P_d$ , of each filter can be reduced by increasing the quality factor, Q, with FC 3 showing the least Q and  $P_d$  variance.

Despite being first-order filters, Figures 5.44 and 5.45 show that FC 1 and 3 are the only configurations that meet Nave's recommendation of a Q between 2 and 4. In addition, FC 3 is able to meet this over a wider range of cutoff frequencies, and its  $P_d$  is less susceptible to quality factor variations. In contrast, both second-order filters can not meet Nave's suggested Q range; the minimum Q which allows FC 4 to lie within the design region, is 6.5, whilst that of FC 2 is about 20 (for a  $f_c$  of 0.1pu). Recall that  $\omega_{cn}$  should be greater than one decade above  $\omega_B$ , to prevent a filter phase delay. Hence  $f_{cn}$  should exceed 0.125pu for the above 4kHz  $f_{sw}$  case.

#### Effect of Varying $C_n$

The design regions of Figure 5.44 are shown for a capacitance of 0.12pu, as this value allows the filter to meet the 0.8 leading power factor whilst operating at 20% rated power. In addition, it also restricts  $L_n$  to a small range of values (see Figure 5.41), which is desired as larger capacitors are generally cheaper and easier to obtain than large inductances. If desired, the range of  $L_n$  can be increased by reducing  $C_n$ , however this increases  $Z_{0n}$ .

The drawback of increasing  $Z_{0n}$  is that  $R_d$  must also be increased to maintain a given Q. This action will increase  $P_d$  for the series connected RLC filters (FC 1 and 4), and reduce losses for parallel connected RLC filters (FC 2 and 3). Consider FC 1 and 4. Assuming the presence of  $R_d$  does not affect the RMS current in the filter, the power loss in both filters is proportional to  $R_d$ . In practise, however, the presence of  $R_d$  will reduce the current in the part of the circuit where  $R_d$  is located and hence the power loss would increase more slowly than  $R_d$ .

A summary of the effect of reducing  $C_n$  on the filter design region is shown in Table 5.6 for each FC and  $C_n$  equal to 0.12, 0.06 and 0.03pu. The table shows that the THD contour of 5% is unaffected by varying  $Z_{0n}$ , whilst the power loss contour of 5% is affected (as explained above). Consider the  $P_d$  contour equal to 5% for each filter configuration, for a fixed  $f_{sw}$ , e.g. 0.2pu. The minimum Q of FC 1 and 4, is shown to be inversely proportional to  $C_n$ , whilst it is directly proportional to  $C_n$  for FC 2, and is unchanged for FC 3.

Assuming that a variation of  $C_n$  and  $L_n$  does not affect the filter currents (for a fixed THD),  $R_d$  does not change for a fixed  $P_d$ . The  $Z_{0n}$ , however, is inversely proportional to  $C_n$  for a fixed  $\omega_{cn}$ . Since Q is directly proportional to  $Z_0$  for FC 1 and 4 (series RLC circuits), the minimum Q corresponding to a fixed  $P_d$  must also be inversely proportional to  $C_n$ . Similarly, the Q of FC 2 is inversely proportional to  $Z_{0n}$ , and hence the minimum Q corresponding to a  $P_d$  of 5% is directly proportional to  $C_n$ .

#### Effect of Varying $f_{sw}$

The effect of increasing the  $f_{sw}$  on each filter's design region is summarised in Table 5.7, for  $f_{sw}$  equal to 4, 7 and 10kHz, and for a fixed  $C_n$  of 0.12pu. The table shows that the design region increases with  $f_{sw}$  for FC 1, 3 and 4, whilst it decreases for FC 2. This occurs as  $\omega_{cn}$  increases with  $f_{sw}$ , which results in an increase in  $L_n$  and hence  $Z_{0n}$ . The minimum Q corresponding to a fixed  $P_d$  is hence inversely proportional to the  $f_{sw}$  for FC 1 and 4. In contrast, the minimum Q for FC 2 is inversely proportional to  $f_{sw}$ , whilst that for FC 3 is not affected (for  $f_c$  equal to 0.2pu). Table 5.6: Effect of varying filter capacitance on filter design region, for filter configurations (FC) 1, 2, 3 and 4. The  $C_n$  is equal to 0.12 (leading power factor limit), 0.06 and 0.03pu. Note that the design region for FC4 is not displayed for  $C_n = 0.03$ pu due to the high quality factor, Q, required.



Table 5.7: Effect of varying switching frequency,  $f_{sw}$ , on filter design region, for filter configurations (FC) 1, 2, 3 and 4. The  $f_{sw}$  is equal to 4, 7 and 10kHz, for a filter capacitance of 0.12pu.



In summary, the design region of FC 1 and 4, is maximised by increasing either  $C_n$  (within the leading power factor limit) or the  $f_{sw}$ . Conversely, the design region of FC 2 can be maximised by reducing either  $C_n$  or the  $f_{sw}$ . Filter configuration 3 is shown to be least affected by either  $C_n$  or  $f_{sw}$  variation, and offers the lowest value of Q. These attributes make FC 3 the filter of choice.

### 5.3.9 Effect of Non-Ideal Current Source

Thus far, it is assumed that the filter is supplied with a unipolar PWM current that is created using an ideal constant current source, however, the proposed GCI uses a PM generator and uncontrolled rectifier. Therefore, the effects of this non-ideal current source (NICS), are examined here. Unlike the ideal PWM current case, the quality factor becomes important as low and medium-frequency harmonics exist; these are caused by the fluctuating output power ( $\alpha_0$ ) and the rectifier ripple ( $f_{rect}$ ), respectively. These harmonics will be amplified if the  $f_{res}$  is equal (or close) to the harmonic frequencies, and if Q exceeds 1. This is highly likely, as Figure 5.44 indicates that the Q for FC 1, 2 and 4, must be greater than 4 to meet the THD and power loss design region. In addition, the resonant frequency will be fixed (due to fixed filter components), whilst the rectifier ripple frequency will vary with turbine (wind) speed.

The THD vs. cutoff frequency is shown in Figure 5.46, using the NICS and for cases i)  $\hat{\alpha}_0$  and  $f_{rect} = 0.2$ pu, and ii)  $\hat{\alpha}_0$  and  $f_{rect} = 0.4$ pu. The plot appears similar to that of the ideal PWM input current case (Figure 5.38), however, the THD is shown to increase at low and medium cutoff frequencies. This is caused by resonance; the extent of which depends on the quality factor, and value of  $\hat{\alpha}_0$ . The figure also identifies the source of harmonics, e.g. those caused by the fluctuating input power ( $\hat{\alpha}_0$ ), the rectifier ripple ( $f_{rect}$ ) or the PWM switching (PWM).

The key conclusion from Figure 5.46 is that the NICS significantly reduces the range of  $f_c$ , for which the inverter is able to meet the grid THD requirement. The inverter is shown to exceed this 5% requirement if  $\hat{\alpha}_0$  exceeds 0.4pu;  $\alpha_0$  can be reduced by over-sizing the machine (as mentioned in Section 5.2.6), however, this does not remove the rectifier ripple harmonics. The ideal solution is to remove the low and medium-frequency harmonics caused by the NICS, such that the filter must only attenuate the high-frequency (PWM) harmonics, and the effects of resonance can be hence be ignored. The low and medium-frequency harmonics can be effectively removed by incorporating a more sophisticated,



e.g. feedback or feed-forward, controller. A feed-forward control algorithm is investigated in Section 5.4.

Figure 5.46: THD vs. cutoff frequency and quality factor (Q), showing the effects of fluctuating output power  $(\hat{\alpha}_0)$  and the rectifier ripple frequency  $(f_{rect})$ , for filter configurations (a) 1 and 3, and (b) 2 and 4. The THD is shown for (top)  $\hat{\alpha}_0$  and  $f_{rect} = 0.2$ pu, and (bottom)  $\hat{\alpha}_0$  and  $f_{rect} = 0.4$ pu. The harmonics sources are identified in the latter.

#### 5.3.10 Alternative (Third-Order) Filter Configurations

Third-order low pass filters that meet the impedance mismatch criteria are shown in Figure 5.47, these include the *parallel* and *series*-damped third-order filters. Note the term series or parallel-damped refers to the connection of the damping resistor with respect to the reactive elements (either  $C_d$  or  $L_d$ ). Both filters have a roll-off rate of -40dB/decade,

however the parallel-damped filter has lower power losses [105]. These filters are beyond the scope of this work and are further discussed in [105,113]. Given that FC 3 is promising it is not necessary to further investigate third-order filters.



Figure 5.47: Third-order variations of the damped LC low-pass filters, including the (a) parallel and (b) series damped configurations.

# 5.4 Feed-Forward Control

# 5.4.1 Introduction

A system which shows feed-forward behaviour responds to a measured input disturbance in a pre-defined way; this differs to feedback systems that deal with any deviation from the desired system output behaviour. Feed-forward systems require a system model and are hence sensitive to model inaccuracies. The controlled variable will deviate from the desired output unless all input disturbances are anticipated [114] and are accurately calculable [115]. Feed-forward controllers hence use sophisticated calculations to accurately model input disturbances [114]. Despite this, feed-forward systems are stable and respond quickly [116]. In contrast, feedback systems do not require a system model, as they are based on closed-loop control. They are hence slower to respond, and may become unstable, however, they are able to deal with unique disturbances.

Conventional voltage-source, current-controlled inverters use feedback control based on sampling the output current. The difference between the instantaneous output and reference currents is known as the error. The controller adjusts the inverter switching signals to minimise the error. In contrast, a feed-forward controller would sample an input parameter, e.g. the machine speed or inverter input current, and apply a predetermined control pattern to the inverter. Both controllers require a fast processor for rapid sampling, however, the feed-forward system generally has lower costs, as the required sensors are generally cheaper. The machine speed can be easily measured using a low-cost Hall-effect sensor or the inverter input current can be measured using a low-cost shunt resistor, whilst the equivalent feedback control system requires a more expensive isolated current sensor.

# 5.4.2 Aim of Proposed Feed-Forward Control

Feed-forward control is investigated here to remove the low and medium-frequency harmonics, due to the fluctuating input power ( $\hat{\alpha}$ ) and the rectifier ripple, respectively. Section 5.2.3 showed that the low-frequency harmonics, and hence the THD, could be reduced by decreasing the modulation index. However, this also reduced the fundamental component and was unable to reduce the rectifier ripple harmonics. The proposed feed-forward (compensation) scheme aims to remove the low and medium-frequency harmonics, by adjusting the modulation index (and hence duty-cycle) in real-time, such that a sinusoidal output current with zero distortion is obtained. Note that the inverter output current obtained using the feed-forward controller is referred to as the *compensated* current. An example of the open-loop inverter input and compensated inverter output currents is shown in Figure 5.48, for a rectifier ripple frequency of 1.8kHz, and a  $\hat{\alpha}_0$  of 0.25pu. The compensated output current,  $i_c(t)$ , has a peak equal to  $\beta\sqrt{3}/2$  pu, as this corresponds to the minimum rectifier ripple. Hence a compensated current whose peak is less than or equal to the minimum open-loop inverter input current will not contain harmonic distortion caused by the rectifier ripple or the fluctuating input power ( $\hat{\alpha}_0$ ).



Figure 5.48: Feed-forward current compensation concept, showing the normalised openloop inverter input current and the rectified compensated inverter output current. The rectifier ripple frequency is shown for 1.8kHz, whilst  $\hat{\alpha}_0$  is equal to 0.25pu. Note that the terms *peak*, *DC* (*mean*) and *minimum* refer to the open-loop inverter input current.

The ideal compensated output current,  $i_{c \ (id)}(t)$ , is expressed by Equation (5.45), which substitutes  $\beta$  (from above) by Equation (5.7), for the ideal case. The experimental compensated current,  $i_{c \ exp}(t)$ , is derived in a similar manner and is shown in Equation (5.46); this is derived by substituting  $\beta$  by Equation (5.8).

$$i_{c\ (id)}(t) = \frac{\sqrt{3}}{2} \sqrt{1 - \widehat{\alpha}_0^2} . \sin(\omega_g t) \quad (pu)$$
(5.45)

$$i_{c (exp)}(t) = \frac{\sqrt{3}}{2} \left( 3.86 \,\widehat{\alpha}_0^{\,5} - 7.47 \,\widehat{\alpha}_0^{\,4} + 4.27 \,\widehat{\alpha}_0^{\,3} - 1.53 \,\widehat{\alpha}_0^{\,2} - 0.116 \,\widehat{\alpha}_0 + 1 \right) . \sin(\omega_g t) \quad (pu)$$
(5.46)
Two types of feed-forward control are proposed to obtain a distortion-free output current. The first samples the generator frequency,  $f_m$ , whilst the second technique samples the inverter input current,  $i_{in}(t)$ . Both techniques instantaneously manipulate the inverter modulation index,  $m_a$ , (and hence the duty-cycle) such that a sinusoidal inverter output current is obtained.

## 5.4.3 Controller 1 (FFC 1): Sample Machine Frequency

Consider the <u>ideal</u> I-V locus. The resulting ideal inverter output current, which contains input current harmonic distortion, is expressed by Equation (5.15). The feed-forward controller is able to effectively remove the input current distortion if the ideal output current is equal to the ideal compensated current, i.e. Equation (5.45). This is summarised in Equation (5.47), which simplifies the ideal output current (right-hand side of the expression). This equality is rearranged and solved for in terms of  $m_a$ ; the resulting required time-varying modulation index,  $m_a(t)$ , is expressed in Equation (5.48), and is shown in Figure 5.49(b). The solution is expressed in terms of rect(t),  $\hat{\alpha}_0$  and  $\sin(\omega_g t)$ . The sinusoid can be stored in a look-up-table, however both rect(t) and  $\hat{\alpha}_0$  must be calculated in real-time; these parameters can be calculated from the machine frequency, using a Hall-effect sensor.

$$\frac{\sqrt{3}}{2}\sqrt{1-\hat{\alpha}_{0}^{2}}.sin(\omega_{g}t) = rect(t).m_{a}(t).sin(\omega_{g}t)\sqrt{1-V_{in}^{2}(t)}$$
(5.47)

$$m_a(t) = \frac{\sqrt{2}}{2.rect(t).sin(\omega_g t).\widehat{\alpha}_0} \sqrt{rect^2(t) - \sqrt{rect^4(t) - 3.rect^2(t).sin(\omega_g t).\widehat{\alpha}_0^2 \left(1 - \widehat{\alpha}_0^2\right)}}$$
(5.48)

A comparison of the inverter performance using both open-loop (uncompensated) and feed-forward compensated control, for the ideal rectifier I-V locus, is shown in Figure 5.49. The figure shows the modulation index and the subsequent inverter input and output currents for  $\hat{\alpha}_0 = 0.7$  pu. The benefit of feed-forward inverter control is clearly seen, i.e. the harmonic components are completely removed and the output current THD is ideally zero. Although the feed-forward controller significantly improves the inverter output current THD, its magnitude is now lower than its open-loop equivalent. This is further discussed for the experimental locus case in Section 5.4.5.



Figure 5.49: Comparison of the (a) open-loop and (b) feed-forward control schemes using the <u>ideal</u> I-V locus for  $\hat{\alpha}_0 = 0.7$  pu. Each control mode shows the modulation index and the resulting inverter input and output currents, from top to bottom, respectively.

Now consider the <u>experimental</u> locus. The time-varying modulation index is determined by a similar approach, however the experimental compensated current (Equation (5.46)) is equated to the experimental inverter output current and rearranged, as summarised by Equation (5.49). Unlike the ideal locus case, a *closed form solution* does not exist; this is verified by *Abel's impossibility theorem* that states that "*polynomial equations higher than fourth degree are incapable of algebraic solution in terms of a finite number of additions, subtractions, multiplications, divisions, and root extractions*" [117].

$$3.86 \,\widehat{\alpha}_0^{\,5} \cdot \sin^{10}(\omega_g t) \cdot m_a^{\,6}(t) - 7.47 \,\widehat{\alpha}_0^{\,4} \cdot \sin^8(\omega_g t) \cdot m_a^{\,4}(t) + 4.27 \,\widehat{\alpha}_0^{\,3} \cdot \sin^6(\omega_g t) \cdot m_a^{\,3}(t) \\ -1.53 \,\widehat{\alpha}_0^{\,2} \cdot \sin^4(\omega_g t) \cdot m_a^{\,2}(t) - 0.116 \,\widehat{\alpha}_0 \cdot \sin^2(\omega_g t) \cdot m_a(t) + 1 \\ -\frac{\sqrt{3}}{2} \sqrt{3.86 \,\widehat{\alpha}_0^{\,5} - 7.47 \,\widehat{\alpha}_0^{\,4} + 4.27 \,\widehat{\alpha}_0^{\,3} - 1.53 \,\widehat{\alpha}_0^{\,2} - 0.116 \cdot\widehat{\alpha}_0 + 1} = 0$$
(5.49)

The experimental I-V locus must be simplified if a closed-form solution is to be found. It is hence approximated, as shown in Equation (5.50). The resulting I-V locus for  $\hat{\alpha}_0 = 0.7$  pu is shown in Figure 5.50; the ideal and experimental loci are shown for reference.



Figure 5.50: Approximated, experimental and ideal rectifier I-V loci.

The small discrepancies between the approximated and measured loci imply that the compensated inverter output current will contain some distortion. This is seen in Figure 5.51, which shows the time-varying modulation index and the resulting inverter input and output currents; the output current contains 1.9% THD. The time-varying  $m_a$  is found using the same approach shown for the ideal case, i.e. by using  $\beta_{app}$  (Equation (5.50)) and hence solving Equation (5.51) in terms of  $m_a(t)$ ; the solution is expressed in Equation (5.52).

$$\frac{\sqrt{3}}{2}\sqrt{1 - (2 - \hat{\alpha}_0^2)\hat{\alpha}_0^2} \cdot \sin(\omega_g t) = rect(t) \cdot m_a(t) \cdot \sin(\omega_g t)\sqrt{1 - (2 - \hat{\alpha}_0^2)V_{in}^2(t)} \quad (5.51)$$

$$m_a(t) = \frac{\sqrt{2.rect(t) \cdot (2 - \hat{\alpha}_0^2)\left(rect(t) - \sqrt{rect^2(t) - 3.\sin^4(\omega_g t)\hat{\alpha}_0^2(2 - 5\hat{\alpha}_0^2 + 4\hat{\alpha}_0^4 - \hat{\alpha}_0^6)}\right)}{2.rect(t) \cdot \sin^2(\omega_g t)\hat{\alpha}_0(2 - \hat{\alpha}_0^2)} \quad (5.52)$$



Figure 5.51: Comparison of the (a) open-loop and (b) feed-forward control schemes using the approximated experimental I-V locus for  $\hat{\alpha}_0 = 0.7$  pu. Each control mode shows the modulation index, inverter input and output currents, from top to bottom, respectively.

Although this feed-forward control algorithm significantly improves the quality of the inverter output current, it is difficult to implement due to the complexity of the required time-varying  $m_a$  solution, which must be calculated in real-time. In addition, the peak compensated inverter output current is less than the peak open-loop current. This is due to the control algorithm that delivers a (compensated) sinusoidal inverter output current whose peak is equal to the minimum inverter input current.

Comparison of Figures 5.49(b) and 5.51(b) show that the peak compensated output current, using the (approximated) experimental I-V locus, is less than that using the ideal locus. This is due to the shape of the approximated experiential locus, which significantly reduces the minimum inverter input current, using open-loop control. Note that the peak experimental compensated output current is also less than the minimum inverter input current (see Figure 5.51(b)). This implies that the fundamental magnitude of the compensated output current can increase without containing distortion providing the peak compensated current is less than the minimum inverter input current. This is further examined in Section 6.3.3.

#### **Rectifier Ripple Phase Angle Errors**

Although the feed-forward control concept has been successfully demonstrated, it should be noted that the phase of the rectifier ripple current waveform is assumed to be known. This, however, will require rotor position information which increases the cost of implementation. The effect of the rectifier ripple phase angle error is analysed below.

The rectifier ripple (RR) phase angle error will affect the feed-forward controller's ability to remove the rectifier ripple harmonics, however, it will not affect the controller's ability to remove the low-frequency harmonics. The maximum phase error is 60°, and as such, the most RR distortion and hence compensated current THD is expected to occur for a phase error of 30°. In contrast the least rectifier ripple distortion and hence compensated output current THD is expected to occur when the rectifier ripple phase angle error is 0 or 60°. This is verified in Figure 5.52, which shows the compensated current THD as a function of rectifier phase angle error, using both the ideal and approximated experimental I-V loci. There is little difference between the ideal and experimental THD curves, except at phase angle errors of between 0 and 10°, and 50 and 60°, where the ideal locus produces a lower THD. Note that the peak compensated current THD is 8.2%, which is about twice that caused by the RR (4.19%), and that the THD curves are symmetrical and centred at 30°.



Figure 5.52: Compensated inverter output current THD vs. rectifier ripple (RR) phase angle error, for the (a) ideal, and (b) approximated experimental I-V loci. The THD is shown for  $\hat{\alpha}_0$  equal to 0.2, 0.4, 0.6 and 0.8pu. The dashed line represents the grid THD limit of 5%.

The above figure indicates that the feed-forward algorithm will only meet the grid THD requirement over a small range of phase angle errors, i.e. less than  $\pm 11^{\circ}$ . This, coupled with the fact that low-cost microcontrollers may be unable to perform such complex  $m_a$  calculations in real-time, justifies the need for a simpler and faster feed-forward control algorithm.

#### 5.4.4 Controller 2 (FFC 2): Sample Inverter Input Current

The second feed-forward controller samples the inverter input current, hence it is not required to be calculated in real-time. This allows the controller to quickly calculate the required duty-cycle and hence modulation, based on the ideal boost equation, as shown in Equation (5.53), which relates the inverter input and output currents,  $i_{in}(t)$  and  $i_{out}(t)$ , respectively. Note that this was previously expressed in terms of the rectifier and wave-shaper output currents (Equation (5.9)).

$$i_{out}(t) = [1 - d(t)] . i_{in}(t)$$
 (5.53)

The term [1 - d(t)], from the above expression, is given by  $m_a(t) \cdot \sin(\omega_g t)$ , as shown in Equation (5.54). The term  $i_{out}(t)$  is now replaced by the appropriate expression for the desired (compensated) output current, and hence the time-varying modulation index can be found. The solution, however, is dependent on the I-V locus used, i.e. Equations (5.55) and (5.56) represent the time-varying modulation indices using the ideal and the experimental I-V loci, respectively.

$$i_{out}(t) = m_a(t) \cdot \sin(\omega t) \cdot i_{in}(t) \tag{5.54}$$

$$m_a(t) = \frac{\frac{\sqrt{3}}{2}\sqrt{1-\hat{\alpha}_0^2}}{i_{in}(t)}$$
(5.55)

$$m_a(t) = \frac{\frac{\sqrt{3}}{2} \left(3.86 \,\widehat{\alpha}_0^{\,5} - 7.47 \,\widehat{\alpha}_0^{\,4} + 4.27 \,\widehat{\alpha}_0^{\,3} - 1.53 \,\widehat{\alpha}_0^{\,2} - 0.116 \,\widehat{\alpha}_0 + 1\right)}{i_{in}(t)} \tag{5.56}$$

The second feed-forward controller is able to calculate the required time-varying modulation index much faster than the first feed-forward scheme, due to the simpler expression for  $m_a(t)$ . Despite this,  $\alpha_0$  must also be calculated, which hence requires a generator speed sensor and knowledge of the generator (I-V locus) and a gird voltage sensor. Note that the calculation time can be reduced by using look-up-tables to store i) the normalised rectifier I-V locus, i.e.  $\beta_0$  vs.  $\alpha_0$ , and the generator speed vs.  $\alpha_0$  information. The second feed-froward controller is selected over the first type, as it is simpler and does not require rotor position information. A summary of the feed-forward controller is shown in Figure 5.53, it replaces the *Modulation Index* box (on the left-hand side) of Figure 4.17.



Figure 5.53: Feed-forward controller summary, showing the input and output stages, and the look-up-tables (LUT). The diamonds represent input stages and include the sampled grid voltage,  $v_g$ , input current,  $i_{in}(t)$ , and generator speed,  $\omega_m$ , whilst the time-varying modulation index,  $m_a(t)$ , is the controller output.

Figure 5.54 compares the inverter performance using both the open-loop and the second feed-forward controller, for the <u>experimental</u> rectifier I-V locus. The figure shows the calculated modulation index and the subsequent inverter input and output currents for  $\hat{\alpha}_0 = 0.7$  pu. The benefit of the feed-forward controller is clearly demonstrated, i.e. the low and medium-frequency harmonics are eliminated. The output current now contains zero THD, irrespective of which I-V locus is used. Note that both feed-forward controllers behave alike when using the ideal I-V locus (see Figure 5.49), hence it is not examined.



Figure 5.54: Comparison of the (a) open-loop and (b) feed-forward compensated control schemes using the experimental I-V locus for  $\hat{\alpha}_0 = 0.7$  pu. Each control mode shows the modulation index, inverter input and output currents, from top to bottom, respectively.

# 5.4.5 Comparison of Feed-Forward and Open-Loop Control

This section compares the inverter's performance using both open-loop and the two feedforward controllers. The inverter output current THD and fundamental magnitudes are compared, over a wide range of wind speeds. The THD and fundamental magnitude is first examined as a function of  $\hat{\alpha}_0$ , as  $\hat{\alpha}_0$  varies with wind and hence turbine speed.

#### Inverter Performance vs. $\hat{\alpha}_0$

This section examines the THD and fundamental magnitude of the inverter output current waveform, when comparing open-loop and both feed-forward controllers, for the <u>experimental</u> locus, over a wide range of  $\hat{\alpha}_0$ . Figure 5.55(a) shows that both of the feedforward control algorithms significantly reduce the inverter output current THD, and allow the inverter to meet the grid THD requirement over a much wider range of  $\hat{\alpha}_0$ , compared to that of the open-loop controller. Note that the open-loop case corresponds to a modulation index of 100%, and the phase angle of the rectifier ripple is assumed to be known for FFC 1.



Figure 5.55: Inverter output current (a) THD, and (b) fundamental magnitude vs.  $\hat{\alpha}_0$  using open-loop and both feed-forward control algorithms. The dashed line in (a) represents the grid THD requirement, whilst in (b) it corresponds to the calculated minimum inverter input current for the experimental locus,  $\sqrt{3}/2 \beta_{exp}$ . Note that the open-loop curve corresponds to a modulation index of 100%.

Comparison of the above THD plots, shows that FFC 1 contains some residual distortion (< 2%), whilst FFC 2 is ideally able to eliminate all of the harmonics. The small level of output current distortion, using FFC 1, is caused by the difference in the approximated and experimental inverter input currents, which itself is a result of the slight discrepancies between the approximated and the experimental I-V loci (recall Figure 5.50). In contrast, the output current THD of FFC 2 is zero, as the inverter input current is measured and not approximated.

Despite the reduction in output current THD, Figure 5.55 shows that the fundamental magnitude of the compensated output current is lower than its open-loop equivalent. Recall that the compensated current should have a fundamental magnitude less than or equal to  $\beta_{exp} \sqrt{3}/2$ , in order to effectively eliminate the rectifier ripple associated harmonics; this is shown in Figure 5.55(b) by the dashed line. The figure shows that FFC1 is able to deliver an output current whose magnitude closely matches that of the expected curve, however, small discrepancies exist at higher values of  $\hat{\alpha}_0$ , due to the approximated experimental I-V locus. In contrast, FFC2 is able to yield a sinusoidal output current whose fundamental is exactly equal to  $\beta_{exp} \sqrt{3}/2$ , as the inverter input current is measured and not approximated.

Although both feed-forward controllers were demonstrated to effectively reduce the inverter output current THD, the second controller performs better as it yields a purely sinusoidal output current with zero distortion. In addition, this controller uses a much simpler calculation of the time-varying modulation index than FFC1. As such, only FFC2 is considered in the following section.

#### Inverter Performance vs. Wind Speed

The inverter output current THD and normalised output power is examined as a function of normalised wind speed. This is shown in Figure 5.56 for both open-loop and feedforward controllers. Note that only FFC 2 is examined. The open-loop inverter output power and current THD is obtained from Figures 5.17(a) and 5.18, respectively, whilst those using FFC 2, are determined from  $\hat{\alpha}$  (Figure 5.17) and Figure 5.55.

The feed-forward controller delivers a sinusoidal output current, with zero distortion, to the grid over the entire range of wind speeds. The open-loop controller, however, can only meet the grid THD requirement below rated wind speed, providing the low-pass filter resonant frequency does not coincide with any harmonics. Despite this, the inverter is able to deliver more power to the grid using open-loop control, as the FFC 2 reduces the fundamental magnitude of the output current in order to remove the rectifier ripple harmonics. Hence, a power quality (THD) vs. quantity (power) trade-off exists below rated wind speed, for a given generator, providing the effects of resonance are ignored.



Figure 5.56: Inverter output (a) current THD, and (b) power, vs. wind speed, for both open-loop and feed-froward (FFC 2) controllers. The dashed line in (a) represents the grid THD requirement, which applies at rated output power.

Despite the trade-off, the benefits of the feed-forward controller significantly outweigh its open-loop alternative, i.e. the feed-forward controller delivers high quality current at all wind speeds, and allows the low-pass filter to be designed for any resonant frequency. Note that the open-loop controller only obtains slightly more power than FFC 2 for a given generator. Hence, the inverter can easily deliver rated power to the grid by simply over-sizing the generator.

# 5.5 Chapter Summary

Open-loop sinusoidal modulation of the current wave-shaper (SMR) switch means that the input current harmonics appear in the output current. The rectifier ripple current by itself causes an output current THD of 4.19%, which is independent of generator speed, whilst that caused by the low-frequency (100Hz power) ripple is proportional to the ratio of the rectifier output voltage under load to its open-circuit voltage and hence inverter output power. This was shown graphically using the I-V locus, as the generator is pushed in to the variable-voltage region, where it is unable to provide constant current.

The grid requires an output current with less than 5% THD, which assuming only low-frequency rectifier ripple are present in the inverter output current, implies the peak inverter output voltage must be less than 23% of the rectified open-circuit voltage. The low-frequency ripple is shown to decrease with modulation index, as this effectively reduces the effective rectifier output voltage and hence output power. Despite the reduced input current ripple, the output current fundamental is also reduced with decreasing modulation index, due to its linear relationship with input current. This causes the output current THD to increase due to the presence of the PWM harmonic components.

A feed-forward controller produces the desired output current waveform despite a varying input current. This is done by either sampling the or accurately calculating the instantaneous inverter input current and adjusting the modulation index (and hence duty-cycle) in real-time.

The high-frequency PWM harmonics of a current-source grid-connected inverter are removed by a capacitive-inductive low-pass grid filter, according to the impedance mismatch criteria. This type of filter is analysed, and it is shown that damping is required to limit the effect of resonance. Four damping resistor locations, and their respective power loss vs. harmonic attenuation (output current THD) trade-offs are discussed. The configuration with the damping resistor connected in parallel to the filter inductor, offers the least power loss, whilst sufficiently attenuating the high-frequency PWM harmonics. In addition, the choice of filter capacitance and inductance is shown to be a trade-off between the cut-off frequency and grid power factor. The analysis shows that a cut-off frequency of at least ten times the fundamental is required to prevent significant filter delay, and that a filter capacitance of less than 0.12pu is required to meet the light-load Australian Standard power factor requirement. A detailed filter design approach is produced, that determines the filter parameters based on the desired cut-off frequency, power loss and ideal output current THD.

# Chapter 6

# Design and Simulation of 1kW Grid-Connected Inverter System

This chapter uses the principles of wind power and the inverter analysis from Chapter 5 to design a 1kW grid-connected wind turbine that delivers high quality power to the grid over a wide range of wind speeds. The chapter begins with a turbine design that allows the inverter to provide rated power at rated wind speed. This is followed by a low-pass filter design that ensures the inverter complies with the grid specifications, at rated wind speed using an open-loop control mode. Finally, the inverter and turbine are shown to meet the grid requirements over a wide range of wind speeds, by employing a simple and low-cost feed-forward controller.

# 6.1 Turbine Sizing and Machine Parameter Selection

The wind turbine needs to be designed such that the inverter is able to deliver rated power, 1kW, to the grid at the chosen rated wind speed, 12m/s. The Fisher & Paykel<sup>®</sup> machine tested in Chapters 2 and 4 is unable to provide the required 1kW+ of power and hence a new generator is required for the proposed 1kW grid-connected inverter. Although electrical machine and turbine blade design is beyond the scope of this work, a suitable turbine and generator can be determined based on a series of assumptions.

## 6.1.1 System Assumptions

This section lists the assumptions made to calculate the size (power rating) of a suitable generator and wind turbine that delivers 1kW of electrical power to the grid at rated wind speed. Note that the assumptions listed regard the inverter, turbine and generator.

# Inverter Assumptions

- the inverter switching frequency is limited to 10kHz, as this limits switching losses,
- the inverter is 95% efficient, based on calculated power loss caused by thyristor and diode voltage drops,
- the maximum damping resistance power loss,  $P_d$ , is limited to 50W (0.05pu),

# Turbine Assumptions

- the peak  $c_p$  is equal to 0.30, for a turbine whose blade radius is about 1m [1],
- a 5 bladed turbine is used, which is identical in shape as that tested in Chapter 3, however, the blade length is changed,
- the blade power coefficient characteristic  $(c_p \text{ vs. } \lambda)$  is based on that determined experimentally, however, its peak  $c_p$  is increased from 0.20 to 0.30, which occurs for a tip-speed ratio,  $\lambda$ , of 7.5; the  $c_p$  curve is shown in Section 6.1.2.

# Generator Assumptions

- the machine has 24 pole-pairs (the same as a Fisher & Paykel<sup>®</sup> type outer rotor surface permanent magnet machine),
- the stator is star (wye) connected,
- the machine is 85% efficient at rated wind speed,
- the stator impedance is mainly inductive, such that the high-inductance characteristic is achieved,
- the back-EMF constant selected such that the ratio of the peak grid voltage to the peak generator rectified output voltage under open-circuit conditions,  $\hat{\alpha}_0$ , is equal to 0.25pu at rated wind speed.

# 6.1.2 Turbine Power and Size Calculations

The wind turbine blade radius, r, is calculated based on the wind power required at rated wind speed. The wind power,  $P_W$ , is calculated knowing the maximum coeffi-

cient of performance,  $c_p$ , and the required wind turbine power,  $P_{WT}$ , as summarised in Equation (6.1); where  $\rho$  and v represent the air density and wind velocity, respectively.

$$P_W = \frac{P_{WT}}{c_p} = \frac{1}{2} \pi r^2 \rho v^3 \tag{6.1}$$

The wind turbine power is calculated from the total inverter input power,  $P_{in}$  and the inverter and generator efficiencies,  $\eta_{inv}$  and  $\eta_{gen}$ , respectively. This is summarised by Equation (6.2), which simplifies the total inverter input power,  $P_{invin}$ , as the summation of inverter output power,  $P_{inv}$ , and the maximum damping resistance power loss,  $P_d$ . Substitution of  $P_{WT}$  in to the above equation allows the turbine radius to be simplified, as summarised by Equation (6.3).

$$P_{WT} = \frac{P_{inv\ in}}{\eta_{inv} \cdot \eta_{gen}} = \frac{P_{inv} + P_d}{\eta_{inv} \cdot \eta_{gen}} \tag{6.2}$$

$$r = \sqrt{\frac{2\left(P_{inv} + P_d\right)}{\eta_{inv}.\eta_{gen}.c_p.\pi.\rho.v^3}} \tag{6.3}$$

The turbine speed can be determined once the turbine radius is calculated. The mechanical speed,  $\omega_m$ , is determined from Equation (6.4), where  $\lambda$  represents the tip-speed ratio. Note that  $\omega_m$  has units of rad/s, which is converted to a rotational speed with units rpm, n, using Equation (6.5); the turbine  $c_p$  curve used here is shown in Figure 6.1.

$$\omega_m = \frac{\lambda \, v}{r} \tag{6.4}$$

$$n = \omega_m \ \frac{60}{2\pi} \tag{6.5}$$

The above equations allow a turbine radius and operating speed to be calculated at rated wind speed, knowing a desired inverter power rating and a turbine blade characteristic (coefficient of performance). A summary of the turbine parameters required for use with the proposed 1kW grid-connected inverter, is summarised in Table 6.1.



Figure 6.1: Turbine coefficient of performance,  $c_p$ , vs. tip-speed ratio,  $\lambda$ , characteristic used to calculate turbine operating speed at rated wind speed. The peak  $c_p$  is equal to 0.30 and occurs for a  $\lambda$  of 7.5. Note that the curve is based on the experimentally measured curve, however, it is scaled to achieve a higher peak  $c_p$ .

Table 6.1	: Summary	of ca	lculated	${\rm turbine}$	properties	that	allows	${\rm the}$	proposed	grid-
connected	d inverter to	deliver	1kW to	the grid	l, at rated v	wind s	speed.			

Property	Value
Rated Inverter power, $P_{inv}$	$1000 \mathrm{W}$
Damping resistor power loss, $P_d$	$50 \mathrm{W}$
Inverter efficiency, $\eta_{inv}$	95~%
Generator efficiency, $\eta_{gen}$	85~%
Wind turbine power, $P_{WT}$	$1300 \mathrm{W}$
Peak $c_p$	0.30
Air density, $\rho$	$1.225 \text{ kg/m}^3$
Blade length (radius), $r$	$1.146 {\rm m}$
Tip-speed ratio, $\lambda$ , for peak $c_p$	7.50
Rated Wind Speed, $v_r$	$12 \mathrm{m/s}$
Rated Turbine mechanical speed, $\omega_m$	78.54  rad/s
Rated Turbine mechanical speed, $n$	$750 \mathrm{rpm}$

# 6.1.3 Generator Equivalent Circuit Parameter Selection

The generator can be designed, as the turbine size and operating speed, at rated wind speed, has been determined. The design process is listed below:

- 1. calculate the required back-EMF constant to give  $\hat{\alpha}_0 = 0.25 \text{pu}$ ,
- 2. find the generator short-circuit current that allows the inverter to deliver rated power, using the feed-forward current compensation scheme, to the grid,

- 3. determine the stator resistance from the assumed generator efficiency and losses, and
- 4. use the short-circuit current and back-EMF constant to calculate the stator impedance, and hence reactance (knowing the resistance).

#### Step 1 – Calculate the Back-EMF Constant

The back-EMF constant, k, is determined such that  $\hat{\alpha}_0$  is equal to 0.25pu at rated wind speed. This allows the peak generator rectified output voltage under open-circuit conditions,  $V_{R pk}$ , to be calculated knowing the peak grid voltage,  $V_{g pk}$ ; recall that  $\hat{\alpha}_0$  is the ratio of these peaks, as summarised by Equation (6.6). Note that the peak rectifier output voltage is greater than the average DC output voltage,  $V_{DC}$ , as shown in Equation (6.7);  $V_{DC}$  itself is expressed relative to the peak phase voltage,  $V_{ph pk}$ , in Equation (6.8) [56].

$$\widehat{\alpha}_0 = \frac{V_{g \ pk}}{V_{R \ pk \ OC}} \tag{6.6}$$

$$V_{DC} = 0.9557 \, V_{R \ pk} \tag{6.7}$$

$$V_{DC} = \frac{\pi}{2} V_{ph\ pk} \tag{6.8}$$

The nominal Australian grid voltage, at the *point of supply*, is  $230 V_{RMS}$ , as set by Australian Standard 60038 [118]. Although a tolerance of +10% and -6% exists, the proposed grid-connected inverter will be designed for a rated voltage of 230 V. This, along with Equations (6.6) - (6.8), implies that the generator peak back-EMF voltage must be equal to 866.7 V, at rated wind speed. This allows the back-EMF constant, to be calculated as summarised by Equation (6.9).

$$k = \frac{E}{\omega_m} \tag{6.9}$$

#### Step 2 – Calculate the Short-Circuit Current

The generator short-circuit (SC) current is calculated knowing the peak compensated inverter output rated current, which is determined using the experimental I-V locus. The peak compensated current must be less than the minimum inverter input current in order to remove both the rectifier ripple and fluctuating output power harmonics. The minimum rectifier (inverter input) current,  $I_R$ , is calculated as the product of  $\beta$  and the minimum rectifier ripple, as summarised in Equation (6.10).

$$\min\left(I_R\right) = \beta \, \frac{\sqrt{3}}{2} \tag{6.10}$$

This concept is illustrated in Figure 5.48 and is again shown in Figure 6.2 for convenience. The open-loop inverter input current is shown for  $\hat{\alpha}_0 = 0.25$  pu, which corresponds to a  $\check{\beta}$  of 0.898 pu, and a compensated inverter output current with a peak of 0.778 pu. The generator peak short-circuit current,  $I_{SC \ pk}$ , can hence be calculated, as shown by Equation (6.11), where  $I_B$  is the inverter base (rated) current. Therefore, the proposed GCI requires a generator peak SC current of 7.91A to deliver rated power to the grid.



Figure 6.2: Normalised inverter input and peak compensated output current, for  $\hat{\alpha}_0$  0.25pu. The rectifier ripple frequency is shown for 1.8kHz, as this corresponds to rated turbine speed (750rpm). Note that the terms *peak*, *DC (mean)* and *minimum* refer to the inverter input current waveform.

$$I_{SC\ pk} = \frac{\sqrt{2}\,I_B}{\min\left(I_R\right)}\tag{6.11}$$

#### Step 3 – Calculate the Stator Resistance

The stator resistance can be determined from the generator copper losses, which is the difference between the assumed generator losses,  $P_L$ , and the assumed iron and friction and windage losses under rated operating conditions. This is summarised by Equation (6.12), where  $P_{IFW}$  and  $P_{CU}$  represent the iron, friction and windage, and copper losses, respectively. Note that  $P_L$  is equivalent to the difference in wind turbine and generator output power.

$$P_L = P_{IFW} + P_{CU} \tag{6.12}$$

The assumed total power loss is obtained from Table 6.1, whilst the iron and friction and windage losses are a function of machine speed. These speed dependent losses are assumed identical to those discussed in Chapter 2, as the generator types are assumed similar. The measured iron and friction and windage losses are shown again in Equation (6.13) for convenience, where  $n_k$  represents the machine speed in k rpm. Once determined, the copper loss and hence stator resistance,  $R_s$ , can be calculated from Equation (6.14), where  $I_{ph}$  is the stator phase current. Note that the mean stator phase current is equal to 0.9538 pu, relative to the generator short-circuit current for  $\hat{\alpha}_0 = 0.25$  pu.

$$P_{IFW} = 42.88 \, n_k^2 + 32.54 \, n_k \tag{6.13}$$

$$R_s = \frac{P_{CU}}{3 \cdot \text{mean} \left(I_{ph}\right)^2} \tag{6.14}$$

#### Step 4 – Calculate the Stator Inductance

The stator inductance,  $L_s$ , is calculated from the stator reactance,  $X_s$ , which itself is determined from the stator impedance,  $Z_s$ , and the stator resistance. The later is summarised by Equation (6.15), where  $Z_s$  is calculated as the quotient of the phase induced back-EMF voltage and phase short-circuit current, as shown in Equation (6.16).

$$X_s = \sqrt{Z_s^2 - R_s^2} \tag{6.15}$$

$$|Z_s| = \frac{E}{I_{SC}} \tag{6.16}$$

The stator inductance is calculated from Equation (6.17), where  $\omega_e$  is the electrical frequency; this is proportional to the machine mechanical speed (in rad/s) and the number of pole-pairs, p, as shown in Equation (6.18).

$$L_s = \frac{X_s}{\omega_e} \tag{6.17}$$

$$\omega_e = n \, \frac{2\pi}{60} \, p \tag{6.18}$$

#### **Generator Summary**

The above procedure is used to design a PM generator that is suitable for use with the proposed grid-connected inverter. Hence, the machine must provide the inverter with sufficient current, at rated wind speed, such that high quality rated power is delivered to the grid. A summary of the calculated generator properties, at rated wind speed, is shown in Table 6.2.

Property	Value
Generator speed, $n$	750 rpm
Mechanical speed, $\omega_m$	78.54  rad/s
Pole-pairs, $p$	24
Mechanical frequency, $f_m$	300  Hz
Rectifier ripple frequency, $f_{rect}$	1.80 kHz
Electrical frequency, $\omega_e$	1885  rad/s
Back-EMF constant, $k$	$7.127 \ V/(rad/s)$
Induced phase voltage, $E$	559.7 V
Short-circuit current, $I_{SC}$	$5.60 \ A$
Phase resistance, $R_s$	1.708 $\Omega$
Phase impedance, $Z_s$	99.94 $\Omega$
Phase reactance, $X_s$	1.708 $\Omega$
Phase inductance, $L_s$	53.0  mH

Table 6.2: Calculated generator parameters for the proposed 1kW grid-connected inverter system. Note that all stator parameters are listed as phase quantities at rated wind speed.

The resulting rectifier I-V locus is shown in Figure 6.3, which shows the rectifier output current vs. rectifier output voltage. The locus shows current and voltage as peak, DC (average), and minimum quantities; recall the inverter input current of Figure 6.2.

The dashed lines represent the peak grid voltage and the minimum peak current for compensation to be achieved. The figure indicates that the feed-forward compensation scheme will effectively remove the input current harmonic distortion, as the minimum peak current, at the grid peak voltage, exceeds that required.



Figure 6.3: Designed generator <u>rectifier</u> output I-V locus, showing the peak, DC (average), and the minimum peak current vs. voltage. The vertical dashed lines represents the peak grid voltage, whilst the horizontal dashed line represents the minimum peak current required to compensate (remove the input current harmonic distortion) the output current. The compensation scheme is shown to be effective at the peak grid voltage.

# 6.2 Low-Pass Filter Design

# 6.2.1 Design Criteria

The filter is designed according to the selection criteria discussed in Chapter 5; these are listed below:

- power factor between 0.8 lead and 0.95 lag between 20% and rated operation,
- $\bullet\,$  output current THD less than 5% at rated operation,
- peak damping resistance power loss of 0.05pu or less (relative to rated power),
- inverter switching frequency less than 10kHz, and
- filter cutoff frequency between 10 and 65pu (relative to the fundamental frequency).

# 6.2.2 Component Selection

The filter components are selected using the normalised low-pass filter analysis of Chapter 5. The procedure is listed below, and yields normalised filter component values. These are used in conjunction with the filter base quantities, to determine actual component values. A list of filter component base quantities is summarised in Table 6.3, for the proposed (1kW) grid-connected inverter.

# **Output Filter Design Procedure**

- 1. select the inverter switching frequency,
- 2. select the low-pass filter configuration,
- 3. select the value of normalised capacitance,  $C_n$ , that meets the leading power factor requirement, i.e.  $C_n \leq 0.12$ pu,
- 4. use the  $P_d$  and THD vs. quality factor and cutoff frequency contour plot to determine a desired power loss and THD, i.e. THD and  $P_d \leq 5\%$ ,
- 5. determine the inductance to match the desired cutoff frequency, and
- 6. calculate the damping resistance that allow the desired quality factor to be achieved.

Property	Value
Base power, $P_B$	$1000 \mathrm{W}$
Base grid voltage, $V_B$	$230 V_{RMS}$
Base impedance, $Z_B$	$52.9 \ \Omega$
Base current, $I_B$	$4.348 \text{ A}_{RMS}$
Fundamental frequency, $f_1$	$50 \mathrm{~Hz}$
Base frequency, $\omega_B$	314.2  rad/s
Base capacitance, $C_B$	$60.17~\mu\mathrm{F}$
Base inductance, $L_B$	$168.4 \mathrm{~mH}$

Table 6.3: Low-pass filter base quantities.

The above design procedure is used to design two CL low-pass filters, to show the effect of varying the switching frequency,  $f_{sw}$ , on the filter performance. The first filter is designed for a  $f_{sw}$  of 4 kHz, as this corresponds to that of the concept demonstration inverter tested in Chapter 4. The second filter is designed for 10kHz operation, as this corresponds to the maximum allowable  $f_{sw}$  (see Section 5.3.4).

Filter configuration 3 was selected for the proposed inverter; the damping resistor of this filter arrangement is connected in parallel to the filter inductor. Though this configuration offers only a first-order attenuation rate beyond the resonant frequency, it can still meet the grid THD requirement and offers the lowest damping resistor power loss,  $P_d$ , as seen in Section 5.3.8.

The normalised filter capacitance is the next parameter to be determined. The inverter is designed to meet the leading power factor requirement of 0.8 at 20% rated volt-amperes. This condition is met for an undamped CL filter when the normalised capacitance is equal to 0.12pu. The introduction of a damping resistor will affect the power factor as some real power will be dissipated in  $R_d$ , however, the power loss is limited to 0.05pu at rated operation. The small power loss only slightly adjusts the inverter power factor, as mentioned in Section 5.3.6, and is hence ignored.

The required filter cutoff frequency is determined using the quality factor, Q, vs. power loss,  $P_d$ , vs. THD tradeoff curves of Section 5.3.8. For convenience these are shown in Figure 6.4 for a normalised filter capacitance of 0.12pu. Figure 6.4(a) shows the tradeoff curve for a switching frequency of 4 kHz, whilst Figure 6.4(b) shows that for  $f_{sw}$  equal to 10 kHz. Each trade-off curve show the chosen filter design point (shaded circle) which corresponds to a Q of 4, based on Nave's recommendation, and an output current THD of 2%, which implies a normalised cutoff frequency of 0.155pu. The resulting filter power loss, under rated conditions, is 0.2% and 0.38% of the rated power, for switching frequencies of 4 and 10kHz, respectively. Note that the THD curves shown assume an ideal current source input and a modulation index,  $m_a$ , of 100% is used with the inverter. It is hence expected that either using the PM generator (non-ideal) current source, or reducing the  $m_a$ , will increase the output current THD.



Figure 6.4: Quality factor, THD, power loss and cutoff frequency tradeoff curves, used to design the 1kW grid-connected inverter, for switching frequencies of (a) 4 kHz, and (b) 10 kHz. The shaded circle represent the chosen filter parameters, i.e. a cutoff frequency of 0.155pu, a quality factor of 4, an output current THD of 2%, a filter power loss of less than 2%.

The above trade-off curves allow the normalised filter components to be evaluated, from which, the filter inductance and damping resistance is calculated using the appropriate equations from Chapter 5. A summary of the filter components, for both switching frequencies, is seen in Table 6.4.

Table 6.4: Output filter component values and corresponding filter parameters for the proposed 1kW grid-connected inverter, for both 4 and 10kHz inverter operation.

Component / Parameter	Switching Frequency			
Component / 1 arameter	4 kHz	$10\mathrm{kHz}$		
Capacitance, $C$	$7.22 \ \mu F$	$7.22 \ \mu F$		
Inductance, $L$	9.1 mH	$1.5 \mathrm{mH}$		
Damping Resistance, $R_d$	137.7 $\Omega$	55.08 $\Omega$		
Resonant Frequency, $f_{res}$	620 Hz	$1.55 \mathrm{~kHz}$		
Quality Factor, $Q$	4	4		

# 6.2.3 Filter Simulation - Ideal Current Source

Each low-pass filter is simulated, using PSIM<sup>®</sup>, in an open-loop manner using an ideal current source. The resulting inverter output current and voltage, at rated and 20% rated operation, is shown in Figure 6.5, for each switching frequency. Note that the power levels are set by adjusted the  $m_a$ . A summary of the inverter performance is shown in Table 6.5. The output current THD and damping resistor power loss is equal to that predicted by Figure 6.4, at rated operation. Note that the THD considers only the PWM harmonics due to the ideal current source input.



Figure 6.5: Simulated ideal inverter output current and voltage at rated and 20% rated volt-amperes, for switching frequencies of (a) 4, and (b) 10kHz. The waveforms shown are the rated current, the 20% rated current, and the inverter output voltage, from top to bottom, respectively. The zero position for each waveform is shown by the arrow on the left. The vertical current and voltage scales are 5A and 200V per division, respectively, whilst the horizontal scale is 5ms/div.

Switching	Apparent	Power	Current	Power Loss	Modulation
Frequency	Power (pu)	Factor (lead)	THD (%)	in $R_d$ (W)	Index $(\%)$
4 kHz	0.201	0.800	6.36	0.323 W	16.0
	1.013	0.993	1.97	$1.909 { m W}$	100
10 l-U-	0.199	0.796	8.65	0.120 W	16.0
10 KHZ	1.006	0.993	2.11	$0.383 \mathrm{W}$	100

Table 6.5: Simulated inverter performance using an ideal current source.

The table indicates that the power factor, THD, and the (self imposed) power loss requirements are met at rated current. In contrast, the inverter output current falls marginally short of the 0.8 leading power factor requirement, for the 10 kHz case. In addition, the output current THD at 20% rated operation is slightly higher than 5%, for each case. This is not an issue as the 5% requirement only applies at rated power. The THD appears to be inversely proportional to the fundamental magnitude, which implies the harmonic components are relatively constant.

Although the current of Figure 6.5 appears to be lagging the grid voltage (more clearly seen for the 20% rated current), it is leading, according to Standards Australia, who specify the power factor must consider the inverter as a load with respect to the grid. Hence, an inverter that supplies a lagging current is equivalent to one drawing a leading current. This convention is used throughout the remainder of the thesis. In addition, the inverter output voltage is simply the grid voltage, which remains unchanged irrespective of the inverter output current. It is hence not seen in the subsequent simulations.

# 6.2.4 Filter Simulation - PM Generator Current Source

The output current of each filter has been shown to meet the grid requirements using an ideal current source. Each filter is hence simulated using the PM generator current source in an open-loop manner; the generator properties are listed in Table 6.2. The filter output current hence contains distortion caused by the rectifier ripple and the fluctuating input current. These aspects, together with the effects of resonance, i.e. harmonic amplification, are taken into consideration when calculating the filter output current THD (see Section 5.3.9).

Figure 6.6 shows the calculated output current THD of each filter, as a function of cutoff frequency, for a modulation index of 100%. The shaded circles represent the calculated THD for each filter. The 4kHz case produces an output current THD of 3.65%, whilst that for the 10kHz case is 10.1%. The THD of the latter is significantly higher than the required 5%. This is due to the effects of resonance, i.e. the rectifier ripple harmonics are amplified, as these lie in close proximity to the cutoff (resonant) frequency.

The simulated inverter output currents, for switching frequencies of 4 and 10kHz, are shown in Figure 6.7. Each subfigure shows three currents, these correspond to the maximum, rated and 20% rated cases, from top to bottom respectively. Recall that the generator is designed to provide a compensated rated current of magnitude 1pu, i.e. it is oversized. Hence, operating the inverter in open-loop control mode, with a modulation index of 100% yields an output current greater than 1pu; in this case the peak current is equal to 1.207pu. Note that the effect of resonance is clearly seen for the 10kHz case.



Figure 6.6: Output current THD prediction using the PM generator current source, for switching frequencies of (a) 4, and (b) 10 kHz. The points represent the calculated THD for a modulation index of 100%.



Figure 6.7: Various simulated inverter output currents, using the PM generator current source and open-loop control, for switching frequencies of (a) 4kHz, and (b) 10kHz. The currents shown are the maximum, rated and 20% rated currents, from top to bottom, respectively. The zero position for each waveform is shown by the arrow on the left. The vertical scales are 5A/div, and the horizontal scale is 5ms per division.

The oversized generator implies that the inverter modulation index must be reduced to obtain rated current. In this case, rated current is delivered to the grid for a  $m_a$  of 81.8%. Similarly, the 20% rated current is obtained when the modulation index is set to 12.9%. This is summarised in Table 6.6, along with power factor and THD information of each current waveform. The power factor is shown to fall marginally short at 20% rated power, whilst meeting the requirement at rated and maximum powers. The THD is again shown to be high for the 20% rated current case, which decreases with increasing power for the 4 kHz switching case. In contrast, the THD is shown to be relatively constant with power for the 10 kHz case. This is expected as the filter cutoff frequency is close to the rectifier ripple frequency (0.155 vs. 0.18pu, respectively). Hence, the harmonics found in the rectifier ripple are amplified, due to resonance (Q = 4), and so the major distortion component of the output current is the rectifier ripple. The rectifier ripple is a constant fraction of the fundamental, due to the action of the current wave-shaper, and hence the THD is expected to be relatively constant.

Note that the rectifier ripple and resonant frequencies were deliberately selected to show the effects of resonance. In addition, the calculated and simulated THD values match closely, i.e. only a small error of 0.6 and 0.2% exist for the 4 kHz and 10 kHz cases, respectively.

Table 6.6: Simulated inverter performance, using the (non-ideal) PM generator current source with open-loop control. Data is provided for switching frequencies of 4 and 10 kHz, and for 20, 100 and 121% rated power conditions.

Switching	Apparent	Power	Current	Power Loss	Modulation
Frequency	Power (pu)	Factor	THD	in $R_d$	Index $(\%)$
	0.201	0.793 lead	6.78~%	$0.334 { m W}$	12.9~%
$4\mathrm{kHz}$	1.001	0.990 lead	4.52~%	$2.831 { m W}$	81.8~%
	1.207	0.992 lead	4.24~%	$2.877 {\rm W}$	$100 \ \%$
	0.200	0.796 lead	9.17~%	$0.134 { m W}$	12.9~%
$10\mathrm{kHz}$	1.000	0.990 lead	10.2~%	$1.498 { m W}$	81.8~%
	1.207	0.992 lead	10.3~%	$1.721 { m W}$	100~%

In addition, the above table also shows the filter power loss, due to the damping resistance. The power loss is shown to decrease as the switching frequency increases. This is expected as the filters have a fixed normalised cutoff frequency, and by increasing the  $f_{sw}$ , the inductance must decrease. To maintain a constant Q, the damping resistance must also decrease; the power loss is proportional to the damping resistance, and hence must also decrease.

#### Suitability of Open-Loop Controller

The current waveforms of Figure 6.7 and the THD information of Table 6.6 suggest that operating the proposed grid-connected inverter in open-loop control mode is suitable only for the 4 kHz switching case at rated wind speed. Recall Figure 5.18, which shows that the open-loop THD will decrease with wind speed (due to  $\hat{\alpha}_0$  decreasing with wind speed). Note that this is valid if the rectifier ripple frequency does not approach the low-pass filter resonant frequency. This can not be guaranteed as the rectifier ripple frequency varies with turbine and hence wind speed.

Consider the 10kHz case. The output current THD is significantly higher than the 5% requirement, at rated wind speed. It is foreseen that the inverter THD will initially increase and then decrease with wind speed. This occurs as the rectifier ripple frequency varies with speed, and that the rectifier ripple at rated wind speed is greater than the resonant frequency, i.e. the rectifier ripple harmonics will be further amplified below rated wind speed. The effects of resonance, however, should be less evident at low wind speeds.

Open-loop operation of the proposed grid-connected inverter is hence deemed unsuitable for both switching frequencies, due to the effects of resonance. The inverter must incorporate a feedback or feed-forward controller to eliminate the open-loop inverter input current (and hence output current) harmonic distortion.

# 6.3 Demonstration of Feed-Forward Control

Two feed-forward control algorithms that effectively removed the open-loop inverter input current harmonics (fluctuating output power and rectifier ripple), were examined in Section 5.4. The first controller (FFC 1) sampled the generator speed, and calculated the distorted input current and hence the required modulation index to remove the harmonics. This was deemed unsuitable as the controller was required to calculated several complex functions in real time. Hence, a second controller (FFC 2) that sampled the inverter input current was investigated. This algorithm also adjusts the modulation index in real-time, however, it is much simpler and faster and hence appropriate for real-time calculations.

# 6.3.1 Control Implementation

The feed-forward controller was implemented in PSIM<sup>®</sup> using a current sensor and current reference command, as summarised by Figure 6.8. Note that the current wave-shaper diode (seen in Figure 4.18) has been removed due to the reverse-blocking capability of the thyristors. The time-varying normalised modulation index is simply the reference current divided by the time-varying inverter input current. The current command,  $I^*$ , must be expressed as a peak quantity, e.g. to obtain a sinusoidal RMS current of 3A, the current reference must be  $4.24 A_{pk}$ . Note that the  $m_a$  will exceed 1pu, if  $I^*$  exceeds the peak inverter input current; this is further discussed in Section 6.3.3.



Figure 6.8: Partial PSIM<sup>®</sup> inverter circuit showing the relevant feed-forward controller implementation. The time-varying normalised modulation index is the quotient of the current command (reference) and sampled rectifier output current.

## 6.3.2 Proof of Concept at Rated Wind Speed

The feed-forward concept is demonstrated for both the 4 and 10 kHz switching cases, in Figure 6.9, which shows the compensated output currents for both the rated and 20% rated cases. This illustrates that the feed-forward control algorithm effectively removes the inverter input current harmonic distortion, this is more clearly seen for the 10 kHz case, where the rectifier ripple component is no longer visible. A summary of the current waveforms, including the THD, power factor and filter power loss is seen in Table 6.7.



Figure 6.9: Simulated compensated inverter output current for (top) rated, and (bottom) 20% rated current, for switching frequencies of (left) 4kHz and (right) 10kHz. The zero position for each waveform is shown by the arrow on the left. The vertical and horizontal scales are 5A and 5ms per division, respectively.

Table 6.7: Current compensated inverter performance summary, using the (non-ideal) PM generator current source and feed-forward controller.

Switching	Apparent	Power	Current	Power Loss
Frequency	Power (pu)	Factor	THD	in $R_d$
41-11-	0.201	0.798 lead	5.93~%	0.330 W
4 K N Z	1.000	0.993 lead	3.69~%	$2.777 { m W}$
10 l-U-	0.200	0.800 lead	5.93~%	0.118 W
10 KHZ	1.008	0.993 lead	2.81~%	$0.697 \mathrm{W}$

The above table indicates that the use of the feed-forward controller allows both inverters to meet the THD and power factor requirements at rated power, whilst both inverters meet the applicable power factor requirements for the 20% rated case (assume that 0.798  $\sim$  0.8). Although both rated current cases produce a THD less than 5%, it is the 10 kHz case that has the lower THD; this is despite it having a cutoff frequency close to the rectifier ripple frequency. This result hence verifies that the feed-forward controller is able to effectively remove the rectifier ripple harmonics.

In addition, the rated compensated output current THD, of both cases, is larger than the designed 2% of Figure 6.4. This occurs as the desired 2% THD is calculated assuming an ideal current source and for a modulation index of 100%, whereas the somewhat oversized generator delivers rated current for an average modulation index of about 80% (see Table 6.6). The resulting compensated current THD is expected to be larger than the desired 2%, as a reduction in  $m_a$  increases the PWM harmonic components whilst decreasing the fundamental magnitude (see Section 5.2.4).

## Comparison to Open-Loop Control Mode

The improvement made by the feed-forward controller is best seen when comparing the compensated and open-loop equivalent currents. This is seen for both the 4 and 10 kHz inverter cases, in Figures 6.10 and 6.11, respectively. The most significant improvement is seen for the rated current, using the 10 kHz case, as the effect of resonance (amplified rectifier ripple harmonics) is removed. A summary of the open-loop and feed-forward compensated currents is also shown in Table 6.8; the compensated cases are highlighted.



Figure 6.10: Simulated inverter output currents for (a) rated, and (b) 20% rated operation, for a switching frequency of  $\underline{4 \text{ kHz}}$ . The waveforms represent currents using (top) openloop, and (bottom) feed-forward controllers. The vertical and horizontal scales are 5A and 5ms per division, respectively.



Figure 6.11: Simulated inverter output currents for (a) rated, and (b) 20% rated operation, for a switching frequency of 10 kHz. The waveforms represent currents using (top) openloop, and (bottom) feed-forward controllers. The vertical and horizontal scales are 5A and 5ms per division, respectively.

Table 6.8: Comparison of open-loop (OL) and feed-forward (FF) compensated inverter performance. Current information is provided for rated and 20% rated power, for switching frequencies of 4 and 10 kHz; the compensated cases are highlighted.

Switching	Apparent	Control	Power	Current	Power Loss
Frequency	Power (pu)	Mode	Factor (lead)	THD (%)	in $R_d$ (W)
	0.200	OL	0.793	6.78	0.334
4 kHz	0.200	$\mathbf{FF}$	0.798	5.93	0.330
4 K11Z	1.000	OL	0.990	4.52	2.831
	1.000	$\mathbf{FF}$	0.993	3.69	2.777
	0.200	OL	0.796	9.17	0.134
10 kHz	0.200	$\mathbf{FF}$	0.800	5.93	0.118
	1.000	OL	0.990	10.2	1.498
	1.000	$\mathbf{FF}$	0.993	2.81	0.697

# 6.3.3 Current Command Variation at Rated Wind Speed

Although the benefit of the feed-forward controller has been clearly demonstrated at rated power, it is also important to investigate the inverter performance over a wide range of powers, as the inverter will operate at sub-rated power levels during periods of low wind speeds. The current command,  $I^*$ , is used to adjust the real and hence apparent output power. Recall that the apparent power is the complex summation of the real and reactive powers, and that the filter capacitor will draw a current of 0.12pu from the grid (for  $C_n = 0.12$ pu). Hence, a current command of 0.16pu is required for the inverter to operate at 20% (0.2pu) rated apparent power.

The current command is varied in two stages, the first investigates the inverter performance for current and hence output powers less than rated, whilst the second stage attempts to extract more than rated power from the generator.

#### Linear Modulation Region $(m_a \leq 100\%)$

The inverter performance is first examined for apparent power ratings between 0.2 and 1pu, as the inverter is required to meet strict power factor standards over this operating range. As mentioned above, the apparent power is set by adjusting the current command and hence real power. This is highlighted in Figure 6.12, which shows the ideal analytical (using the equations from Section 5.3.6) and (PSIM<sup>®</sup>) simulated apparent power as a function of normalised current command, for switching frequencies of 4 and 10 kHz.



Figure 6.12: Apparent power, S, vs. normalised current command,  $I^*$ , for switching frequencies of (a) 4, and (b) 10 kHz. The solid lines represent the ideal analytical power, whilst the points correspond to simulated data.

The inverter power factor is displayed in Figure 6.13, as a function of normalised apparent power, for both switching frequencies. The figure shows the ideal analytical and simulated data, as well as the grid requirements; the latter is represented by the dashed lines. The inverter is shown to meet the grid power factor requirements between 20% and rated apparent power, and has a leading power factor over the entire range of S.



Figure 6.13: Inverter power factor vs. normalised apparent power, S, for switching frequencies of (a) 4, and (b) 10 kHz. The solid lines and points represent analytical and simulated data, respectively, whilst the dashed lines correspond to the grid requirements.

The current THD is now examined, as the power factor requirements are met. Figure 6.14 shows the simulated inverter output current THD as a function of normalised apparent power. Note that the predicted THD is not shown due to the complexity of the calculation. The grid THD requirement is shown at rated power. power. The inverter is shown to meet the 5% THD requirement for both 4 and 10 kHz switching frequencies; the 10 kHz case has the lowest THD.



Figure 6.14: Simulated inverter output current THD vs. normalised apparent power, S, for switching frequencies of (a) 4, and (b) 10 kHz. The points represent the simulated data, whilst the triangle correspond to the grid requirement at rated power.

#### Over-Modulation Region $(m_a > 100\%)$

Increasing the current command beyond 1pu is likely to cause output current distortion, as the peak output current will be greater than the minimum value of the rectifier current ripple. This is illustrated in Figure 6.15, for current commands of 1.1 and 1.3pu for  $\hat{\alpha}_0 = 0.25$  pu. Note that that the current scale is normalised, i.e. it is expressed relative to rated inverter current. The figure shows that the maximum compensated current of about 1.15pu should be achievable, however, the input current will contain some of the rectifier ripple harmonics and hence the output current THD will increase. In practise, the output current THD increases, if the  $I^*$  is greater than about 1.1pu (for  $\hat{\alpha}_0 = 0.25$  pu).



Figure 6.15: Normalised inverter input and commanded output currents for current command,  $I^*$ , equal to (a) 1.1 and (b) 1.3pu.

Consider Figure 6.15(b). The peak current command is shown to exceed the peak rectifier ripple current. The feed-forward control algorithm will attempt to extract the desired power, under this condition, however, it will fail at the peaks of the sinusoid. This is best understood by revisiting the PWM derivation. Recall that the current wave-shaper PWM duty-cycle command (DCC) is determined by comparing the product of a normalised rectified grid-frequency sinusoid and the time-varying modulation index (sinewave reference), with a normalised switching frequency triangular wave. Hence, a modulation index greater than 1pu, implies that the duty-cycle command (actually 1-d) will also exceed 1pu. This occurs around the peaks of the compensated current, and implies that the current wave-shaper (CWS) switch will remain continuously open until the duty-cycle command falls below 1pu. The inverter behaves similar to open-loop mode in this situation, i.e. the effects of resonance are seen in the output current.
An example of the compensation failure, caused by over-modulation, is seen in Figure 6.16, for a  $I^*$  of 1.3pu. The figure shows the time-varying  $m_a$ , and the resulting: DCC, CWS gate signal and the filtered output current, for  $f_{sw}$  of 4 and 10 kHz. The figure illustrates that the CWS switch remains continuously open if the DCC exceeds 1pu, during which time the input current harmonics pass directly to the output where they can be amplified by the output filter resonance. This is clearly seen in the output currents, particularly for the 10 kHz case. In addition, the output current also shows a some oscillations once the DDC falls below 1pu; this is caused by the filter step response.



Figure 6.16: Inverter control signals and filter output current for a current command of 1.3pu, for switching frequencies of (a) 4kHz, and (b) 10kHz. The waveforms shown are the time-varying modulation index, duty-cycle command, current wave-shaper gate signal, and filter output current, from top to bottom, respectively. The output current shows the effect of resonance when the duty-cycle command exceeds 1 pu. The vertical scales are 0.5pu per division for the top three waveforms, and 5A per division for the output current (bottom waveform); the horizontal scale is 2ms per division.

The feed-forward current compensation control algorithm has been shown to fail for current commands that exceed approximately 1.1pu. It is hence expected that as the  $I^*$ increases beyond this value, the compensated current (and hence apparent power) will begin to saturate. This is seen in Figure 6.17, which shows the apparent power as a function of current command up to 1.5pu.



Figure 6.17: Apparent power, S, vs. normalised current command,  $I^*$ , over a wide range of values, for switching frequencies of (a) 4, and (b) 10 kHz. The solid lines represent analytical data, whilst the points correspond to simulated data. The apparent power begins to saturate beyond current current commands of approximately 1.1pu.

The resulting (displacement) power factor is expected to approach unity for increasing S, as the unity power factor inverter output current dominates the 0.12pu leading current drawn by the capacitor. This is seen in Figure 6.18, which shows the analytical and simulated power factor as a function of normalised apparent power. This is shown for both switching frequencies, and for current commands up to 1.5pu. The power factor is shown to remain within the grid specifications, i.e. between 0.8 lead and 0.95 lag for inverter power ratings between 20% rated and rated, despite the output current distortion. Note that the displacement power factor considers only the phase angle of the fundamental, whilst the true RMS power factor also considers the output current THD, hence, output current distortion will slightly reduce the true RMS power factor.

In contrast, the output current THD is expected to sharply increase with increasing apparent power; beyond about 1.1pu. The rise in THD is caused by two factors, firstly the clipping of the peaks of the output current, due to the limited input current, causes thirdorder harmonic distortion. Secondly, the saturation of the feed-forward control algorithm allows the rectifier current ripple to appear on the output which can be amplified by the



Figure 6.18: Inverter power factor vs. normalised apparent power, S, over a wide range S, for switching frequencies of (a) 4, and (b) 10 kHz. The solid lines represent calculated data, whilst the points correspond to simulated data. The power factor is shown to approach unity as the apparent power increases.

output filter. This was previously seen in Figure 6.16. The output current THD is plotted in Figure 6.19, as a function of S, for both inverter switching frequencies. The THD of the 10 kHz case is shown to increase at a faster rate than that for the 4 kHz case, as the rectifier ripple frequency and output filter resonant frequency are close to each other.



Figure 6.19: Simulated inverter output current THD vs. apparent power, S, for switching frequencies of (a) 4, and (b) 10 kHz, over a wide range of S. The THD sharply increases for  $S \ge 1.1$  pu, and at a faster rate for the 10 kHz switching case.

The above results suggest that in order to comply with the grid THD requirement, the rated current command should be defined as less than 1.1pu.

## 6.4 Inverter Simulation for Wide Wind Speed Range

The inverter has been shown to meet the grid power factor requirements at 20% rated and rated apparent powers, and the THD requirement at rated power. However, this is valid only for rated wind speed, and hence the inverter is examined over a wide range of wind speeds, to determine its suitability for grid-connected applications.

### 6.4.1 Turbine Characteristics

The wind turbine, whose properties are listed in Table 6.1, is simulated over a wide range of wind speeds, as seen in Figure 6.20. The figure shows inverter, generator and turbine output and wind power, as a function of wind speed, v, and assumes a constant inverter efficiency of 95%, a constant generator efficiency of 85%, and a peak  $c_p$  of 0.30. The desired inverter power increases with the cube of v until rated wind speed, after which, the power is maintained at rated power. Note that an S of 0.2pu occurs at v = 6.5m/s, where the real power is 0.16pu; recall the capacitor draws 0.12pu reactive power for all v.



Figure 6.20: Calculated inverter, generator and turbine output power, and total wind power vs. wind speed characteristic, based on assumed efficiencies. The dashed lines represent rated inverter power and wind speed, whilst the dashed-dotted lines correspond to 20% rated power (v = 6.5m/s, P = 160W).

The inverter, and hence turbine, operate such that the output power is maximised below rated wind speed and maintained at its rated value above rated wind speed. The turbine coefficient of performance,  $c_p$ , and tip-speed ratio,  $\lambda$ , should be kept constant at their optimal values below rated wind speed, and must decrease as v increases beyond rated. This implies the generator speed increases linearly with v up to rated, and must also decrease beyond rated wind speed. This action results in an a high  $\hat{\alpha}_0$  for low wind speeds, as the peak grid voltage is fixed. The value of  $\hat{\alpha}_0$  decreases to the desired 0.25pu at rated wind speed, and increases with v beyond rated. This information is summarised in Figure 6.21; the dashed lines represent rated wind speed.



Figure 6.21: Turbine and generator operating characteristics vs. wind speed characteristic. The curves shown include the turbine (a) coefficient of performance,  $c_p$ , and (b) tip-speed ratio,  $\lambda$ , and the generator (c) speed, and (d) value of  $\hat{\alpha}_0$ . The dashed lines represent rated wind speed.

Figure 6.22 shows the resulting  $\hat{\alpha}$  and the normalised minimum inverter input current,  $\check{\beta}$ , as a function of wind speed. Recall that the modulation index increases with the cube of wind speed (see Figure 5.17(a)). As such,  $\hat{\alpha}$  increases with the square of wind speed, until rated; it then increases beyond rated, as the turbine and hence generator speed decreases (see Figure 6.21(c)). Note that Figure 6.22(b) also shows  $\sqrt{3}/2\check{\beta}$ , as this corresponds to the peak compensated inverter output current that contains zero low and medium-frequency harmonics. The compensated output current has a peak of 0.778pu at rated wind speed (as seen in Figure 6.2), which corresponds to rated inverter current.



Figure 6.22: Generator (a)  $\hat{\alpha}$ , and (b)  $\check{\beta}$  vs. wind speed. Note that  $\sqrt{3}/2 \check{\beta}$  is also shown in (b) as this corresponds to the peak compensated output current;  $\check{\beta}$  is equal to 0.778pu at rated wind speed, which corresponds to rated inverter output current. The dashed line represents rated wind speed.

## 6.4.2 Power Control Modes

Although Figure 6.22(b) shows that the inverter can obtain rated current (and hence power) at rated wind speed, the figure suggests that the magnitude of the compensated output current must decrease above rated wind speed, in order to obtain distortion free output current. The modulation index and hence inverter output power, however, are ideally maintained at 1pu above rated wind speed to obtain rated inverter output power. It is hence foreseen that the compensated output current will contain some harmonic distortion above rated wind speed, unless the current command,  $I^*$ , is reduced above rated wind speed. This concept is illustrated in Figure 6.23, which shows the current commands for two control modes. Both control modes increase power, with the cube of wind speed, below rated wind speed, however, differ above rated wind speed. The first control mode (CM) attempts to maintain rated power, whilst the second mode reduces power above rated wind speed, by matching it to  $\check{\beta}$ , shown as the dashed line, such that the compensated current contains zero distortion. Note the  $I^*$  is only shown beyond a wind speed of 6.5m/s as this corresponds to the 20% rated apparent power wind speed.



Figure 6.23: Current command,  $I^*$ , vs. wind speed for control modes (a) 1, and (b) 2. The control modes differ beyond rated wind speed, where control mode 1 maintains rated output power, whilst mode 2 reduces the output power. The dashed and vertical dashed lines represent the maximum current command, and rated wind speed, respectively.

The inverter is simulated using both control modes in Section 6.4.4, where the effects of each control mode on the output current THD and magnitude, are examined above rated wind speed.

### 6.4.3 Optimised Component Selection

The components for the high power grid-connected inverter are carefully selected to optimise the overall inverter efficiency. Hence, components with low forward voltage drops and on resistance are selected to reduce conduction losses, whilst components with short on and off times are selected to reduce switching losses. A summary of the key inverter components is shown in Table 6.9. Note that the thyristors selected are designed for phase control applications and as such have a turn-off time,  $t_q$ , of  $100\mu$ s, which effectively limits the inverter switching frequency to 10 kHz. Commutation at the zero-crossings is ensured, for the 10 kHz case, as discussed in Section 4.3.3.

Component Property	Bridge Rectifier	MOSFET	Thyristors
Manufacturer	SEMIKRON	ST	IR
Part Number	DBI 25-14	STP25NM60N	12 TTS08 PbF
Rating	25A, 1400V	12.8A,600V	12.5A, 800V
Voltage Drop	1V (per diode)	-	1.2V
On Resistance	-	$0.17\Omega$	-
Latching Current	-	-	$50 \mathrm{mA}$
Holding Current	-	-	$30 \mathrm{mA}$

Table 6.9: Optimised inverter semiconductor properties.

Note: IR and ST denote International Rectifier, and STMicroelectronics NV, respectively.

An equivalent IGBT (STP25NM60N) was also considered for the current wave-shaper switch, however, the *turn on* and *turn off* times were significantly longer than those of the MOSFET and it hence had a higher switching loss. Though the IGBT has slightly lower conduction losses compared with the MOSFET, the IGBT produces the higher overall power (switching + conduction) loss. This, along with each device turn on and off times, is summarised in Table 6.10, for a switching frequency of 10kHz.

Table 6.10: MOSFET vs. IGBT turn on and off times, and maximum calculated losses, for a switching frequency of 10kHz.

Property	MOSFET	IGBT
On Time	18ns	70ns
Off Time	24ns	790 ns
Switching Loss	0.21W	4.3W
Conduction Loss	8.6W	7.4W
Total Loss	8.8W	11.7W

## 6.4.4 Inverter Simulations

The inverter using feed-forward compensation is now simulated using both control modes, at four wind speeds of interest: 6.5, 12, 15 and 17m/s. The former corresponds to 20% rated power, whilst the remaining wind speeds correspond to rated power (at and beyond rated v). The inverter input and output currents are shown for these cases in Figure 6.24, using power <u>control mode 1</u> (CM 1). The output current is shown to have low distortion for 20% and rated powers, however, the cases beyond rated wind speed show increasing current distortion as at times the command current exceeds the input current (as



expected); this is more clearly seen at the wind speed of 17m/s.

Figure 6.24: Simulated inverter input and filtered output currents, using power <u>control mode 1</u>, for wind speeds of (a) 6.5, (b) 12, (c) 15, and (d) 17m/s. The zero position for each waveform is shown by the arrow on the left. The vertical and horizontal scales are 5A and 5ms per division, respectively.

The inverter is next simulated using power <u>control mode 2</u> (CM 2), which for wind speeds that exceed rated, reduces the inverter output power to the maximum value while maintaining a low output current THD. The simulated inverter input and output currents are shown in Figure 6.25, for wind speeds of 15 and 17m/s only; this is due to identical current commands, and hence output currents, below rated wind speed.



Figure 6.25: Simulated inverter input and filtered output currents, using power <u>control mode 2</u>, for wind speeds of (a) 15, and (b) 17m/s. The zero position for each waveform is shown by the arrow on the left. The vertical and horizontal scales are 5A and 5ms per division, respectively.

The compensated output currents, of the above figure, show a greatly reduced amount of output current distortion compared to those obtained using CM 1. This is due to the reduced current command, which is equal to the minimum inverter input (rectifier output) current; this action slightly reduces the turbine power and hence speed. Therefore, the output current THD is reduced at the expense of the output current (and hence output power), i.e. a power vs. current THD trade-off exists beyond rated wind speed. This is summarised in Figure 6.26, which shows the apparent power and output current THD as a function of wind speed, for both control modes. Note that the apparent power curve of Figure 6.26(a) also shows the ideal power vs. wind speed relationship, i.e. maximum power below and rated power above rated wind speed. In contrast, Figure 6.26(b) compares the inverter output apparent power and current THD of CM 1, for comparison.

The output current THD of CM1 and 2 is identical to that previously seen in Figure 6.14, which also varies the current command between 0.16 to 1pu. Beyond rated wind speed, the current THD rapidly increases for CM1 in a similar fashion to that previously shown in Figure 6.19, where the current command increased beyond 1pu. This is similar to control mode 1, however, the current command remains at 1pu, whilst the minimum rectifier current falls below 1pu. Despite the difference in current commands, the effects of increasing the  $I^*$  or reducing the turbine and hence generator speed, are identical. In contrast, the output current THD, using CM2 only marginally increases beyond rated wind speed, and is hence selected for use with the proposed grid-connected inverter.



Figure 6.26: Inverter (top) apparent power, S, and (bottom) output current THD, vs. wind speed, for control modes (a) 1, and (b) 2. The vertical and horizontal dashed lines represent rated wind speed, and the grid THD requirement at rated power; the latter is hence only shown at and beyond rated wind speed. Subfigure (a) also compares the ideal inverter power, with that obtained using control mode 1 (points). Similarly, subfigure (b) compares the inverter power and output current THD obtained using control modes 1 for reference; this is shown as a solid line.

#### 6.4.5 Efficiency Analysis

The inverter is shown to meet the grid requirements, over a wide range of wind speeds, using control mode 2. The inverter efficiency, using this control mode, is hence examined. The inverter power-loss breakdown is shown in Figure 6.27, which shows the simulated power loss of various stages of the grid-connected inverter, including the three-phase uncontrolled rectifier, the current wave-shaper (CWS), the unfolding circuit and the output low-pass filter.



Figure 6.27: Break-down of inverter power loss vs. inverter output power, showing the rectifier, current wave-shaper, unfolding circuit, filter and hence total power loss.

The rectifier power loss of the above figure is shown to be relatively constant. This occurs as the PM generator operates in the constant current region of the I-V locus, over the wide range of inverter output power, which essentially maintains a constant inverter input current. The resulting rectifier power loss is proportional to the current and fixed diode voltage drops. The CWS power loss, which is proportional to the square of the RMS MOSFET current, is shown to decrease with increasing inverter output power. This occurs as the average MOSFET duty-cycle decreases as the current command (and hence inverter output current and power) increase. In contrast, the power loss of the unfolding circuit and filter power loss is proportional to the inverter output current and hence power. Note that the filter power loss appears constant, as it is small in magnitude compared to the inverter output power, however, it slightly increases with inverter output power.

The constant and decreasing power loss of the rectifier and CWS imply that their respective efficiencies will increase with inverter output power. In contrast, the increasing output filter and unfolding circuit power losses imply that their respective efficiencies are approximately constant. This is seen in Figure 6.28, which shows the simulated efficiency

of the rectifier, CWS, unfolding circuit and the low-pass filter. In addition, the figure also shows the overall inverter efficiency, which is simply the product of the efficiency of each stage of the inverter. As such, the inverter efficiency increases with inverter output power.



Figure 6.28: Inverter efficiency analysis, showing the rectifier, current wave-shaper, unfolding circuit, filter and overall inverter efficiency vs. output power.

The overall efficiency increases from about 86%, at 20% rated apparent power (real output power of 160W), to about 97.1% at rated power. This peak efficiency of 97.1% is slightly higher than other transformerless grid-connected inverter topologies (recall Table 1.5), and is mainly due to its lack of a DC link inductor and output transformer, which avoid copper and iron losses associated with each.

## 6.5 Chapter Summary

A 1kW grid-connected inverter based on the proposed topology was designed and simulated. The low-pass filter was designed based on the detailed analysis procedure described in Chapter 5. It was designed to ideally deliver a current with 2% THD to the grid, whilst meeting the light-load leading power factor requirements. Simulations were used to show that the inverter could meet the grid THD and power factor requirements with an ideal current source. However, it was unable to meet the grid THD requirement using the proposed generator. This was due to the use of open-loop sinusoidal modulation, which caused low-frequency and rectifier ripple input current harmonics to appear in the output current.

The introduction of feed-forward input current compensation, based on sampling the inverter input current, eliminated the effect of the low-frequency and rectifier ripple harmonics. The inverter demonstrated its ability to reduce output power, at rated wind speed, by reducing the reference current, whilst meeting the required power factor and THD standards. The inverter also showed it could increase its power, beyond its rated value, by increasing the current reference, i.e. by over-modulating the inverter. This, how-ever, increased the output current distortion, as the current wave-shaper switch remained open for a certain time, which allowed the effects of resonance to manifest itself around the peaks of the compensated current.

The turbine was simulated over a wide range of wind speeds from 6.5–12m/s, corresponding to 20% to 100% of rated output power. The inverter showed it can maximise the turbine and hence inverter power below rated wind speed. The turbine was also simulated up to wind speeds of 20m/s and the inverter was shown to be able to control the generator to keep the turbine speed below its rated value. Under these conditions with the standard control algorithm, the inverter can still produce useful output power but is not able to maintain rated output power without exceeding the THD limit, due to the reduced turbine speed.

The output current distortion, at high wind speeds, occurred as a result of overmodulation. This was resolved by introducing a new control concept that maximised the current reference and hence power whilst meeting grid THD standards. This resulted in a small reduction of current reference, i.e. an output current of 0.93pu was delivered to the grid whilst meeting the grid requirements for a wind speed of 17m/s. Despite the slight power reduction, the turbine was operated in this mode to meet the grid THD standards.

# Chapter 7

# **Conclusions and Future Work**

### 7.1 Summary and Conclusions

This thesis examined and validated the use of a low-cost power converter topology to control a small-scale wind turbine, for both standalone and grid-connected applications. The proposed system used a switched-mode rectifier (SMR) together with a high-inductance permanent magnet (PM) generator to produce a controlled output current. The highinductance characteristic allows the PM generator to operate as a speed-dependent constant-current source, unlike traditional low-inductance generators that act as speeddependent constant-voltage sources.

The proposed topology has the following advantages: i) it offers simple output current control, ii) avoids the size, cost, loss and reliability issues associated with energy storage elements, such as DC link inductors or capacitors, and iii) is able to extract power at low speeds. The low-speed generator output power capability with the SMR, however, was shown to be only half of that attainable using an inverter, which thus may require over-sizing the generator for a particular wind turbine. Despite this limitation, the SMR was shown to obtain the same output power at high speeds, and is significantly cheaper and simpler than the inverter.

The turbine speed was controlled by adjusting the SMR duty-cycle, which controlled the generator output current, and hence torque and speed. This method was shown to provide adequate control flexibility for a small-scale wind turbine, based on simulation and experimental testing. The turbine could be controlled such that i) the turbine output power is maximised below rated wind speed, ii) the turbine output power is maintained at its rated value above rated wind speed, and iii) the turbine speed is reduced at high wind speeds to prevent turbine damage.

The thesis is divided into two main parts. The first part examines the use of the SMR as a DC-DC converter to charge a battery, where the duty-cycle is kept constant under steady-state conditions, and is only adjusted to maximise or reduce power as the wind speed changes. The converter's ability to control the generator power and speed are simulated and experimentally verified using a dynamometer and a small wind tunnel. It was also shown that the SMR can deliver a controllable output current into a fixed voltage-source load making it suitable for a standalone system, such as a battery charger.

The second part of the thesis examined the design, construction, and testing of a novel grid-connected inverter (GCI) topology based on a sinusoidally modulated SMR and line-frequency commutated unfolding circuit. The topology was simulated and a 150W prototype was tested using a dynamometer. The output power was controlled by adjusting the modulation index in an open-loop manner. It was found that the topology was initially unable to meet the grid THD requirement, due to low-order current harmonics caused by resonance in the output CL filter. Each of the key parts of the inverter, including the generator acting as a non-ideal current source, the PWM switching scheme, and the low-pass filter were analysed in detail. A feed-forward control algorithm was also investigated and was shown to be able to remove the effect of the input current harmonics caused by the PM generator and rectifier current source.

The inverter analysis results were used to design a high-powered (1kW) grid-connected inverter to meet the Australian Standards. This involved selecting a suitable turbine diameter, equivalent-circuit parameters for the generator, and configuration and component values for the low-pass filter. The 1kW inverter was simulated over a wide range of wind speeds, and was shown to adequately control the small-scale wind turbine, whilst meeting the appropriate grid requirements.

# 7.2 Original Contributions

This work contains significant original contributions, all of which result from the investigation of controlling a small-scale wind turbine using the SMR topology together with a non-salient high-inductance PM generator. The SMR topology is simulated and tested for the DC-DC power converter case, whilst a novel GCI topology is designed, simulated, analysed and experimentally tested for the grid-connected inverter. A summary of the key original contributions is shown below; these are separated into their respective parts.

## Part I: Investigation of Switched-Mode Rectifier for Standalone Power Converter

- The high-inductance PM test machine was characterised, via open-circuit, short-circuit and DC stator resistance tests. Its open and short-circuit iron, friction and windage losses were also analysed.
- The generator's ability to operate in the constant-current region for prolonged periods of time, without the stator temperature significantly increasing, was demonstrated.
- The generator was modelled and its I-V and P-V loci were simulated and experimentally verified for a wide range of generator speeds. The fixed ratio of the rectifier's AC input current to DC current output was experimentally verified, whilst the ratio of AC input voltage to the DC output voltage was investigated and shown to vary with rectifier input voltage.
- The SMR's ability to control the high-inductance PM generator using a dynamometer and fixed voltage-source load, was simulated and experimentally demonstrated, i.e. the SMR output current and hence output power and torque were controlled by adjusting the duty-cycle.
- The coefficient of performance curve for a small-scale wind turbine was estimated using tests in a wind tunnel.
- The SMR's ability to control a small-scale wind turbine, using a small wind tunnel and a fixed voltage-source load, was simulated and experimentally verified, i.e. the

SMR output current and hence turbine power and torque were controlled by adjusting the duty-cycle. The principles of maximum power point tracking and over-speed protection was demonstrated.

## Part II: Investigation of Grid-Connected Inverter based on Switched-Mode Rectifier Topology

- A novel low-cost grid-connected inverter topology was designed and simulated.
- A 150W prototype inverter was constructed and tested, for both resistive and gridconnected loads, using a dynamometer test rig.
- The non-ideal inverter input current was analysed, which identified and quantified the harmonic distortion caused by the rectifier ripple and the 100Hz grid power fluctuation.
- The required CL output low-pass filter and its various damped configurations were thoroughly analysed.
- A normalised filter design approach based on the power loss vs. harmonic attenuation vs. power factor trade-off, was developed, and it was shown that the parallel-damped inductor configuration offered the best performance.
- Two feed-forward control algorithms were analysed and shown to effectively remove the output current harmonic distortion caused by the non-ideal current source.
- A high-powered (1kW) grid-connected inverter was designed. This was based on the Australian Standard grid requirements, and the non-ideal PM generator current source, low-pass filter, and feed-forward controller analyses.
- The final 1kW grid-connected inverter design, which contains an optimised lowpass filter and a feed-forward controller, was simulated and its performance was characterised. Its ability to adequately control a small-scale wind turbine over a wide range of wind speeds was demonstrated.

## 7.3 Recommendations for Future Work

Despite the successful demonstration of small-scale wind turbine control, using a lowcost high-inductance PM generator and a low-cost SMR power converter topology, for both standalone and grid-connected applications, some areas of this work require further investigation. These are discussed in the paragraphs below.

The DC analytical model requires refinement, as it is currently unable to accurately predict the rectifier output current vs. voltage locus. This is due to simplifying assumptions, such as the use of a loss less rectifier, and the presence of sinusoidal machine phase currents and voltages. It is recommended that the ratio of DC to AC rectifier voltages be further examined, and that the machine voltage and current waveforms are compared for various generator speeds and load conditions.

It is recommended that the small-scale wind turbine, high-inductance PM generator and the SMR be tested at higher wind speeds, using a larger wind tunnel. This is required to further validate the SMR's ability to adequately control a small-scale wind turbine over a wide range of wind speeds.

The experimentally measured and simulated open-loop inverter output current THD should be further examined for the artificially grid-connected case, as these values do not agree, despite the close matching output current waveforms. The series-connected isolation and autotransformers can be modelled more accurately, such that the simulated harmonics, caused by resonance, better match those measured experimentally.

The high-powered (1kW) grid-connected inverter, including the optimised low-pass CL filter and feed-forward controller should be constructed and tested to verify operation. The inverter, combined with a suitable wind turbine and high-inductance generator, should be tested over a wide range of wind speeds using a wind tunnel, to confirm its performance.

Finally, a closed-loop controller should be designed such that the inverter operates in both power maximisation and constant power modes autonomously.

# Appendix A

# PWM Control Strategies and Low-Pass Filter Design Trade-Offs

## A.1 PWM Switching Schemes

PWM control signals are applied to semiconductor switches at high-frequencies to converter DC voltages / currents to AC voltages / currents. Many PWM schemes exist, these include both *bipolar* and *unipolar sinusoidal*, *selective harmonic elimination*, and *current hysteresis* PWM schemes.

#### A.1.1 Bipolar and Unipolar Pulse-Width Modulation

Bipolar and Unipolar schemes are used to generate a pulse-width modulated signal, whose fundamentals resembles low-frequency sinusoids. The PWM signals are derived by comparing a sinusoidal reference with a triangular wave of at least seven times the fundamental frequency. Examples of the derivation, actual control signal and fundamental waveforms are shown in Figure A.1. The fundamental is obtained by using a low-pass filter filtering.

#### A.1.2 Selective Harmonic Elimination

This scheme uses a simple hardware and software structure to delivers a PWM control signal, as seen in Figure A.2. The signal has five notches and five triggering angles,  $\alpha_1 \dots \alpha_5$ . By careful selection of five angles, a desired fundamental component can be determined which eliminates four harmonics, based on fourier analysis. Typically the 5<sup>th</sup>, 7<sup>th</sup>, 11<sup>th</sup> and 13<sup>th</sup> order harmonics are removed, which increases the magnitude of higher

#### APPENDIX A. PWM CONTROL STRATEGIES AND LOW-PASS FILTER DESIGN TRADE-OFFS



Figure A.1: Principle of sinusoidal pulse-width modulation, showing (a) bipolar, and (b) unipolar inverter output voltages. Each subfigure also shows the sinusoidal reference, the triangular, and the fundamental output sinusoidal waveforms.

order harmonics. This scheme is based on a feed-forward control, where the values of  $\alpha_i$  are stored in a look-up-table.



Figure A.2: Harmonic elimination PWM control signal.

### A.1.3 Current Hysteresis

This control scheme confines the inverter output current within a hysteresis band, as seen in Figure A.3, for a large and small hysteresis band. The output current is detected and compared to a reference current waveform. If the output current exceeds the hysteresis band, the switch control is changed to reduce the output current. Similarly, when the output current reaches the lower hysteresis band, the control signal is adjusted to increase output current. The result is a zig-zag type output current, with the deviation from the reference controlled by the height of the hysteresis band. Note that the switching frequency increases as the hysteresis band decreases.

The grid voltage is used as a current reference, however, harmonics found in the grid voltage are hence found in the output current. The cost of a hysteresis current controller is high due to the use of a closed-loop feedback controller, which requires current sensors and a fast analogue-to-digital (AD) converter.



Figure A.3: Principle of current hysteresis control scheme, showing output current of (a) large, and (b) small hysteresis band width.

## A.1.4 Space Vector Modulation

This feedback control scheme is based on the current hysteresis scheme. Each phase current error is considered simultaneously, and the controller produces the switch signal that delivers the smallest change in current with respect to time.

## A.2 Low-Pass Filter Design

### A.2.1 Power Loss vs. THD Trade-Off - Unipolar PWM Case

The filter design power loss vs. output current THD trade-off, for filter configurations (FC) 1–4, was discussed in Section 5.3.8. It is desired to design a filter in the design region, in which the power loss and current THD are less than 5% each. Examples of the operating points, and the design region, is represented by the shaded box, were earlier seen in Figure 5.43, for FCs 1–4 and cutoff frequencies of 0.2 and 0.3pu (relative to the PWM switching frequency). The cutoff frequency, and design region is again shown in Figure A.4, however, the cutoff frequencies now range from 0.10 to 0.45pu. The figure indicates that the filter can be designed with greater flexibility for lower cutoff frequencies. The figure also shows that the design region is only met providing the cutoff frequency is less than 0.3pu.



Figure A.4: Filter output current THD vs. damping resistor power loss,  $P_d$ , vs. filter cut-off frequency,  $f_c$ . The shaded region indicates the design region, where both the THD and the power loss is limited to 5%. Note that the cut-off frequency is expressed relative to the PWM switching frequency,  $f_{sw}$ , and ranges from 0.1 to 0.45pu, at increments of 0.05pu.

# Appendix B

# **Relevant Publications**

This appendix contains reprints of the two conference papers which summarise the main points of this thesis.

# B.1 Wind Turbine Control using SMR Paper

This paper entitled "Investigation of Switched-Mode Rectifier for Control of Small-Scale Wind Turbines" was presented at the IEEE Industry Applications Conference in Hong Kong, October 2005, and appears in pages 2849 – 2856 of the conference proceedings.

# Investigation of Switched-Mode Rectifier for Control of Small-Scale Wind Turbines

D.M. Whaley, W.L. Soong and N. Ertugrul School of Electrical and Electronic Engineering University of Adelaide Adelaide, Australia

Whaley, D.M., Soong, W.L. & Ertugrul, N. (2005) Investigation of switched-mode rectifier for control of small-scale wind turbines.

Conference Proceedings, IEEE Industry Applications Conference, Fortieth IAS Annual Meeting, Conference Record of the 2005 IEEE, Hong Kong, v. 4, pp. 2849-2856

NOTE:

This publication is included on pages 275-282 in the print copy of the thesis held in the University of Adelaide Library.

It is also available online to authorised users at:

http://dx.doi.org/10.1109/IAS.2005.1518864

## B.2 Novel Low-Cost Grid-Connected Inverter Paper

This paper entitled "Investigation of a Low-Cost Grid-Connected Inverter for Small-Scale Wind Turbines Based on a Constant-Current Source PM Generator" was presented at the IEEE Industry Electronics Conference in Paris, November 2006, and appears in pages 4297 – 4302 of the conference proceedings.

#### Investigation of a Low-Cost Grid-Connected Inverter for Small-Scale Wind Turbines Based on a Constant-Current Source PM Generator

David M. Whaley<sup>1</sup><sup>\*</sup>, Gurhan Ertasgin<sup>1</sup>, Wen L. Soong<sup>1</sup>, Nesimi Ertugrul<sup>1</sup>, James Darbyshire<sup>2</sup>, Hooman Dehbonei<sup>2</sup>, Chem V. Nayar<sup>2</sup>

<sup>1</sup>University of Adelaide, Adelaide <sup>2</sup>Curtin University of Technology, Perth AUSTRALIA <sup>\*</sup> dwhaley@eleceng.adelaide.edu.au

Whaley, D.M., Ertasgin, G., Soong, W.L., Ertugrul, N., Darbyshire, J. Dehbonei, H. & Nayar, C.V. (2006) Investigation of a low-cost grid-connected inverter for small-scale wind turbines based on a constant-current source PM generator.

Conference Proceedings, IEEE Industrial Electronics Conference, IECON 2006, 32nd Annual Conference of IEEE, Paris, pp. 4297-4302

NOTE:

This publication is included on pages 285-290 in the print copy of the thesis held in the University of Adelaide Library.

It is also available online to authorised users at:

http://dx.doi.org/10.1109/IECON.2006.347938

# Appendix C

# Microcontroller Code

## C.1 Switched-Mode Rectifier

The C code listed below is that used by the microcontroller to output a fixed duty-cycle PWM signal, for a switching frequency of 4 kHz. The code makes use of both software and hardware, the latter is used to adjust the duty-cycle in increments of 1%.

```
/****
      SAMPLE PROGRAM 1 FOR MSA0654-MEAUST BOARD
                                                        */
/*
       FILENAME: PWM-percentage.c
                                                        */
      WRITTEN BY: David M Whaley 04-05-2004
DESCRIPTION:
.
*/
*/
*/
      -- Demonstrates the use if INTO and INT1 to increase or decrease a percentage
       -- Exercises port P0 and P1 to turn display
                                                        */
       this percentage on LED1 and LED2.
                                                        */
*/
*/
*/
*/
*/
*/
       -- uses timer A2 to refresh LED's at 100 \, \mathrm{Hz}
       -- uses timers A1 and A2 to make a PWM signal
       the duty cycle is the displayed percentage,
       the same as on the LED's
        *** NOTE: there is a 14mV offset with the
       PWM signal (output pin)
       Copyright 2004 by DM WHALEY
        All rights reserved. See Terms and Conditions
        document\ regarding\ the\ use\ of\ this\ sample
                                                        */
       program .
#include "sfr62.h"
#include "leddata.h"
void main(void);
void initimerA0(void);
void initport(void);
void INTOint(void);
void INTlint(void):
void timerA2int(void);
#pragma INTERRUPT INTOint
```

**#pragma** INTERRUPT INT1int **#pragma** INTERRUPT timerA2int

```
unsigned char counter = 0;
unsigned int sw_freq = 4;
                                           // set switching frequency (in kHz)
unsigned int sw_cycles;
void main(void)
{
         sw_cycles = 16000 / sw_freq; // sets timer using f00
         initimerA0();
         initport();
                                            // set int0 to level 2
         int0ic = 0x02;
                                            // set int1 to level 1
         intlic = 0x02;
         ta2mr = 0x80;
                                            // set up timerA0 for
                                            // for 100Hz LED refresh rate
         ta2 = 0x1388;
                                            // set interrupt level for update (=1)
         ta2ic = 0x01;
         tabsr \mid = 0 \times 04;
                                            // starts timerA2 counting
         while (1):
}
void initimerA0(void)
{
         ta0mr = 0x00;
                                   // set timer A0 for f1 use
         ta0 = sw_cycles;
                                   // set reload register for appropriate period / frequency
                                   // set timer A1 for f1 use and one-shot timer mode
         ta1mr = 0x16;
                                   // timer A1 event/trigger select bit - TA0 overflow selected
         \mathrm{tr}\,\mathrm{g}\,\mathrm{s}\,\mathrm{r}~=~0\,\mathrm{x}\,0\,2\;;
                                   // starts timerA0
         tabsr = 0x01;
                                   // set timer A1 register for 0 cycles \Rightarrow 0.0ms ie 0% duty
         ta1 = 0;
         tabsr \mid = 0x02;
                                   // timerA1 starts counting
}
void initport(void)
{
         \mathrm{pd}0\ =\ 0\,\mathrm{xFF}\,;
                                   // output mode
         pd1 = 0xFF;
                                   // output mode
                                   // no pull up for P0_0 to P0_3
         pu00 = 0;
         pu01 = 0;
                                   // no pull up for P0_4 to P0_7
                                   // no pull up for P1_0 to P1_3
         pu02 = 0;
                                   // initial data to Port 0
// initial data to Port 1
         p0 = 0 x 00;
         \mathrm{p1} \;=\; 0\,\mathrm{xFF}\,;
         p0 = 0 \times 00;
}
void INT0int(void)
{
         unsigned int newtimervalue;
         if (counter < 100)
         {
                  counter++;
                                    // increase PWM duty-cycle by 1%
         }
         else
                                    // or make duty-cycle\ 0\% if already 100\%
         {
                  counter = 0;
         }
         newtimervalue = (sw_cycles/100)*counter;
         tal = newtimervalue;
                                   // set timer A1 register for counter % duty
         tabsr = 0x02;
                                   // timerA1 starts counting
}
```

```
void INT1int(void)
{
        unsigned int newtimervalue;
         if (counter > 0)
         {
                  counter --;
                                    // decrease PWM duty-cycle by 1%
// or make duty-cycle 100% if already 0%
         }
         else
         {
                  counter = 100;
        }
         newtimervalue = (sw_cycles/100) * counter;
                                   // set timer A1 register for counter % duty
// timerA1 starts counting
         tal = newtimervalue;
         tabsr = 0x02;
}
void timerA2int(void)
{
         static unsigned char led_number = 0;
        unsigned char offset;
                                                     // enable interrupts
        asm("fset_I");
         if (led_number == 0)
         {
                  led_number = 1;
                 p0 = 0xFF;

p1 = 0xFD;
                                                      // disable LED 1 and enable LED2
                  if (counter == 100)
                                                      // displays '1H' for 100% duty cycle
                  {
                           p0 = 0x89;
                                                      // displays 'H' on LED 2
                                                      // displays '1' on LED 1
                  }
                  else
                  {
                           offset = counter \% 10;
                           p0 = leddigit[offset]; // display remainder (1's figure) on LED 2
                  }
        }
         else
         {
                  \texttt{led_number} = 0;
                  \mathrm{p0} \ = \ 0\,\mathrm{xFF}\,;
                  \mathrm{p1}\ =\ 0\,\mathrm{xFE}\,;
                                                      // disable LED 2 and enable LED1
                  if (counter == 100)
                                                      // displays '1H' for 100% duty cycle
                  {
                                                      // displays '1' on LED 1
                           p0 = leddigit[1];
                                                       // displays 'H' on LED 2
                  }
                  else
                  {
                           offset = counter / 10;
                           p0 = leddigit[offset]; // display quotient (10's figure) on LED 1
                  }
     }
```

```
}
```

## C.2 Grid-Connected Inverter

The C code listed below is that used by the microcontroller to control the grid-connected inverter in open-loop mode. It makes use of both software and hardware interrupts to synchronise the inverter output current to the grid, whilst allowing the user to control the magnitude by adjusting the modulation index.

```
\textit{FILENAME: PWM-} csi-unf.c
        WRITTEN BY: David M. Whaley 08-03-2006
       DESCRIPTION:
       -- Demonstrates the use if timer B0 to
       interrupt and increase a PWM duty-cycle,
        -- Exercises port P0 and P1 to turn display
       this\ percentage\ on\ LED1\ and\ LED2.
                                                        -- uses timer B1 to refresh LED's at 100 Hz
        -- uses timers A1 and B2 to make a PWM signal
       -- uses timers A4 & A3 to fire thyristors on
        -- uses INTO abd INT1 to increase and decrease
       the modulation index, respectively
        -- the modulation index is the displayed
        on the LED's
        *** NOTE: there is a 14mV offset with the
       PWM signal (output pin)
        Copyright 2006 by David M. WHALEY
        All rights reserved. See Terms and Conditions
        document\ regarding\ the\ use\ of\ this\ sample
        program.
     *****
#include "sfr62.h"
#include "leddata.h"
void main(void);
void initimers(void);
void initport(void);
void INTOint(void);
void INTlint(void);
void INT3int(void);
void timerBlint(void);
void timerB0int(void);
#pragma INTERRUPT INT0int
#pragma INTERRUPT INT1int
#pragma INTERRUPT INT3int
#pragma INTERRUPT timerBlint
#pragma INTERRUPT timerB0int
unsigned char d = 100;
unsigned int sw_freq = 4;
                                        // set switching frequency (in kHz)
unsigned int sw_cycles;
unsigned int thyristorfiring = 2;
unsigned int hundredduty = 2;
unsigned int ma = 100;
unsigned int da = 100;
unsigned int anothercounter = 0;
```

```
void main(void)
{
         sw_cycles = 16000 / sw_freq; // sets timer using f00
         initimers();
         initport();
                                                // set int0 to level 2
         int0ic = 0x02;
                                      // set into to level 2
// set int1 to level 2
// set int3 to level 3 (i.e. highest) rising edge
         intlic = 0x02;
         int3ic = 0x13;
         while (1):
}
void initimers(void)
{
                                      // set timer B2 for f1 use
// set reload register for apprproate period / frequency
         tb2mr = 0x00;
         tb2 = sw_cycles;
                                      // set up timerB1 for
         tb1mr = 0x80;
                                      // for 100Hz LED refresh rate
// set interrupt level for update (=1)
         tb1 = 0x1388;
         t b 1 i c = 0 x 0 2;
                                       // set up timerB0 for
         tb0mr = 0x80;
                                      // for 2kHz interrupt rate
         tb0 = 0xFA;
         t b 0 i c = 0 x 0 1;
                                       // set interrupt level for update (=1)
         ta1mr = 0x16;
                                       // set up timerA0 to use overflow trigger, and pulse TA1 output pin
                                      // set up timerA0 to use overflow trigger, and pulse TA2 output pin
         ta4mr = 0x16;
                                       // used to control thyristor gate fring pulses for opposite devices
                                      ^{\prime\prime} // set up timerA0 to use overflow trigger, and pulse TA3 output pin ^{\prime\prime} // used to control THY gate fring pulses for opposite devices to TA2
         ta3mr = 0x16;
                                       // set timerA1 TB2 overflow , TA3 \ensuremath{\mathfrak{C}} TA4 use falling edge of TA1
         trgsr = 0x01;
                                      // start timersA1-3 & timerB0-2
         tabsr = 0xFA;
         ta1 = 0;
                                      // set timerA1 register for 0 cycles \implies 0.0ms ie 0% duty
         ta4 = 0;
                                      // set all thyristors off
                                      // set all thyristors off
         ta3 = 0;
}
void initport(void)
{
                                      // output mode
         pd0 = 0xFF;
                                      // output mode
// no pull up for P0_0 to P0_3
         pd1 = 0xFF;
         pu00 = 0;
                                      // no pull up for P0_4 to P0_7
// no pull up for P1_0 to P1_3
         pu01 = 0;
         pu02 = 0;
                                      // initial data to Port 0
// initial data to Port 1
         p0 = 0 x 00;
         p1 = 0xFF;
         p0 = 0 x 00;
}
void INTOint(void)
{
         asm("fset_I");
                                      // enable interrupts
         if (ma < 100)
         {
                                      // increase modulation
                   ma++;
         }
                                      // index by 1% (if < 100%)
         else
         {
                  ma = 0;
                                      // otherwise set it to 0%
         }
}
```

```
void INTlint(void)
{
         asm("fset_I");
                                  // enable interrupts
         if (ma > 0)
         {
                                    // decrease modulation
                 ma - -;
                                    // index by 1% (if > 0%)
         }
         else
         {
                                    // otherwise set it to 100\%
                 ma = 100;
         }
}
void INT3int(void)
{
         int 3ic = 0x00;
                                    // disbale int3 interrupt as zero-crosing
                                    // detection circuit output contains noise
                                    // and can interrupt accidentally
                                    // resets LUT vlaue
// *** USED TO SET +VE OR -VE OUPUT
         d = 100:
         thyristorfiring = 1;
                                    // CURRENT POLARITY ***
                                    // *** set to 1 or 2 ***
                                    //\ when \ set \ to \ 1, \ white \ pulses \ at \ zero \ volts
                                    // when set to 2, white pulses at rising edge trigger
         anothercounter=0;
}
void timerBlint(void)
{
         static unsigned char led_number = 0;
         unsigned char offset;
         asm("fset_I");
                                                      // enable interrupts
         if (led_number == 0)
         {
                  led_number = 1;
                  p0 = 0xFF;
                  p1 = 0 x FD;
                                                      // disable LED 1 and enable LED2
                  if (ma == 100)
                                                      // displays '1H' for 100% duty cycle
// displays 'H' on LED 2
// displays '1' on LED 1
                  {
                           p0 = 0x89:
                  }
                  else
                  {
                           offset = ma \% 10;
                           p0 = leddigit[offset]; // display remainder (1's figure) on LED 2
                  }
         }
         else
         {
                  led_number = 0;
                  p0 = 0xFF;
                  p1 = 0 x FE;
                                                      // disable LED 2 and enable LED1
                  if (ma == 100)
                                                      // displays '1H' for 100% duty cycle
// displays '1' on LED 1
// displays 'H' on LED 2
                  {
                           p0 = leddigit[1];
                  }
                  else
                  {
                           offset = ma / 10;
                           p0 = leddigit[offset]; // display quotient (10's figure) on LED 1
                  }
        }
}
```
```
void timerB0int(void)
{
        unsigned int newtimervalue;
        unsigned int newtimervalue2;
                                               // enable interrupts
        asm("fset_I");
        anothercounter++;
        switch(d)
        {
                case 100: d = 84;
                        if(thyristorfiring = 2)
                        {
                                thyristorfiring = 1;
                                                                 // identify which thyristors
                                                                 // are conducting
                                                                 // turn thyristor 1 & 3 pair off
                                ta3 = 0;
                                ta4 = (sw_cycles/100)*15;
                                                                 // turn thyristor 2 & 4 on
                        }
                        else
                        {
                                thyristorfiring = 2;
                                                                 // turn thyristor 2 & 4 off
                                ta4 = 0;
                                ta3 = (sw_cycles / 100) * 15;
                                                                 // turn thyristor 1 & 3 pair on
                        }
                break;
                case 84: d = 69; break;
                case 69: d = 55; break;
                case 55: d = 41; break;
                case 41: d = 29: break:
                case 29: d = 19; break;
                case 19: d = 11; break;
                case 11: d = 5; break;
                case 5: d = 1; break;
                case 1: d = 0; break;
                case 0: d = 2; break;
                case 2: d = 6; break;
                case 6: d = 12; break;
                case 12: d = 20; break;
                case 20: d = 30; break;
                case 30: d = 42; break;
                case 42: d = 56; break;
                case 56: d = 70; break;
                case 70: d = 85; break;
                case 85: d = 100;
                        ta3 = 0;
                         ta4 = 0; break;
        }
        if(anothercounter > 38)
        {
                int 3ic = 0x13;
                                                 // set int3 to level 3 (i.e. highest) rising edge
                                                 // as this is close to 1 period of grid frequency
        }
                                                 // adjusted duty-cycle
        da = 100 - ma + (ma*d)/100;
        newtimervalue = (sw\_cycles/100)*da;
        ta1 = newtimervalue;
                                                 // set timer A1 register for da % duty
}
```

## References

- L.L. Freris, Ed., Wind Energy Conversion Systems. Englewood Cliffs, New Jersey: Prentice Hall, 1990.
- [2] G. Boyle, Ed., *Renewable Energy: Power for a Sustainable Future*. New York: Oxford University Press in association with the Open University, 1996.
- [3] D. Le Gouriérés, *Wind Power Plants: Theory and Design*. Oxford; New York: Pergamon Press, 1982.
- [4] T. Ackermann and L. Söder, "An Overview of Wind Energy-Status 2002," in Renewable and Sustainable Energy Reviews, vol. 6, 2002, pp. 67–128.
- [5] J.F. Walker and N. Jenkins, Wind Energy Technology. New York: John Wiley, 1997.
- [6] J.F. Manwell, J.G. McGowan, and A.L. Rogers, Wind Energy Explained: Theory, Design and Application. Chichester; New York: Wiley, 2002.
- [7] Global Wind Energy Council, "2006 Brochure," Brussels, June 2006.
   [Online]. Available: http://www.gwec.net/fileadmin/documents/Publications/ GWEC\_brochure\_2006.pdf
- [8] International Energy Agency and Organisation for Economic Co-operation and Development, *Electricity Information 2007*. Paris: Organisation for Economic Cooperation and Development: International Energy Agency, 2007.
- [9] L.H. Hansen, F. Blaabjerg, H.C. Christensen, U. Lindhard, and K. Eskildsen, "Generators and Power Electronic for Wind Turbines," in *Proceedings of the IEEE Industrial Electronics Conference*, vol. 3, 2001, pp. 2000–2005.

- [10] J. Marques, H. Pinheiro, H.A. Grndling, J.R. Pinheiro, and H.L. Hey, "A survey on variable-speed wind turbine system," in *Cientifico Greater Fórum of Brazilian Electronics of Power*, 2003, pp. 732–738.
- [11] P. Gipe, "How much energy does it take to build a wind system in relation to the energy it produces?" in *Wind Energy Weekly*, no. 521, 1992, pp. 1–2, accessed 13-January-2007. [Online]. Available: http://www.awea.org/faq/bal.html
- [12] P.W. Carlin, A.S. Laxson, and E.B. Muljadi, "The History and State of the Art of Variable-Speed wind Turbine Technology," in *Wiley InterScience Journal of Wind Energy*, vol. 6, 2003, pp. 1291–59.
- [13] J. Whale, "Small Wind Turbines," in Women Leaders on the Uptake of Renewable Energy Seminar, 2001, pp. 131–134.
- [14] U.S. Department of Energy, "Small Wind Energy Systems for the Homeowner," U.S. Department of Energy: Energy Efficiency and Renewable Energy, Tech. Rep. FS 135, Jan. 1997.
- [15] "Global Wind 2005 Report," Global Wind Energy Council, Brussels, Tech. Rep., Apr. 2006. [Online]. Available: http://www.gwec.net/fileadmin/documents/ Publications/GWEC-Global\_Wind\_05\_Report\_low\_res\_01.pdf
- [16] A. Cameron, "Changing winds: BTM's worl dmarket update," in *Renewable Energy World: Review Issue 2006-2007*, vol. 9, no. 4, July 2006, pp. 56–66.
- [17] R. Wiser and M. Bolinger, "Annual Report on U.S. Wind Power Installation, Cost, and Performance Trends: 2007," U.S. Department of Energy: Energy Efficiency and Renewable Energy, Tech. Rep. LBNL-275E, May 2008, accessed July 16th, 2008. [Online]. Available: http://eetd.lbl.gov/ea/ems/reports/lbnl-275e.pdf
- [18] American Wind Energy Association: Small Wind Turbine Committee, "The U.S. Small Wind Turbine Industry Roadmap: A 20-year industry plan for small wind trubine technology," American Wind Energy Association, Tech. Rep., 2002.
- [19] American Wind Energy Association, "Home and Farm Wind Energy Systems: Reaching the Next Level," American Wind Energy Association, Tech. Rep., June 2002. [Online]. Available: http://www.awea.org/smallwind/documents/ AWEA\_SWT\_Market\_Study\_6-05.pdf

- [20] Danish Wind Industry Association, "Betz' Law," accessed 15-March-2004. [Online]. Available: http://www.windpower.org/en/tour/wres/betz.htm
- [21] K. Tan and S. Islam, "Optimum Control Stratergies in Energy Conversion of PMSG Wind Turbine System without Mechanical Sensors," in *IEEE Transactions on Energy Conversion*, vol. 19, no. 2, June 2004, pp. 392–399.
- [22] E. Hau, Windturbines: Fundamentals, Technologies, Application and Economics. New York: Springer, 2000.
- [23] D.A. Spera, Ed., Wind Turbine Technology: Fundamental Concepts of Wind Turbine Engineering. New York: American Society of Mechanical Engineers, 1994.
- [24] P. Bauer, J.W.H. De Haan, and M.R. Dubois, "Windenergy and Offshore Windparks: State of the Art and Trends," in *Proceedings of the European Conference on Power Electronics and Motion Control*, 2002.
- [25] Rainbow Power Company, "Price List," 2008, accessed 10-April-2008. [Online]. Available: http://www.rpc.com.au/pricelists/pricelist-retail.pdf
- [26] T. Markvart, "Microgrids: Power systems for the 21st century?" *Refocus*, vol. 7, no. 4, pp. 44 48, 2006.
- [27] A. Grauers, "Efficiency of Three Wind Energy Generator Systems," in *IEEE Trans*actions on Energy Conversion, vol. 11, no. 3, 1996, pp. 650–657.
- [28] P. Mutschler and R. Hoffmann, "Comparison of Wind Turbines Regarding their Energy Generation," in *Proceedings of the IEEE Power Electronics Specialists Conference*, vol. 1, 2002, pp. 6–11.
- [29] D. Corbus and D. Prascher, "Analysis and Comparison of Test Results from the Small Wind Research Turbine Test Project," in *Collection of the 2005 ASME Wind Energy Symposium Technical Papers*, no. NREL/CP-500-36891, Jan. 2005.
- [30] P. Migliore, J. van Dam, and A. Huskey, "Acoustic tests of small wind turbines," in *American Institute of Aeronautics and Astronautics*, 2004.
- [31] K. Christensen, "Catalogue of Small Windmills from 5W to 50kW 2006," Nordic Folkecenter for Renewable Energy, Tech. Rep., 2006.

- [32] J.A. Baroudi, V. Dinavahi, and A.M. Knight, "A review of power converter topologies for wind generators," in *Renewable Energy*, vol. 32, no. 14, Nov. 2007, pp. 2369–2385.
- [33] Z. Chen and E. Spooner, "Wind Turbine Power Converters: a Comparative Study," in Proceedings of the IEE Power Electronics and Variable Speed Drives, 1998, pp. 471–476.
- [34] S.J. Collocott, J.B. Dunlop, H.C. Lovatt, and V.S. Ramsden, "Rare-earth permanent magnets: New magnet materials and applications." [Online]. Available: citeseer.ist.psu.edu/709861.html
- [35] A.M. EL-Refaie and T.M. Jahns, "Optimal Flux Weakening in Surface PM Machines using Fractional-Slot Concentrated Windings," in *IEEE Transactions on Industry Applications*, vol. 41, no. 3, May-Jun. 2005, pp. 790–800.
- [36] S. Morimoto, H. Nakayama, M. Sanada, and Y. Takeda, "Sensorless Output Maximization Control for Variable-Speed Wind Generation System Using IPMSG," in *IEEE Transactions on Industry Applications*, vol. 41, no. 1, Jan.-Feb. 2005, pp. 60–67.
- [37] R. Jones and G.A. Smith, "High Quality Mains Power from Variable-Speed Wind Turbines," in *Journal of Wind Engineering*, vol. 18, no. 1, 1994, pp. 45–50.
- [38] A.B. Raju, K. Chatterjee, and B.G. Fernandes, "A Simple Maximum Power Point Tracker for Grid Connected Variable Speed Wind Energy Conversion System with Reduced Switch Count Power Converters," in *Proceedings of the IEEE Power Electronics Specialists Conference*, vol. 2, 2003, pp. 748–753.
- [39] G. Venkataramanan, B. Milkovska, V. Gerez, and H. Nehrir, "Speed Operation of Permanent Magnet Alternator Wind Turbine Using a Single Switch Power Converter," in *Transactions of the ASME, Journal of Solar Energy Engineering*, vol. 118, no. 4, Nov. 1996, pp. 235–238.
- [40] Z. Chen and E. Spooner, "Voltage source inverters for high-power, variable-voltage DC power sources," in *Proceedings of the IEE Generation*, Transmission and Distribution, vol. 148, no. 5, Sept. 2001, pp. 439–447.

- [41] S. Hao, G. Hunter, V. Ramsden, and D. Patterson, "Control System Design for a 20kW Wind Turbine Generator with a Boost Converter and Battery Bank Load," in *Proceedings of the IEEE Power Electronics Specialists Conference*, vol. 4, no. 2, 2001, pp. 2203–2206.
- [42] M.M. Prats, J.M. Carrasco, E. Galvn, J.A. Snchez, and L.G. Franquelo, "A New Fuzzy Logic Controller to Improve the Captured Wind Energy in a Real 800 kW Variable Speed Variable Pitch Wind Turbine," in *Proceedings of the IEEE Power Electronics Specialists Conference*, vol. 1, 2002, pp. 101–105.
- [43] G.A. Smith, "A Novel Converter for VSCF Wind Turbines," in *Renewable Energy*, vol. 9, no. 1-4, 1996, pp. 853–857.
- [44] S.-H. Song, S.-I. Kang, and N.-K. Hahm, "Implementation and Control of Grid Connected AC-DC-AC Power Converter for Variable Speed Wind Energy Conversion System," in *Proceedings of the IEEE Applied Power Electronics Conference and Exposition*, vol. 1, 2003, pp. 154–158.
- [45] K.T. Chau, Y. Fan, and M. Cheng, "A Novel Three-Phase Doubly Salient Permanent Magnet for Wind Power Generation," in *Proceedings of the IEEE Industry Applications Conference*, vol. 1, Oct. 2004.
- [46] Y. Higuchi, N. Yamamura, M. Ishida, and T. Hori, "An improvement of performance for small-scaled wind power generating system with permanent magnet type synchronous generator," in *Proceedings of the IEEE Industrial Electronics Conference*, vol. 2, 2000, pp. 1037–1043.
- [47] E.C.F. Lovelace, "Optimization of a Magnetically Saturable Interior Permanent-Magnet Synchronous Machine Drive," Ph.D. dissertation, Department of Electrical Engineering and Computer Science, Massachusetts Institute of Technology, USA, June 2000.
- [48] F. Liang, J. Miller, and S. Zarei, "A Control Scheme to Maximize Output Power of a Synchronous Alternator in a Vehicle Electrical Power Generation System," in *Proceedings of the IEEE Industry Applications Conference*, vol. 2, 1996, pp. 830–835.
- [49] M. Naidu, N. Boules, and R. Henry, "A Novel Power Generation System for Ground Vehicles," in *Proceedings of the IEEE Digital Avionics Systems Conference*, vol. 2, 1999, pp. 8.C.4–1 – 8.

- [50] G. Hassan, D.J. Perreault, and T.A. Keim, "Design of Dual-Output Alternators with Switched-Mode Rectification," in *IEEE Transactions on Power Electronics*, vol. 20, no. 1, 2005, pp. 164–172.
- [51] J. Rivas, D.J. Perreault, and T. Keim, "Performance Improvement of Alternators With Switched-Mode Rectifiers," in *IEEE Transactions on Energy Conversion*, vol. 19, no. 3, 2004, pp. 561–568.
- [52] D.J. Perreault and V. Caliskan, "Automotive Power Generation and Control," in *IEEE Transactions on Power Electronics*, vol. 19, no. 3, 2004, pp. 618–630.
- [53] W.L. Soong and N. Ertugrul, "Inverterless High-Power Interior Permanent-Magnet Automotive Alternator," in *IEEE Transactions on Industry Applications*, vol. 40, no. 4, Jul-Aug 2004, pp. 1083–1091.
- [54] L.M. Tolbert, W.A. Peterson, M.B. Scudiere, C.P. White, T.J. Theiss, J.B. Andriulli, C.W. Ayers, G. Farquharson, G. Ott, and L.E. Seiber, "Electronic Power Conversion System for an Advanced Mobile Generator Set," in *Proceedings of the IEEE Industry Applications Conference*, vol. 3, 2001, pp. 1763–1768.
- [55] H.-J. Gutt and J. Müller, "New Aspects for Developing And Optimizing Modern Motorcar Generators," in *Proceedings of the IEEE Industry Applications Confer*ence, vol. 1, 1994, pp. 3–8.
- [56] T.M. Jahns and V. Caliskan, "Uncontrolled Generator Operation of Interior PM Synchronous Machines Following High-Speed Inverter Shutdown," in *IEEE Transactions on Industry Applications*, vol. 35, no. 6, Nov./Dec. 1999, pp. 1347–1357.
- [57] S. Sivakumar, T. Parsons, and S.C. Sivakumar, "Modeling, Analysis and Control of Bidirectional Power Flow in Grid Connected Inverter Systems," in *Proceedings of* the IEEE Power Conversion Conference, vol. 3, 2002, pp. 1015–1019.
- [58] M. Calais, J. Myrzik, T. Spooner, and V. Agelidis, "Inverters for Single-Phase Grid Connected Photovoltaic Systems - an Overview," in *Proceedings of the IEEE Power Electronics Specialists Conference*, vol. 4, 2002, pp. 1995–2000.
- [59] Siemens, "High Voltage Direct Current Transmission Proven Technology for Power Exchange," accessed 30-November-2008. [Online]. Available: http: //www.rmst.co.il/HVDC\_Proven\_Technology.pdf

- [60] Z. Chen, "Compensation Schemes for a SCR Converter in Variable Speed Wind Power Systems," in *IEEE Transactions on Power Delivery*, vol. 19, no. 2, 2004, pp. 813–821.
- [61] J. Svensson, "Grid-Connected Voltage Source Converter Control Principles and Wind Energy Applications," Ph.D. dissertation, Department of Electric Power Engineering, Chalmers University of Technology, Göteborg, Sweden, Mar. 1998.
- [62] International Energy Agency, "Grid-Connected Photovoltaic Power Systems: Survey of Inverter and Related Protection Equipments Task V," Central Research Institute of Electric Power Industry, Customer Systems Department; 2-11-1, Iwado Kita, Komae-shi, Tokyo 201-8511, Japan, Tech. Rep. IEA-PVPS T5-05:, 2002.
- [63] A. Grauers, "Synchronous generator and frequency converter in wind turbine applications: system design and efficiency," Licentiate Thesis, Department of Electrical Machines and Power Electronics, Chalmers University of Technology, Göteborg, Sweden, May, 1994.
- [64] F.Z. Peng, "Z-Source Inverter," in *IEEE Transactions on Industry Applications*, vol. 39, no. 2, 2003, pp. 504–510.
- [65] Z. Chen and E. Spooner, "Grid interface options for variable-speed permanentmagnet generators," in *Proceedings of the IEE Electric Power Applications*, vol. 145, no. 4, 1998, pp. 273–283.
- [66] S. Iwata, S. Masukawa, and S. Iida, "A Simplified Cascade Current-Source Inverter Interconnected to a Single-Phase Three-Wire Distribution System and Its Application to Electric Energy Storage Systems with Batteries," in *Electrical Engineering in Japan*, vol. 150, no. 2, 2005, pp. 50–61.
- [67] N. Mohan, T.M. Undeland, and W.P. Robbins, *Power Electronics: Converters, Applications, and Design*, 3rd ed. Hoboken, New Jersey: John Wiley & Sons, Inc., 2003.
- [68] Q. Wang and L. Chang, "PWM Control Strategies for Wind Turbine Inverters," in Proceedings of the IEEE Canadian Conference on Electrical and Computer Engineering, vol. 1, 1998, pp. 309–312.

- [69] B.M. Han, H.J. Kim, and S.T. Baek, "New soft-switching current source converter for photovoltaic power system," in *Electrical Engineering*, vol. 86, 2004, pp. 285–291.
- [70] A.R. Oliva, S.S. Ang, J.C. Balda, and H.G. Chiacchiarini, "Harmonic Distortion Reduction in Power Inverters," in *Proceedings of the IEEE Power Electronics Specialists Conference*, vol. 2, 2004, pp. 1226–1231.
- [71] Y. Jung, G. Yu, J. Choi, and J. Choi, "High-Frequency DC Link Inverter for Grid-Connected Photovoltaic System," in *Proceedings of the IEEE Photovoltaic Specialists Conference*, 2002, pp. 1410–1413.
- [72] M. Meinhardt, "State of the Art and Future Power Electronics for the Use in gridtied and Off-grid PV-Systems: The Industrial Persepective," in *Tutorial of the European Conference on Power Electronics and Applications*, no. 3, Sept. 2005.
- [73] R.L. Steigerwald, A. Ferraro, and F.G. Turnbull, "Application of Power Transistors to Residential and Intermediate Rating Photovoltaic Array Power Conditioners," in *IEEE Transactions on Industry Applications*, vol. IA, no. 2, Mar. 1983, pp. 254–267.
- [74] F. Scapino and F. Spertino, "Circuit Simulation of Photovoltaic Systems for Optimum Interface between PV Generator and Grid," in *Proceedings of the IEEE Industrial Electronics Conference*, vol. 2, 2002, pp. 1125–1129.
- [75] M. Calais, V.G. Agelidis, and M.S. Dymond, "A Cascaded Inverter for Transformerless Single-Phase Grid-Connected Photovoltaic Systems," in *Pergamon Journal on Renewable Energy*, vol. 22, 2001, pp. 255–262.
- [76] P. Welter, "Power up, prices down, grid connected inverter market survey (Leistung rauf, Preise runter Markitübersicht netzgekoppelter Wechselrichter, in German)," *PHOTON-das Solarstrom Magizin (German Solar Electricity Magazine)*, vol. 3, pp. 48–57, 1999.
- [77] C.V. Nayar, H. Dehbonei, and L. Chang, "A Low Cost Power Electronic Interface for Small Sclae Wind Generators in Single Phase Distributed Power Generation System," in *Proceedings of Australasian Universities Power Engineering Conference*, 2005.
- [78] S. Heier, Grid Integration of Wind Energy Conversion Systems. New York: John Wiley, 1998, translated by Rachel Waddington.

- [79] I. Schiemenz and M. Stiebler, "Control of a Permanent Magnet Synchronous Generator Used in a Variable Speed Wind Energy System," in *Proceedings of the IEEE International Electric Machines and Drives Conference*, 2001, pp. 872–877.
- [80] S.B. Kjaer, J.K. Pedersen, and F. Blaabjerg, "A Review of Single-Phase Grid-Connected Inverters for Photovoltaic Modules," in *IEEE Transactions on Industry Applications*, vol. 41, no. 5, 2005, pp. 1292–1306.
- [81] A. Bendre, I. Wallace, J. Nord, and G. Venkataramanan, "A Current Source PWM Inverter With Actively Commutated SCRs," in *IEEE Transactions on Power Electronics*, vol. 17, no. 4, 2002, pp. 461–468.
- [82] Y. Chen and K.M. Smedley, "A Cost-Effective Single-Stage Inverter With Maximum Power Point Tracking," in *IEEE Transactions on Power Electronics*, vol. 19, no. 5, Sept. 2004, pp. 1289–1294.
- [83] E. Biesen, "Continuously Variable," in APPLIANCE Magazine Part 2: Motors & Air-Moving Devices, Oct. 2006, accessed September 24th, 2008.
   [Online]. Available: http://www.appliancemagazine.com/zones/supplier/18\_motor\_mfg/editorial.php?article=1561&zone=18&first=1
- [84] N. Bianchi, M. Dai Prè, L. Alberti, and E. Fornasiero, Theory and Design of Fractional-Slot PM Machines, 1st ed. Padova, Italy: Coop. Libraria Editrice Uinversità di Padova, 2007.
- [85] J.F. Gieras and M. Wing, Permanent Magnet Motor Technology, 2nd ed. New York: Marcel Dekker, 2002.
- [86] W. Wu, V.S. Ramsden, T. Crawford, and G. Hill, "A Low-Speed, High-Torque, Direct-Drive Permanent Magnet Generator For Wind Turbines," in *Proceedings of* the IEEE Industry Applications Conference, vol. 1, 2000, pp. 147–154.
- [87] J. Cros and P. Viarouge, "Synthesis of High Performance PM Motors With Concentrated Windings," in *IEEE Transactions on Energy Conversion*, vol. 17, no. 2, June 2002, pp. 248–253.
- [88] C.-Z. Liaw, W.L. Soong, B.A. Welchko, and N. Ertugrul, "Uncontrolled generation in interior permanent-magnet Machines," in *IEEE Transactions on Industry Applications*, vol. 41, no. 4, Jul.-Aug. 2005, pp. 945–954.

- [89] J.R. Hendershot, Jr. and T.J.E. Miller, Design of Brushless Permanent-Magnet Motors. Hillsboro, OH; Oxford: Magna Physics Publishing and Oxford University Press, 1994.
- [90] J. Chen, C.V. Nayar, and L. Xu, "Design and Finite-Element Analysis of an Outer-Rotor Permanent-Magnet Generator for Directly Coupled Wind Turbines," in *IEEE Transactions on Magnetics*, vol. 36, no. 5, 2000, pp. 3802–3809.
- [91] T. Wildi, *Electrical Machines, Drives, and Power Systems*, 6th ed. Upper Saddle River, New Jersey: Prentice Hall, 2006.
- [92] POWERSIM, "Frequently-Asked Questions," accessed 10-April-2006. [Online]. Available: http://www.powersimtech.com/faq.html
- [93] F. Mazda, *Power Electronics Handbook*, 3rd ed. Oxford: Newnes, 1997.
- [94] P.H. Mellor, R. Wrobel, and N. McNeill, "Investigation of Proximity Losses in a High Speed Brushless Permanent Magnet Motor," in *Proceedings of the IEEE Industry Applications Conference*, vol. 3, no. 2, 2006, pp. 1514–1518.
- [95] EcoInnovation Web site, accessed 10-March-2005. [Online]. Available: http: //www.ecoinnovation.co.nz
- [96] Standards Australia, "Grid connection of energy systems via inverters. Part2: Inverter requirements," Standards Australia International, Sydney, Tech. Rep. AS 4777.2-2002, 2002, accessed April 7, 2005.
- [97] A.M. Gargoom, "Digital Signal Processing Techniques for Improving the Automatic Classification of Power Quality Events," Ph.D. dissertation, The School of Electrical and Electronic Engineering, The University of Adelaide, Australia, Sept. 2006.
- [98] Standards Australia, "Grid connection of energy systems via inverters. Part3: Grid Protection Requirements," Standards Australia International, Sydney, Tech. Rep. AS 4777.3-2002, 2002, accessed April 7, 2005.
- [99] C. Qiao and K.M. Smedley, "Three-Phase Grid-Connected Inverters Interface for Alternative Energy Sources with Unified Constant-Frequency Integration Control," in *Proceedings of the IEEE Industry Applications Conference*, vol. 4, 2001, pp. 2675– 2682.

- [100] G.K. Andersen, C. Klumpner, S.B. Kjaer, and F. Blaabjerg, "A New Green Power Inverter for Fuel Cells," in *Proceedings of the IEEE Power Electronics Specialists Conference*, vol. 2, 2002, pp. 727–733.
- [101] CapSite 2007, "Electrolytic capacitors," accessed 10-April-2007. [Online]. Available: http://my.execpc.com/~endlr/electrolytic.html
- [102] M. Meinhardt and P. Mutschler, "Inverters without transformer in grid connected photovoltaic applications," in *Proceedings of the European Conference on Power Electronics and Applications*, vol. 3, 1995, pp. 86–91.
- [103] R. Naik, N. Mohan, M. Rogers, and A. Bulawka, "A Novel Grid Interface, Optimized for Utility-Scale Applications of Photovoltaic, Wind-Electric, and Fuel-Cell Systems," in *IEEE Transactions on Power Delivery*, vol. 10, no. 4, Oct. 1995, pp. 1920–1926.
- [104] B.K. Bose, Power Electronics and AC Drives. Englewood Cliffs: Prentice Hall, 1986.
- [105] M.J. Nave, Power Line Filter Design for Switched-Mode Power Supplies. New York: Van Nostrand Reinhold, 1991.
- [106] J.H.R. Enslin and P.J.M. Heskes, "Harmonic Interaction Between a Large Number of Distributed Power Inverters and The Distribution Network," in *IEEE Transactions* on Power Electronics, vol. 19, no. 6, Nov. 2004, pp. 1586–1593.
- [107] A. Ambardar, Analog and Digital Signal Processing, 2nd ed. Pacific Grove; California: Brooks/Cole Publishing Company, 1999.
- [108] P. Khamphakdi and W. Khan-ngern, "The Analysis of Output filter for Grid Connected Single Phase Full Bridge Inverter Based PSpice Simulation Technique," in Proceedings of the International Conference on Electromagnetic Compatibility, 2005.
- [109] P. Scherz, Practical Electronics for Inventors. New York: McGraw-Hill, 2000.
- [110] elektronik 21 GmbH, "Audio frequency feedback from wind turbines into the power grid," Mollenbachstraße 19, D-71229 Leonberg, Germany, Tech. Rep., accessed January 19th, 2009. [Online]. Available: http://www.elektronik21.de/ info/wp-content/uploads/2008/05/e21-rueckwirkungwea-en.pdf

- [111] L.A. Serpa, S. Ponnaluri, P.M. Barbosa, and J.W. Kolar, "A Modified Direct Power Control Strategy Allowing the Connection of Three-Phase Inverters to the Grid Through *LCL* Filters," in *IEEE Transactions on Industry Applications*, vol. 43, no. 5, Sep.-Oct. 2007, pp. 1388–1400.
- [112] K.H. Ahmed, S.J. Finney, and B.W. Williams, "Passive Filter Design for Three-Phase Inverter Interfacing in Distributed Generation," in *Journal on Electrical Power Quality and Utilisation*, vol. XIII, no. 2, 2007, pp. 49–58.
- [113] M. Sclocchi, "Input Filter Design for Switching Power Supplies," TechOnLine, Nov. 2002. [Online]. Available: http://archived.techonline.com/community/ related\_content/21453
- [114] P.W. Murrill, Fundamentals of Process Control Theory, 2nd ed. Research Triangle Park, N.C: Instrument Society of America, 1991.
- [115] J.R. Leigh, Control Theory, 2nd ed. London: Institution of Electrical Engineers, 2004.
- [116] M.M. Gupta and N.K. Sinha, Intelligent Control Systems : Theory and Applications, 2nd ed. Piscataway, NJ: IEEE Press, c1996., 1996.
- [117] E.W. Weisstein, "Abel's Impossibility Theorem," accessed 27-September-2007.
   [Online]. Available: http://mathworld.wolfram.com/AbelsImpossibilityTheorem. html
- [118] Standards Australia, "Standard Voltages," Standards Australia International, Strathfield, NSW, Tech. Rep. AS 60038-2002, 2000, accessed July 15th, 2008.